THE ROLE OF TECHNOLOGY PARADOX IN MOBILE TECHNOLOGY USAGE ON THE RELATIONSHIP BETWEEN CONTEXTUAL FACTORS AND TASK PERFORMANCE

MARIA E M A A ALHADAD

UNIVERSITI TEKNOLOGI MALAYSIA

THE ROLE OF TECHNOLOGY PARADOX IN MOBILE TECHNOLOGY USAGE ON THE RELATIONSHIP BETWEEN CONTEXTUAL FACTORS AND TASK PERFORMANCE

MARIA E M A A ALHADAD

A thesis submitted in fulfilment of the requirements for the award of the degree of Doctor of Philosophy

Azman Hashim International Business School Universiti Teknologi Malaysia

NOVEMBER 2021

DEDICATION

This thesis is dedicated to the memory of my mother and father

"May Allah rest their souls in peace."

ACKNOWLEDGEMENT

"In the name of Allah, Most Gracious, Most Merciful."

I thank Allah the almighty for giving me the strength and patience throughout the whole Ph.D. journey to reach this point where I am now and complete the thesis work. A special thank you goes to my supervisor Professor Dr. Rosmini for her generous support and guidance and for constantly encouraging me all along, which helped me a lot in surviving my Ph.D. A special thank you also goes to Dr. Theresa Ho Char Fi for her guidance and valuable insights.

I extend my sincere appreciation to the Vice Voce committee: Chairman Professor Dr. Asan Ali bin Golam Hassan, Professor Dr. Zafir Khan Mohamed Makhbul, and Assoc. Professor Dr. Maslin Masrom. Thank you so much for your questions, recommendations, and time spent reading and examining my thesis to provide fresh eyes and alternative viewpoints where needed. A special thanks to the Azman Hashim International Business School (AHIBS)- Universiti Teknologi Malaysia (UTM), the AHIBS Ph.D. cohort, and to all the academics I met along the way.

To my family, especially my eldest sister Mona, you have always been my biggest supporter, and I will be forever grateful. Also, thank you to everyone who has contributed to this work and cooperated with me by devoting their valuable time to help me in completing this work.

ABSTRACT

Information and communication technologies (ICTs), particularly in the form of smartphones, have introduced a considerable amount of flexibility in the work environment in terms of 'time and space'. One of the important purposes of using these technologies in the workplace is to enhance employee task performance. Although, the use of ICTs can enhance the quality and ease with which employees communicate and organize work, the implications of ICTs in organizational contexts can also be paradoxical with co-existing positive and even negative aspects. This can trigger tension in employees and affect the behavioral aspects of task performance. This research is focused on examining the mediating role played by technology paradox (TCP) between its contextual factors (CNF) and the employee task performance (TSP). A two-phase explanatory sequential design of the mixed- methods approach is applied. In the first phase, quantitative analysis with a cross- sectional survey employing a sample size of 451 respondent has been done. The respondents were employees whose carrying out tasks using private smartphones in different organizations in Kuwait. The data was analyzed using Partial Least Squares - Structural Equation Modelling (PLS-SEM). In the second phase, the quantitative results obtained in the first phase were further explained through the analysis of the qualitative data (i.e in-depth interviews). The in-depth interviews were carried out with five respondents from the earlier set selected based on snowballing sampling. The findings of the research revealed that CNF has a positive relationship with both the TCP and TSP, and the TCP also has a positive relationship with TSP. Further, the TCP has a positive mediating effect between CNF and TSP. The results of this research have provided both theoretical and practical implications. The formation of a positive behavioral influence on the user of technology determines the success of TSP. Also, it suggests managers to provide a context which is congenial to the smartphone usage in the workplace for its most productive use towards TSP.

ABSTRAK

Teknologi maklumat dan komunikasi (ICTs), terutamanya dalam bentuk telefon pintar, telah memperkenalkan sejumlah besar fleksibiliti dalam lingkungan kerja dari segi 'waktu dan ruang'. Salah satu tujuan penting menggunakan teknologi ini di tempat kerja adalah untuk meningkatkan prestasi tugas pekerja. Walaupun penggunaan ICTs dapat meningkatkan kualiti, dan kemudahan di mana pekerja berkomunikasi serta mengatur kerja, terdapat implikasi dalam konteks organisasi, menjadi paradoks samada positif atau negatif. Ini boleh mencetuskan ketegangan pada pekerja dan mempengaruhi tingkah laku dan aspek pelaksanaan tugas. Penyelidikan ini fokus untuk mengkaji peranan pengantara yang dimainkan oleh teknologi paradoks (TCP) antara faktor kontekstualnya (CNF) dan prestasi tugas pekerja (TSP). Reka bentuk jujukan penjelasan dua fasa bagi pendekatan kaedah gabungan (kuantitatif dan kualitatif) telah digunakan. Pada fasa pertama, analisis kuantitatif berbentuk tinjauan keratan lintang, menggunakan sampel 451 responden telah dijalankan. Responden adalah pekerja yang menjalankan tugas menggunakan telefon pintar persendirian di organisasi berbeza di Kuwait. Data dianalisis menggunakan Kuasa Dua Terkecil Separa - Pemodelan Persamaan Struktur (PLS-SEM). Pada fasa kedua, keputusan kuantitatif yang diperolehi dalam fasa pertama dijelaskan lagi melalui analisis data kualitatif (iaitu temu bual mendalam). Temu bual mendalam telah dijalankan dengan lima orang responden daripada set awal yang dipilih berdasarkan persampelan bola salji. Penemuan penyelidikan menunjukkan bahawa CNF mempunyai hubungan positif dengan kedua-dua TCP dan TSP, dan TCP juga mempunyai hubungan positif dengan TSP. Selanjutnya, TCP mempunyai kesan pengantaraan positif antara CNF dan TSP. Hasil penyelidikan ini memberikan implikasi teori dan praktikal. Pembentukan pengaruh tingkah laku yang positif terhadap pengguna teknologi menentukan keberhasilan TSP. Juga, ia mencadangkan pengurus untuk menyediakan konteks yang sesuai dengan penggunaan telefon pintar di tempat kerja untuk kegunaannya yang paling produktif terhadap TSP.

TABLE OF CONTENTS

TITLE

DEC	LARATION	iii
DED	ICATION	iv
ACK	NOWLEDGEMENT	v
ABST	ΓRACT	vi
ABST	ГКАК	vii
TAB	LE OF CONTENTS	viii
LIST	OF TABLES	xiii
LIST	OF FIGURES	XV
LIST	OF ABBREVIATIONS	xvi
LIST	OF APPENDIX	xviii
CHAPTER 1	INTRODUCTION	1
1.1	Introduction	1
1.2	Background of the Study	1
1.3	Problem Statement	7
1.4	Research Questions	9
1.5	Research Objectives	10
1.6	Research Significance	14
1.7	Research Scope	15
1.8	Operational Definitions	20
1.9	Organization of the Study	22
CHAPTER 2	LITERATURE REVIEW	25
2.1	Introduction	25
2.2	Theory of Paradox (ToP)	25
2.3	Technology Paradox (TCP)	28
2.4	The Technology Paradox in Mobile Technology Usage	31

	2.4.1	The Impact of the Paradoxes of Mobile Technology	32	
	2.4.2	Mobile Technology Paradox Perception	34	
	2.4.3	Technology Paradox, Email Overload, and Coping	35	
	2.4.4	Telecommuting and Connectivity Paradox	36	
	2.4.5	Smartphones and Autonomy Paradox	38	
	2.4.6	Technology Paradox in the Context of Flexible Work Design, Autonomy, and Work Intensification	39	
	2.4.7	Impact of Smartphone Usage on Health and Safety, Job Satisfaction, Performance, Productivity, and Work-Life Balance	41	
	2.4.8	Technology Paradox (TCP) and Contextual Factors (CNF)	42	
		2.4.8.1 Technological Context (TCX)	43	
		2.4.8.2 Organizational Context (OCX)	44	
2.5	Task I	Performance (TSP)	45	
2.6	Suppo	orting Theories	48	
	2.6.1	Theory of Technological Mediation (TTM)	49	
	2.6.2	Job Demands and Resources Model (JD-R Model)	51	
	2.6.3	Integrated Technology, Organization, and Environment Framework and The Unified Theory of Acceptance and Use of Technology	52	
27	Hypot	hesis Development	55	
2.,	2.7.1	The Direct Relationship Between Contextual Factors and the Technology Paradox	55	
	2.7.2	The Direct Relationship between the Technology Paradox and Task Performance	57	
	2.7.3	The Direct Relationship Between Contextual Factors and Task Performance	61	
	2.7.4	The Indirect Relationship between CNF and TSP through the TCP as a Mediator	63	
2.8	The C	onceptual Model	65	
2.9	Summ	ary of Literature Review	69	
2.10	Research Gaps			

2.11	Sumn	nary		82
CHAPTER 3	RESE	EARCH M	IETHODOLOGY	85
3.1	Introd	uction		85
3.2	Resea	rch Desig	n (Elements of the Research Strategy)	85
	3.2.1	Research	n Strategy	87
		3.2.1.1	Research Paradigm in Mixed- Methods Research	89
	3.2.2	Extent of	f Research Interference with the Study	94
	3.2.3	Time Ho	rizon	95
	3.2.4	Unit of A	Analysis	95
	3.2.5	Sampling	g Design	96
		3.2.5.1	The Population of Interest	96
		3.2.5.2	The Rationale for the Sample Selection	98
		3.2.5.3	The Sample Design	102
		3.2.5.4	Sample Size	104
	3.2.6	Data Col	llection Methods	107
	3.2.7	Measure	ment and Measures	110
		3.2.7.1	Operationalizing the Concepts	111
		3.2.7.2	Determining the Content of the Survey	120
		3.2.7.3	Establishing a Response Format (Questionnaire)	122
		3.2.7.4	Qualitative Interview Questions	126
		3.2.7.5	Coding	127
		3.2.7.6	Assessing the Reliability and Validity through the Confirmatory Factor Analysis	128
	3.2.8	Data An	alysis	136
		3.2.8.1	Data Screening	138
	3.2.9	Thematic	c Analysis	145
	3.2.10) Mediatic	on Analysis	148
3.3	Summ	nary	-	149

CHAPTER 4	RESULT	152
4.1	Introduction	152
4.2	Demographic Profile	
4.3	Data Screening	156
	4.3.1 Missing Data and Outliers	156
	4.3.2 Normal Distribution of the Data	162
4.4	Reflective Measurement Model	163
4.5	Structural Model	169
	4.5.1 Results of Main Hypothesis (Direct Effects)	169
	4.5.2 Mediating Influence (Main Hypothesis)	170
	4.5.3 Sub-Hypotheses (Direct and Mediating)	172
4.6	Qualitative Phase Results	180
	4.6.1 Stage 1 – Pre-Coding Stage: Data Familiarization	181
	4.6.2 Stage 2 – Generating Initial Codes	181
	4.6.3 Stage 3 – Searching for Themes	182
	4.6.4 Stage 4 – Reviewing Potential Themes	194
	4.6.5 Stage 5 – Defining and Naming Themes	194
	4.6.1 Stage 6 – Producing the Report	198
4.7	Summary	201
CHAPTER 5	DISCUSSION	203
5.1	Introduction	203
5.2	Discussion on the Results	203
5.3	Achievement of the Research Objectives	207
	5.3.1 Discussions on RO1: To Identify the CNFs that Drive the Emergence of the TCP in MT Usage Based on the Experiences of Employees in Kuwait	208
	5.3.1.1 The Relationship between TCX and Both the EEP (H_1) and the IDP (H_2)	209
	5.3.1.2 The Relationship between OCX and Both EEP (H_3) and IDP (H_4)	212

	5.3.2	Discussion on RO2: Relationships Between the and Employee TSP in MT	To Identify the TCP and Both CNF Usage	215
	5.3.3	Discussion on RO3: T Dimensional Level the Between CNF, the TCP, ar MT Usage	o Identify at the Interrelationships ad Employee TSP in	216
		5.3.3.1 The Relationship EEP and the IDF	b Between Both the and TSP (H5, H6)	216
		5.3.3.2 The Relationsh TCX and OCX a	ip Between Both nd TSP (H7, H8)	219
	5.3.4	Discussion on RO4: 7 Mediating Role of the TCF between CNFs and Emp Usage	To Determine the P in the Relationship loyee TSP in MT	222
		5.3.4.1 The Relationsh TCX and OC Mediated by the (H9, H10, H11, 1	ips between Both X and TSP are EEP and the IDP H12)	225
	5.3.1	Discussion on RO5: ' Employee TSP can be Effective MT Usage in Env the TCP	To Explain How Enhanced Through ironments Involving	230
		5.3.1.1 TCX		231
		5.3.1.2 OCX		232
		5.3.1.3 EEP		235
		5.3.1.4 IDP		237
5.4	Summ	ry		238
CHAPTER 6	CON	LUSION		240
6.1	Introd	ction		240
6.2	Theore	ical Contributions		240
6.3	Practic	al Contributions		243
6.4	Study	imitations		251
6.5	Future	Research Directions		253
6.6	Conclu	ding Remarks		255
REFERENCES LIST OF PUBLI	CATIC	NS		259 347

LIST OF TABLES

TABLE NO.	TITLE	PAGE
Table 1.1	Indicators of ICT usage in Kuwait (Communication and Information Technology Regulatory Authority [CITRA], 2020)	2
Table 1.2	A summary of the research problem, questions, objectives, and associated hypotheses	12
Table 2.1	Paradoxes of technology adapted from Mick and Fournier (1998)	29
Table 2.2	Paradoxes of MT adapted from Järvenpää et al. (2005) and Schlanchter et al. (2018)	31
Table 2.3	The key variables and hypotheses	68
Table 2.4	Summary of literature review	70
Table 2.5	Overlap of types of TCPs	78
Table 2.6	A summary of research gaps	80
Table 3.1	F tests - Linear multiple regression: Fixed model, R ² deviation from zero	106
Table 3.2	A summary of the items incorporated in the questionnaire and their sources before factor reduction	117
Table 3.3	Validity threats and strategies to minimize threats, adapted from Creswell and Plano Clark (2017, p. 252)	135
Table 3.4	The variance inflation factor (VIF) and tolerance values (TOL) of the variables of the study	144
Table 3.5	Inter-item correlation matrix	145
Table 4.1	Profile of respondents ($N = 451$)	154
Table 4.2	The histogram and boxplot	157
Table 4.3	The quarter and three-fourth percentile values and the outlier data point	162
Table 4.4	Results for reflective lower-order constructs	165
Table 4.5	Discriminant validity	167
Table 4.6	Hypothesis testing results - direct effect (main hypothesis)	170

Table 4.7	Mediating effect of the technology paradox	171
Table 4.8	Hypothesis testing results - direct and mediating relationships (lower order)	175
Table 4.9	Summary of research findings	177
Table 4.10	The profile of the interview respondents	180
Table 4.11	The initial codes from word counts	182
Table 4.12	Words related to the respondents' expressions (color coding according to similar words)	190
Table 4.13	Using the words to move toward the themes	193
Table 4.14	The main theme, sub-themes, brief descriptions, and sample excerpts	199
Table 6.1	A Summary of the research gaps, research questions, findings, conclusions, contributions, and implications	248

LIST OF FIGURES

FIGURE NO	D. TITLE	PAGE
Figure 1.1	Social media usage in Kuwait (Kemp, 2017, 2018, 2019, 2020)	4
Figure 1.2	Smartphone penetration rate among businesses (CAIT, 2016)	4
Figure 1.3	The evolution of planning toward new Kuwait 2035 adapted from (Mahdi, 2018)	16
Figure 2.1	The direct relationship between contextual factors and task performance	54
Figure 2.2	The conceptual model	67
Figure 3.1	Elements of the research adapted from Sekaran and Bougie (2016)	86
Figure 3.2	Distribution plot	106
Figure 3.3	Phases of thematic analysis adapted from Braun and Clarke (2006, p. 78)	147
Figure 4.1	Reflective measurement model (lower order)	168
Figure 4.2	Structural model (higher-order)	172
Figure 4.3	Structural model (lower order)	176
Figure 4.4	Refined theoretical l model	179
Figure 4.5	Visual representation of (Theme 1)	195
Figure 4.6	Diagrammatic representation of (Theme 1)	196
Figure 4.7	Visual of organization factors linked to performance (Theme 2)	196
Figure 4.8	Diagrammatic representation of (Theme 2)	197
Figure 4.9	Visual of organization support linked to technology acceptance/rejection (Theme 3)	197
Figure 4.10	Diagrammatic representation of Theme 3	198

LIST OF ABBREVIATIONS

CB	-	Covariance Based			
CAIT	-	The Central Agency for Information Technology			
CBS	-	Central Statistical Bureau			
CDT	-	Cognitive Dissonance Theory			
CFA	-	Confirmatory Factor Analysis			
CITRA	-	Communication and Information Technology Regulatory			
		Authority			
CPS	-	Coping Strategies			
CR	-	Composite Reliability			
CNF	-	Contextual Factors			
ESD	-	Explanatory Sequential Design			
DV	-	Dependent Variable			
EEP	-	Empowerment/Enslavement Paradox			
GCC	-	Gulf Cooperation Council			
GDP	-	Gross Domestic Product			
HDR	-	Human Resources development			
HR	-	Human Resources			
HRM	-	Human Resources Management			
ICTs	-	Information and Communication Technologies			
IDP	-	Independence/Dependence Paradox			
IoT	-	Internet of Things			
IR	-	Interview Respondent			
IS	-	Information Systems			
IV	-	Independent Variable			
MMR	-	Mixed Methods Research			
MV	-	Mediating Variable			
NDP	-	National Development Plan			
OCX	-	Organizational Context			
PLS	-	Partial Least Squares			
SEM	-	Structural Equation Modelling			

SNSs	-	Social Networking Sites
TAM	-	Technology Acceptance Model
ТСР	-	Technology Paradox
TCX	-	Technological Context
TOE	-	Technology, Organization, and Environment Model
ToP	-	Theory of Paradox
TTF	-	Task Technology Fit
TTM	-	Theory of Technological Mediation
TSP	-	Task performance
VIF	-	Variance Inflation Factor

LIST OF APPENDIX

APPENDIX	TITLE	PAGE
Appendix A	Back-To-Back Translation	291
Appendix B	Pilot Test Survey Questionnaire	297
Appendix C	Pilot Test Results Of The Survey Questionnaire	304
Appendix D	The Final Survey Questionnaire	309
Appendix E	Qualitative Interview Protocol	312
Appendix F	Empowerment Theme	314
Appendix G	Enslavement Theme	317
Appendix H	Independence Theme	318
Appendix I	Dependence Theme	319
Appendix J	Technological Context Theme	320
Appendix K	Orgnaitaional Theme	333
Appendix L	Task Performance Theme	342
Appendix M	The Mediation Effect of The Technology Paradox	345

CHAPTER 1

INTRODUCTION

1.1 Introduction

This chapter introduces a general background of the study and research topic followed by the problem statement, research questions, and objective. Subsequently, this study's scope and contribution to the relevant body of knowledge are briefly discussed. The definitions of the research constructs and their dimensions are contextbased and differ depending on the study being analyzed; hence, these definitions have been operationalized, as applicable to this research. The chapter ends with a summary of the overall outline of this thesis

1.2 Background of the Study

Information and communications technologies (ICTs) are highly prevalent nowadays, in the form of networks, devices, and applications, used to connect to multiple sources of information to share knowledge (Rice & Leonardi, 2013). Furthermore, economies all over the world are increasingly incorporating ICTs into their industries, business processes, and government services to increase efficiency and productivity and improve the quality of life of everyone (Parra et al., 2021). Similarly, multiple oil-based economies such as Kuwait, where oil exports constitute most of the country's gross domestic product, have been spearheading a drive toward smart cities and related digitalization processes. Kuwait's aim is to achieve economic diversification through its New Kuwait: Vision 2035 strategy (Ringel, 2021; Saxena & Al-Tamimi, 2018), which targets the reduction of both the country's dependence on the oil sector and its fluctuating oil prices (Al-Mutairi et al., 2020). ICTs lie at the heart of Kuwait's new strategy, and Kuwait's government and policymakers in the country are now implementing measures to ensure the increased adoption of ICTs, to the point where such technologies are used in all sectors to leverage the capabilities of rapid technological innovations and developments and boost the productivity of the workforce. According to Huawei Kuwait - ICT Talent Development 2020, 91% of Kuwait's nationals are employed in sectors that are currently at high risk of disruption by new technologies. Employed new technologies and new work structures have resulted in changes to the organization and execution of work in various work environments (Ter Hoeven et al., 2016). It is now difficult to find a work setting where workers do not require the support of ICTs, particularly smartphones, to perform their professional tasks. Smartphones have become an integral part of contemporary workers' lives; even many unskilled labor tasks now require the use of smartphones. There is no doubt that the proliferation of mobile technologies, which combine multiple technologies in a single device, has changed the nature of work and the behaviors of employees who use such devices for work-related purposes.

The present study was conducted in Kuwait, which has witnessed substantial increases in smartphone and mobile social media usage in recent years. Table 1.1 illustrates the usage of ICTs in Kuwait based on Kuwait National ICT Figures; a report published in 2020 while this research was being conducted.

Table 1.1Indicators of ICT usage in Kuwait (Communication and InformationTechnology Regulatory Authority [CITRA], 2020)

Indicators of ICT Usage in Kuwait	Percentage (%)		
Mobile-cellular subscriptions	178.5%		
Active smartphone users	99.5%		
Daily Internet usage per individual	98%		

The information in Table 1.1 indicates that Kuwait is a leading country in the use of smartphone technology (CITRA, 2020).

Kuwait has a dynamic telecommunications sector that emphasizes its mobile infrastructure and services. Zain, Saudi Telecom Company, and Ooredoo are the three leading operators offering mobile platform services on one of the most advanced mobile networks in the Middle East, covering 100% of Kuwait's land area. This system may account for the increase in mobile subscriptions and the rapidly increasing number of people with access to cellphones and mobile Internet connections (CITRA, 2020).

The intensity of ICT usage has increased during the COVID-19 pandemic. In 2020, Internet penetration increased by 0.6%, reaching 99% overall. Similarly, mobile connections increased by 4.5% (7.38 million) and were equivalent to 174% of the total population (4.20 million). In addition, the number of social media users increased by 3.8%, reaching 99% (CITRA, 2020). These growth trends are expected to continue.

Kuwait's culture emphasizes social relationships, oral communication, and face-to-face interaction. Thus, Kuwaiti people like to personalize their relationships and frequently engage in social interaction. However, changes in culture are likely to bring about changes in technology usage and vice versa (Guimarães, 2020). Specifically, the proliferation of new ICTs coupled with the extensive use of the Internet has contributed to the emergence of other methods of socializing that Arabic people are now using, including text-based and social networking communication, through which they can hide their real identities and talk with others in a noncommittal manner (Rouibah & Hamdy, 2009).

Figure 1.1 illustrates social media usage in Kuwait (per the total population) from 2017 to 2020. As shown in the figure, Kuwait has witnessed a surge of 70.69% in social media penetration, jumping from 58% to 99%. Furthermore, the proportion of active mobile social media users increased by 69.49%, from 59% to 100% (CITRA, 2019; Kemp, 2020).



Figure 1.1 Social media usage in Kuwait (Kemp, 2017, 2018, 2019, 2020)

In business contexts, according to the Consolidated Kuwait National ICT Indicators Report (2016), a survey on ICT usage in the government sector that targeted IT decision-makers (N = 254), each representing a separate organization, the Internet has become a critical channel for accessing information regarding government services in Kuwait; less than 7% of Kuwaiti people now obtain information through offline channels. Within the report, another survey on the device penetration rate among businesses (N = 3018) found that the PC smartphone penetration rate was 84%. One of the main priorities of Kuwaiti businesses is the mobilization of enterprise applications; 34% of organizations are now using smartphone applications, and 27% are using custom-developed smartphone applications. Finance and accounting applications had the highest levels of penetration, with 80% of businesses already using them (Figure 1.2) (Central Agency for Information Technology [CAIT], 2016).



Figure 1.2 Smartphone penetration rate among businesses (CAIT, 2016)

Recent reports on Kuwait's pre-pandemic ICT usage (CAIT, 2016; CITRA, 2019), as well as reports on digital device usage in the country (Kemp, 2020), have indicated a growing presence of ICTs, particularly in the form of smartphones, in

people's personal and professional lives. As the smartphone penetration rate increases, so does the level of dependency on these devices (Shin, 2014). Furthermore, based on the recent rise in mobile technology (MT) usage among Kuwaiti people, it is safe to assume that workers in Kuwait are becoming increasingly dependent on smartphones in the workplace. Higher dependence on smartphones means higher relevance of this technology in terms of one's daily professional routine (Lee et al., 2018; Shu et al., 2011).

In recent years, many businesses have recognized the importance of responding to the reverse technology-adoption life cycle, which is currently being observed in workplaces as a result of increasingly more employees bringing experience with consumer technologies to their work environments (Niehaves et al., 2013). Rather than providing employees with new smartphones, organizations that have embraced this growing trend have started rolling out bring-your-own-device (BYOD) policies, which allow employees to use their own devices at work and, in turn, promote autonomy in task performance (TSP), strengthen relationships among workers and their superiors, and encourage knowledge-sharing (Pitichat, 2013). In this manner, BYOD policies contribute substantially to higher work performance; many related studies have reported that employees are more likely to be productive when using their own devices for business-related communication (Andriole, 2012). Technology, however, can be considered a double-edged sword; to leverage the benefits of smartphones also means to experience the inevitable negative aspects of these devices (Järvenpää & Lang, 2005) and therefore generate a so-called technology paradox (TCP), a phenomenon that is likely to influence aspects of an employee's professional performance, such as their TSP (Miron-Spektor et al., 2018).

The concept of the TCP has been extensively researched since it was introduced by Mick and Fournier (1998), a study that resulted in a taxonomy of eight TCPs that users commonly encounter and must deal with. Since then, multiple scholars have studied, directly or indirectly, conflicts in new technologies, such as mobile technologies (De Vries Kedem et al., 2020; Sowon et al., 2019). By contrast, some studies have adopted a *paradox*ical lens to explore the contradictory impact of technology on various aspects of people's lives (Arnold, 2003; Cavazotte et al., 2014;

Fonner & Roloff, 2012; Mazmanian, 2013). Prior research has highlighted two contextual factors (CNF) that act as drivers of the TCPs, such as empowerment– enslavement paradox (EEP) and the independence dependence paradox (IDP) that are likely to manifest in the workplace; these factors are the technological (TCX) and organizational (OCX) contexts (Järvenpää & Lang, 2005; Schlachter et al., 2018; Sowon et al., 2019). For example, smartphones empower employees to coordinate and manage tasks regardless of time and location while enslaving them by making them permanently available for work; such availability intensifies work pressure and leads to closer monitoring and supervision by superiors.

Nonetheless, the TCP relationship, with its antecedent (CNF) and outcomes (TSP) related to employees in the workplace, remains a relatively underexplored area that warrants further research. Previous studies have paid little attention to how paradoxes emerge and evolve owing to various actors (Aust et al., 2015) or how they impact employee performance (Miron-Spektor et al., 2018). Thus, the literature lacks evidence of the existence of an integrative model relating the TCP to CNF and the TSP of employees. Furthermore, the role of the TCP as a mediator between CNF and TSP in organizational contexts, particularly in the Middle East, is underresearched.

The TCP phenomenon in relation to MT usage, a highly disruptive innovation, is now more relevant than ever before. The COVID-19 pandemic brought about several changes worldwide in 2020, such as lockdowns, which have not only transformed people's style of work but also altered the manner in which many professional tasks are executed. Employees are increasingly using Internet-based video-conferencing services, like Zoom, to communicate and continue to work from home, leading to a surge in Internet service usage from 40% before the lockdowns to 100% during the lockdown (De et al., 2020). Consequently, workplace monitoring and problems related to technostress are likely to become more prevalent alongside a notable spike in the presence of digital devices, which in turn escalates the likelihood of experiencing the TCP. Thus, there is a need to understand how employees perceive the TCP phenomenon in relation to MT usage, a context that accentuates the presence of this latent TCP.

1.3 Problem Statement

Mobile technology (MT) has a high potential to become one of the most effective and efficient government tools to offer services to the public. To capitalize on the proliferation of smartphones and social networks in Kuwait, the Kuwaiti government has been transforming its activities according to the demands of convenience and efficiency in interactions for all parties, including citizens, businesses, and all government units. There has been a shift toward a smart or mgovernment (mobile-application-based) strategy that involves the utilization of many types of wireless mobile technologies, services, applications, and devices. In addition, incorporating MT as an additional delivery channel for e-government services could further enhance such services, provide benefits for all (Almie, 2018) and, most importantly, increase employee productivity, which is a key objective of Kuwait's National Development Plan and New Kuwait: Vision 2035 (Mahdi, 2018).

MT, however, has paradoxical implications that manifest in users' usage experiences; these implications are referred to as the TCP. That is, there is an inherent, irresolvable contradiction in such technology; trying to focus on only its positive aspects may defensively trigger its negative aspects. Similarly, concealing the complexity of this interrelationship may activate counterproductive reinforcement cycles that intensify tension, a byproduct of such a paradox (Lewis, 2000; Lewis & Smith, 2014; Smith & Lewis, 2011), which in turn can further affect employee performance (Miron-Spektor et al., 2018).

The sharp rise in the mobile subscription base in Kuwait, coupled with high smartphone usage, a rate of 99% per individual (Kemp, 2020), has increased employees' dependence on smartphones to complete their professional tasks. In addition, the absence of clear BYOD policies that regulate the use of employees' private devices at work alongside a lack of policies and regulations related to e-usage, especially in the government sector (Al-Mutairi et al., 2018), may further intensify the contradictory effects of technology, namely the EEP and IDP. The lack of such policies may influence how employees perceive these paradoxes. Consequently, the TCP is likely to manifest in the experiences of employees in Kuwait. The TCP is perceived

and managed differently by employees working for different organizations; thus, TSP varies among businesses (Järvenpää & Lang, 2005). However, it is unclear how employees in Kuwait perceive the impact of the contradictory aspects of smartphone usage in OCX.

MT provides a rich source of both ironic and paradoxical examples. Arnold (2003) asserted that an understanding of the place of such technology in the world is inadequate if irony and paradox are not considered (Arnold, 2003). Advocates of the paradoxical lens have explored the influence of smartphone usage on various aspects of users' lives (Borges & Joia, 2014; Cavazotte et al., 2014; Mazmanian et al., 2013); however, such studies remain few in number (Sowon et al., 2019), particularly those related to smartphone usage for work-related purposes in non-Western contexts. There is a dearth of literature that adequately explains the mechanisms or contexts behind the quantitative interrelationships between the TCP, CNF, and the TSP of employees; this paper sheds light on several problems associated with these interrelationships.

First, multiple scholars have highlighted a methodological gap due to a lack of variation in research methods that causes overlapping categorization of types of paradoxes. Hence, there is a need to address the phenomenon of the TCP by using other methods, such as quantitative or mixed methods approaches, to verify extant paradoxes in non-Western contexts and to fill the theory application gap, particularly with respect to the theory of paradox (ToP), through which new insights can be generated (Lewis & Smith, 2014). The existing contextual imbalance in theoretical evidence reveals a lack of studies on the TCP from non-Western perspectives. Developed theoretical models (Järvenpää & Lang, 2005; Mick & Fournier, 1998) illustrating the role of the TCP have yet to be comprehensively tested in practical managerial scenarios to support extant models in the literature.

Second, owing to unsound methodologies and variations in existing empirical studies, the following three prominent knowledge gaps need to be filled: (1) How CNF, namely, technological (TCX) and organizational (OCX) contexts act as driving factors in the emergence of the TCP remain unclear (Chae & Yeum, 2010; Järvenpää & Lang, 2005; Mick & Fournier, 1998); (2) the impact of the TCP on certain job aspects, such

as employee TSP (Miron-Spektor et al., 2018), and how the TCP plays a mediating role between CNF and employee TSP remains underexplored; and (3) the results of some early studies (Barley et al., 2011; Cavazotte et al., 2014; Fonner & Roloff, 2010; Schlachter et al., 2018; Ter Hoeven et al., 2016) that encountered the TCP in multiple contexts, including the use of ICTs in the form of smartphones, are inconsistent and conflicting and thus require further research in new contexts.

Therefore, there is a need to address existing gaps in the literature by empirically investigating the relationships of the TCP with CNF and TSP in MT usage and examining the mediating role of the TCP between CNF and TSP. Such an investigation should be conducted based on the experiences of employees who use smartphones for work-related purposes to derive both theoretical and practical implications. Professionals and policymakers in Kuwait must be made aware of the conflicting effects of the use of MT, and understanding TCP usage is a key step in understanding user experiences by providing a more comprehensive view of technology use for work-related purposes (Huang & Zhang, 2019).

1.4 Research Questions

The problem statement in this study led to the development of five research questions to be answered in order to address existing research gaps and to attain the aim of this study. Mixed methods research (MMR) employs an explanatory sequential design (ESD) approach that starts with a quantitative phase (QUAN phase) followed by a qualitative phase (QUAL phase). To address the research questions, the researcher adopted a pragmatic worldview that enabled the employment of multiple methods of data collection to inform the problem under investigation without the need to consider the dichotomy between positivism and interpretivism; the focus of this study was the primary research question as opposed to the methods used (Creswell & Plano Clark, 2017). The following five research questions were proposed:

- **RQ1** Which contextual factors drive the emergence of the technology paradox in mobile technology usage based on the experiences of employees in Kuwait?
- **RQ2** Does the technology paradox in mobile technology usage have relationships with contextual factors and employee task performance?
- **RQ3** Which specific dimensions of contextual factors and the technology paradox in mobile technology usage are related to employee task performance?
- **RQ4** Does the technology paradox in mobile technology usage mediate the influence of contextual factors on employee task performance?
- **RQ5** How can employee task performance be enhanced through effective mobile technology usage in environments involving the technology paradox?

1.5 Research Objectives

This study had two primary aims: (1) to address gaps in the literature related to the TCP phenomenon in MT usage by empirically investigating the TCP's relationships with CNF and TSP to explain how it influences employees' TSP based on the experiences of professionals in Kuwait who use their smartphones for workrelated purposes, and (2) to determine how the TCP mediates the influence of CNF on TSP. To accomplish these aims, the following research objectives were developed:

- **RO1** To identify the contextual factors that drive the emergence of the technology paradox in mobile technology usage based on the experiences of employees in Kuwait.
- **RO2** To identify the relationships of the technology paradox with contextual factors and employee task performance in mobile technology usage.
- **RO3** To identify, at the dimensional level, the interrelationships between contextual factors, the technology paradox, and employee task performance in mobile technology usage.

- **RO4** To determine the mediating role played by the technology paradox between contextual factors and employee task performance in mobile technology usage.
- **RO5** To explain how employee task performance can be enhanced through effective mobile technology usage in environments involving the technology paradox.

A summary of the research problem, questions, objectives, and associated hypotheses is provided in Table 1.2.

Research Problem	Research Questions	Research Objectives	Research Hypotheses
	RQ ₁ Which CNF drive the emergence of the TCP in MT usage based on the experiences of employees in Kuwait?	RO_1 To identify the CNF that drive the emergence of the TCP in MT usage based on the experiences of employees in Kuwait.	 H1: There is a positive relationship between TCX and EEP. H2: There is a positive relationship between TCX. and IDP. H3: There is a positive relationship between OCX and EEP. H4: There is a positive relationship between OCX and IDP.
To address existing gaps in the literature by empirically investigating the relationships of the TCP, in terms of its dimensions, with CNF and TSP in MT usage	RQ ₂ Does the TCP in MT usage have relationships with CNF and employee TSP?	RQ ₂ To identify the relationships of the TCP with CNF and employee TSP in MT usage.	 H_A: There is a positive relationship between CNF and the TCP. H_B: There is a positive relationship between the TCP and TSP. H_C: There is a positive relationship between CNF and TSP.
	RQ ₃ Which specific dimensions of CNF and the TCP in MT usage are related to employee TSP?	RO₃ To identify, at the dimensional level, the interrelationships between CNF, the TCP, and employee TSP in MT usage.	 H₅: There is a positive relationship between EEP and TSP. H₆: There is a positive relationship between IDP and TSP. H₇: There is a positive relationship between TCX and TSP. H₈: There is a positive relationship between OCX and TSP.

Table 1.2A summary of the research problem, questions, objectives, and associated hypotheses

Research Problem	Research Questions	Research Objectives	Research Hypotheses
	RQ ₄ Does the TCP in MT usage mediate the influence of CNF on employee TSP?	RO ₄ To determine the mediating role played by the TCP between CNF and employee TSP in MT usage.	H _D : The relationship between CNF and TSP is mediated by the TCP.
To investigate the mediating role of the TCP in MT usage between CNF and TSP			 H₉: The relationship between TCX and TSP is mediated by EEP. H₁₀: The relationship between OCX and TSP is mediated by EEP. H₁₁: The relationship between TCX and TSP is mediated
			by IDP. H ₁₂ : The relationship between OCX and TSP is mediated by IDP.
	RQ ⁵ How can employee TSP be enhanced through effective MT usage in environments involving the TCP?	RO ₅ To explain how employee TSP can be enhanced through effective MT usage in environments involving the TCP.	

Table 1.2A summary of the research problem, questions, objectives, and associated hypotheses (continued)

1.6 Research Significance

This research makes several contributions to the limited body of knowledge related to the TCP.

First, regarding its theoretical contribution, this research fills a theoretical gap that exists in the literature by extending previous research on the theory of paradox (ToP) in the context of technology to enrich relevant theories, including the job demands and resources model (Demerouti et al., 2001), the theory of technological mediation (Verbeek, 2016), and the integrated technology, organization, and environment (Tornatzky & Fleischer, 1990) and the unified technology acceptance use model (Ventakesh et al., 2013). By complementing or combining the ToP with previously applied theories, this research provides a thorough understanding of how the associated theories can define the nature of the TCP, as well as explain and understand how other theories capture the nature of tensions, which in turn, can provide more theoretical insights and managerial suggestions (Aust et al., 2017; Fairhurst et al., 2016; Lewis & Smith, 2014).

Furthermore, this research contributes a model for understanding the use of MT in the workplace for work-related purposes; the model may add value to the body of knowledge on this subject. The model differs from traditional theories in information systems (IS) that use rational technology adoption theories to investigate the impact of ICTs yet rarely address the concept of the TCP. Drawing on existing theoretical models in the literature that link CNF to the TCP (Järvenpää & Lang, 2005; Mick & Fournier, 1998), and based on a recommendation to extend existing research on the TCP to include newer variables (Ter Hoeven et al., 2016), this study extended these existing models by linking the TCP to CNF and a new variable, namely TSP. In addition, the theoretical model developed in this study was empirically tested by studying the interrelationships between the three research constructs CNF, the TCP, and TSP, as well as the mediating role of the TCP between CNF and TSP. In this way, this study fills the empirical and practical knowledge gaps that exist in the literature.

Second, regarding contextual significance, this research fills a population gap and thus may add to the body of knowledge regarding employees' perceptions of the TCP in a non-Western context, namely, Kuwait, a country with high smartphone engagement.

Third, this research fills a methodological gap by employing an MMR approach to address an under-researched and complex phenomenon, namely the TCP. Therefore, this research leads to greater insights into the research problem compared with previous related studies (Creswell & Plano Clark, 2017).

Finally, regarding its practical significance, this research raises awareness of the TCP; such awareness is required to enhance employee TSP and, therefore, could help organizations and policymakers. In addition, the developed model could be extended in future research or applied in studies conducted in other Gulf Cooperation Council (GCC) countries.

The contributions of this research lie in the identification and filling of extant research gaps, the development of a model that explains the antecedents (CNF) and outcomes (TSP) of the TCP, and the mediating role of the TCP between CNF and TSP to draw implications and bring about transformations in working environments such that TSP is enhanced in the presence of the TCP rather than in its absence.

1.7 Research Scope

Geographically, this research encompasses Kuwait, a country located in the Middle East with a population of 4,464,521 (Central Statistical Bureau [CSB], 2020). The percentage of expatriates working in all sectors is 79.59%, compared with Kuwaitis, who account for only 20.40% of the total workforce. Most Kuwaitis work in the public sector and prefer managerial jobs. By contrast, expatriates tend to work in fields that are of little interest to Kuwaiti nationals, such as manufacturing, construction, retail, wholesale, and hospitality (Labor Market Information System, 2020).

Kuwait's small population and relatively huge oil reserve enable its citizens to enjoy a high standard of living, which is reflected in the exponential rise in ICT usage, especially smartphones. Owing to volatile oil prices, Kuwait has recognized a need to reduce its heavy dependency on the oil sector by transforming itself into a digital economy through its New Kuwait: Vision 2035 strategy (Saxena & Al-Tamimi, 2018). Technology is the essence of this strategy, the aim of which is the construction of a smart city from scratch. Figure 1.3 presents the evolution of the planning stages of New Kuwait: Vision 2035 alongside the strategy's objectives (Mahdi, 2018).



Figure 1.3 The evolution of planning toward new Kuwait 2035 adapted from (Mahdi, 2018)

In 2016, the Kuwaiti government launched a new ICT regulatory body, namely CITRA, which is involved with Internet management, public sector IT development, smart government strategies, national cybersecurity, standards oversight, and investment in IT. Chinese technology firm Huawei signed a memorandum of understanding with CITRA that covers knowledge sharing, technology, and consultancy to support the government's New Kuwait: Vision 2035 strategy. Huawei has been providing infrastructure solutions for mobile operators, as well as other services, for more than ten years. In 2017, Huawei launched its Innovation and Training Center in Kuwait, where it offers training programs for more than 1000 trainees per year from the public and private sectors. The center also provides Huawei solutions for smart cities and smart government, and applications for the Internet of Things. In addition to Huawei, CITRA has also signed memoranda of understanding with other renowned companies, including Cisco Systems, Amazon Web Services, Microsoft, and Whale Cloud (CITRA, 2019).

New Kuwait: Vision 2035 seeks to convert conventional services into eservices to combat administrative corruption that was elevated prior to COVID-19, especially the adverse implications of indigenous practices like "wasta," a relationship-based approach to recruitment and compensation (Budhwar et al., 2019). In other words, the use of influence, networks of people, or connections to get things done. The digitalization of services contributes to the reform of administrative and bureaucratic practices that are pervasive in Kuwait to reinforce transparency and accountability, increase the productivity of the workforce, and achieve human resource development; these goals are the primary objectives of New Kuwait: Vision 2035 (Mahdi, 2018). The government and private sectors are required now more than ever to raise the efficiency of the workforce in the local market and change the public mindset regarding how routine tasks should be executed.

The global pandemic has accentuated the critical role of ICTs, as many pandemic-related disruptions to essential aspects of people's lives (i.e., health care, education, livelihood) have been overcome by the capabilities of ICTs (Parra et al., 2021). Although Kuwait has made strides in transforming itself into a digital economy in recent years, many problems have surfaced during the ongoing pandemic, such as a lag in the digitalization of many pivotal operations, to which the country has engaged in a rather reactive response, namely, the development of Kuwait Vision 2035. The COVID-19 crisis coupled with declining oil prices has increased the pressure on Kuwait, which is now forced to expedite the digitalization of its operations through ICTs. With rapid changes both globally and locally, there is a need to understand employees' perspectives regarding how their smartphones contribute to their professional TSP.

In addition, the ongoing pandemic has brought about several changes resulting from adopted policies related to curfews, be they partial or total, that have reshaped the nature of work during the pandemic. The Kuwaiti government launched its Matta platform, an appointment-scheduling smartphone app that enables government agencies to identify booked appointments by scanning a quick response code on a smartphone in order to register the entry and exit of visitors to all government buildings listed on the platform. In response, the Kuwait Civil Service Commission (CSC), with the approval of the Council of Ministers, has published a guide that covers the procedures and rules for a gradual return to work and introduces a flexible working system in the form of weekly schedules. Employees can work remotely using the rotation system, without rotation, or on-site on the condition that they do not violate the maximum limit for workers specified in the guide to prevent the spread of the virus (CSC, 2020). Similarly, workers in the private sector have been adhering to instructions and precautionary measures by limiting the number of workers on-site and promoting flexible working and teleworking.

Remote working or teleworking (Belzunegui-Eraso & Erro-Garcés, 2020; Boell et al., 2016; Fonner & Roloff, 2012), telecommuting (Chang et al., 2021; Weiwei et al., 2021), and flexible working (Ter Hoeven et al., 2016; Yunus et al., 2021) are all terms that imply some degree of autonomy or empowerment to employees to fulfill their tasks in the way they prefer, choosing the technology they want. However, a coexisting sense of enslavement that cannot be disentangled and is aggravated owing to perpetual connectivity and increased responsiveness to colleagues and superiors inevitably causes dynamic tensions that arise because of the experience of a paradox (Lewis, 2000; Smith & Lewis, 2011). Studies have revealed that some technologies, such as smartphones, are prone to causing more tension than others. Such findings have highlighted the presence of latent tensions, which consequently are now more easily identifiable to scholars (Jarzabkowski et al., 2013). Paradoxes are dynamic in nature; they evolve, change, and fade in and out of people's lives. Such dynamism explains why the perception of existing paradoxes in the literature is likely to change (Chae & Yeum, 2010; Smith & Lewis, 2011) and thus needs to be verified in non-Western contexts, which was one of the objectives of this research.

It would be logical and meaningful to explore multiple ICTs instead of just one, as the use of the same range of capabilities on different devices suggests more commonalities than differences among the devices themselves (Schroeder, 2010). Nonetheless, this study mainly focused on smartphone technology, the most prevalent ICT, for several reasons explained as follows. First, as mentioned, regarding the TCP in relation to ICTs, the scope of this study was limited to smartphones. Smartphones are designed to be carried 24/7 and support their users in multiple ways. The result is that individuals can become strongly attached to their smartphones (Rush, 2011) and expect others to be constantly available for communication throughout the day. Such expectations can generate stress or anxiety when the device is not within immediate reaching distance (Carbonell et al., 2018). Furthermore, such constant connectivity can blur the boundaries between personal and professional domains (Ramarajan & Reid, 2013; Reyt & Wiesenfeld, 2015). Irrespective of their job roles, employees can assume multiple roles simultaneously when using their smartphones for work; for example, an individual can act as a private citizen or an employee. Consequently, the outcomes are generally affected by work-related CNF, which in turn alter how technology is experienced (Middleton et al., 2014).

Second, regarding this study's theoretical foundation, the scope of this research extended to behavioral aspects of the TCP. Although technology is paradoxical, research has revealed that mobile users tend to feel the impact of the contradictory nature of smartphones more strongly than that of any other form of technology (Arnold, 2003; Mick & Fournier, 1998). The integration of multiple technologies into a single device accentuates this paradoxical nature of technology. Furthermore, smartphone users more often engage in close and personal relationships with their devices than do users of any other ICT (Chae & Yeum, 2010; Järvenpää & Lang, 2005). In this manner, a smartphone is an intensely personal device. The close relationship of a user combined with a high intensity of smartphone usage can sooner or later trigger a conflict, where the user experiences simultaneous positive and negative consequences of using their smartphone. In summary, the ambiguities and paradoxes that emerge from smartphone use can provoke internal conflicts in users (Bruzzi & Joia, 2015; Clegg et al., 2002).

Finally, regarding the purpose of this study, the scope of this research extended to include an investigation of the mediating role of the TCP between its antecedent, namely CNF, and its outcome, namely TSP. Although the smartphone is highly ubiquitous, it remains a relatively new technological device that is still young and
constantly evolving (Wilmer et al., 2017). Consequently, the coexisting positive and negative effects of smartphone use are conceptually inseparable and grow in strength with upgrades and new releases (Järvenpää & Lang, 2005).

Thus, the paradoxical implications of smartphones should be examined and analyzed to find solutions in order to mitigate the tension generated by encountering conflict associated with this technological device (Bruzzi & Joia, 2015). In addition, there is a need to understand the paradoxical nature of smartphones in order to better understand the role of technology in both the business world and everyday life (Arnold, 2003). Contemporary workers are operating in increasingly dynamic, uncertain, and ambiguous business environments (Jules & Good, 2014). Therefore, the ability to manage paradoxes is pivotal to corporate success in this highly digitalized world (Järvenpää & Lang, 2005). Recent studies have emphasized the importance of recognizing tensions in order to improve TSP (Ingram et al., 2016; Leung et al., 2018; Liu et al., 2020; Miron-Spektor et al., 2018). There is ample scope for research related to the TCP in such a country as Kuwait; such research could lead to the development of a model representing the GCC countries.

1.8 Operational Definitions

Definitions of the terms used in this research are based on context. The operational definitions applicable to the context of this research are provided as follows:

Contextual Factors (CNF): CNF is composed of two factors, namely TCX and OCX, which act as driving forces in the emergence of the TCP (Järvenpää & Lang, 2005).

Technological Context (TCX): TCX is represented by private smartphone usage during a typical workday due to a perceived need or a job requirement. The value or benefit of smartphones cannot be realized without continued usage of these devices (Deng et al., 2010). Therefore, TCX is a factor that can affect workers' motivation and usage patterns, increasing their susceptibility to the TCP (Järvenpää & Lang, 2005).

Organizational Context (OCX): OCX is represented by organizational strategies, policies, and assigned tactics that support the adoption of private smartphones for work-related purposes (Järvenpää & Lang, 2005) when implementing BYOD policies; OCX facilitates conditions that directly affect usage intensity Buettner, 2015; Van Deursen et al., 2015; Venkatesh et al. 2003).

Technology Paradox (TCP): The TCP is a situation, act, or behavior that seems to have contradictory or inconsistent qualities. The TCP encompasses two dimensions, namely the EEP and IDP, which represent the contradictory aspects of smartphones use (Järvenpää & Lang, 2005; Mick & Fournier, 1998). Focusing on only the positive aspects of these devices may defensively trigger negative effects and activate counterproductive reinforcement cycles that intensify tension (Lewis & Smith, 2014; Smith & Lewis, 2011).

Empowerment–Enslavement Paradox (EEP): The use of private smartphones can promote a sense of empowerment and autonomy in users that may positively affect their TSP; however, at the same time, smartphones can hamper users' freedom because of the increased connectivity and responsiveness that result from smartphone use (Järvenpää & Lang, 2005; Mazmanian et al., 2013; Schlachter et al., 2018).

Independence–Dependence Paradox (IDP): The use of private smartphones provides opportunities for workers to experience a form of independence in terms of time and space and how they execute professional tasks; however, at the same time, smartphones can increase dependency and exacerbate feelings of helplessness (Borges & Joia, 2013; Järvenpää & Lang, 2005; Leonardi et al., 2010).

Task Performance (TSP): The use of private smartphones to perform assigned tasks can enhance the quality of tasks completed at work through increased flexibility in handling different tasks and learning modern methods useful to work;

however, at the same time, it can hinder task completion due to frequent interruptions (Järvenpää & Lang, 2005; Chae & Yeum, 2010).

Mobile Technology (MT): Mobile technologies are handheld IT artifacts that encompass hardware (devices), software (interfaces and applications), and communication (network services). All these features are intertwined; therefore, the device, interface, and applications are considered to be one integrated system that can create value for users. The term "cell phone" includes advanced cell phones or smartphones; the terms "cell phone" and "smartphone" are used interchangeably in this research.

1.9 Organization of the Study

Chapter Two: Chapter Two provides a structured literature review. In addition, a discussion on each of the constructs in this research is presented, and the research gap that was observed during the literature is described. A summary of the literature review is also included. Furthermore, the proposed research model and the research hypotheses and propositions developed based on the study purpose and research objectives are described. The research model illustrates the interrelationships between the three latent variables CNF, the TCP, and TSP alongside their dimensions. This model is based on the main theories discussed in Chapter Two. Finally, a summary of Chapter Two is provided.

Chapter Three: The research methodology adopted in this study is explained in Chapter Three, alongside the reasons for choosing this specific methodology from among the available methods highlighted in this research. In addition, Chapter Three discusses the nature of this research and the variables involved and explains all methods used. The rationale behind the sample selection is discussed, and the methodology adopted for the preparation of a self-administered survey questionnaire is explained. The procedures adopted to assess reliability and validity are also reported. Furthermore, the data collection process is explained, all employed statistical procedures are described, and the procedure for validating the questionnaire is explained. Chapter Three describes the data analysis process, for which an MMR approach with ESD was employed., the assumptions underlying structural equation modeling (SEM) are presented, as are missing values and outliers, sample estimation, the normality of distribution, multicollinearity, and the reliability, validity. In addition, an evaluation of the model and its salient features is presented. Finally, a summary of Chapter Three is provided.

Chapter Four: Chapter Four reports the results of the quantitative and qualitative analysis of the data and provides answers to the research questions. The results are in the form of descriptive statistics followed by the detailed results of the quantitative and qualitative phases of research. Chapter Four concludes with a summary of the findings.

Chapter Five: Chapter Five discusses the results obtained in the previous chapter, interprets the results and their relevance to the literature review, and provides the answers to the research questions to fulfill the objectives of this study. The results are corroborated with earlier studies. Finally, a summary of Chapter Five is provided.

Chapter Six: Chapter Six discusses the theoretical contributions and managerial implications of this research. Furthermore, the limitations of the study and suggestions for future research are described. Chapter Six finally provides several conclusions drawn from this research, thereby highlighting its importance in relation to the technology paradox.

REFERENCES

- Abdollahi, B., & Ibrahim, N. A. (2006). Empowerment golden key of human resources management. *Thehran: Wrayesh Publications*, 50-68.
- Ahmad, A., & Omar, Z. (2015). Improving organizational citizenship behavior through spirituality and work engagement. *American journal of applied sciences*, *12*(3), 200-215.
- Ahmad, S., Zulkurnain, N., & Khairushalimi, F. (2016). Assessing the validity and reliability of a measurement model in structural equation modeling (SEM).
 British Journal of Mathematics & Computer Science, 15(3), 1–8. https://doi.org/10.9734/bjmcs/2016/25183
- Aagaard, J. (2017). Introducing postphenomenological research: a brief and selective sketch of phenomenological research methods. *International Journal of Qualitative Studies in Education*, 30(6), 519–533. https://doi.org/10.1080/09518398.2016.1263884
- Alexandrov, A. (2010). Characteristics of Single-Item Measures in Likert Scale Format. *Electronic Journal of Business Research Methods*, 8(1), pp1-12.
- Almie, M. (2018). Mobile government: Kuwait as a case study. *Journal of Engineering Research and Application*, 8(10), 75–79. https://doi.org/10.9790/9622-0810027579
- Al-Mutairi, A., Naser, K., & Fayez, F. (2018). Obstacles toward adopting electronic government in an emerging economy: Evidence from Kuwait. *Asian Economic* and Financial Review, 8(6), 832–842.

https://doi.org/10.18488/journal.aefr.2018.86.832.842

- Al-Mutairi, A., Naser, K., & Fayez, F. (2020). Factors discourage Kuwaiti employees to participate in the private sector labour force (Kuwaitization). *International Journal of Organizational Analysis*, 28(5), 969–983. https://doi.org/10.1108/IJOA-07-2019-1839
- Aizwamri, A. A. F., & Hussain, N. (2017). The impact of smartphones on work productivity as perceived by employees at a government department in salalah, oman. *Al-'abqari: Journal of Islamic Social Sciences and Humanities*, 12, 25-33.

- Albaum, G. (1997). The Likert scale revisited. *Market Research Society*. *Journal.*, 39(2), 1-21.
- Andriole, S. J. (2012). Seven indisputable technology trends that will define 2015. Communications of the Association for Information Systems, 30(4), 61–72. https://doi.org/10.17705/1cais.03004
- Arthur, W., Bennett, W., Edens, P. S., & Bell, S. T. (2003). Effectiveness of training in organizations: A meta-analysis of design and evaluation features. Journal of Applied Psychology, 88(2), 234-245.
- Ajzen, I. (1991). The theory of planned behavior. Organizational Behavior and Human Decision Processes, 50(2), 179–211. https://doi.org/10.1016/0749-5978(91)90020-T
- Ali, M. A., (2005). Globalization and Industrial Relations of China, India, and South Korea: An Argument for Divergence. *Seminar Research Paper Series*. Paper 17. Retrieved from https://digitalcommons.uri.edu/lrc_paper_seri/17http://digitalcommons.uri.edu
- Almeida, F. (2018). Strategies to perform a mixed methods study. European Journal

/lrc_paper_series/17.

of Education Studies, 5(1), 137-151.

- Alter, S. (2013). Work system theory: overview of core concepts, extensions, and challenges for the future. *Journal of the Association for Information Systems*, *1*4(2), 72-121.
- AlZwamri, F. A. A., & Hussain, N. (2017). The impact of smartphones on work productivity as perceived by employees at a government department in Salalah, Oman. AL-ABQARI: Journal of Islamic Social Sciences and Humanities.
- Andriopoulos, C., & Gotsi, M. (2017). Methods of paradox. In W. K. Smith, P. Jarzabkowski, M. W. Lewis, & A. Langley (Eds.), *The Oxford Handbook of Organizational Paradox* (pp. 513-527). https://doi.org/10.1093/oxfordhb/9780198754428.013.26
- Arnold, M. (2003). On the phenomenology of technology: The "Janus-faces" of mobile phones. *Information and Organization*, 13(4), 231-256. https://doi.org/10.1016/S1471-7727(03)00013-7

- Arthur, W., Bennett, W., Edens, P. S., & Bell, S. T. (2003). Effectiveness of training in organizations: A meta-analysis of design and evaluation features. *Journal of Applied Psychology*, 88(2), 234-245.
- Arpaci, I., Yardimci, Y. C., Ozkan, S., & Turetken, O. (2012). Organizational adoption of information technologies: A literature review. *International Journal of ebusiness and egovernment Studies*, 4(2), 37-50.
- Arslan, M., Zaman, R., & Phil, M. (2014). Effect of empowerment on job performance: A study of software sector of Pakistan. *Research on Humanities and Social Sciences*, 4(27), 23-27.
- Aust, I., Brandl, J., Keegan, A., & Lensges, M. (2017). Tensions in managing human resources. *The Oxford Handbook of Organizational Paradox*, 413.
- Aust, I., Brandl, J. & Keegan, A. (2015). State-of-the-art and future directions for HRM from a paradox perspective: Introduction to the Special Issue. *German Journal of Human Resource Management*, 29(3-4), 194-213.
- Avasilcăi, S., & Rusu, G. (2015, October). Innovation management based on proactive engagement of customers: A case study on LEGO Group. Part II: Challenge of engaging the digital customer. In *IOP Conference Series: Materials Science and Engineering* (Vol. 95, No. 1, p. 012144). IOP Publishing.
- Ayyagari, R., Grover, V., & Purvis, R. (2011). Technostress: technological antecedents and implications. *MIS quarterly*, 35(4), 831-858.
- Bakker, A. B., & Demerouti, E. (2007). The Job Demands-Resources model: State of the art. *Journal of Managerial Psychology*, 22(3), 309–328. https://doi.org/10.1108/02683940710733115
- Bakker, A. B., & De Vries, J. D. (2021). Job Demands–Resources theory and selfregulation: New explanations and remedies for job burnout. *Anxiety, Stress, & Coping*, 34(1), 1-21.
- Baker, J. (2012). The technology–organization–environment framework. *Information systems theory*, 231-245.
- Barker, J. R. (1993). Tightening the iron cage: Concertive control in self-managing teams. *Administrative science quarterly*, 1(1), 408-437.
- Barley, S. R., Meyerson, D. E., & Grodal, S. (2011). E-mail as a source and symbol of stress. *Organization Science*, 22(4), 887-906. https://doi.org/10.1287/orsc.1100.0573

- Basu, E., Pradhan, R. K., & Tewari, H. R. (2017). Impact of organizational citizenship behavior on job performance in Indian healthcare industries. *International Journal of Productivity and Performance Management*, 66(6), 780-796.
- Becker, H. (1998). Tricks of the trade. How to think about your research while you're doing it. Chicago: University of Chicago Press.
- Belzunegui-Eraso, A., & Erro-Garcés, A. (2020). Teleworking in the context of the Covid-19 crisis. Sustainability (Switzerland), 12(9), 1–18. https://doi.org/10.3390/su12093662
- Benson, G. S., Finegold, D., & Mohrman, S. A. (2004). You paid for the skills, now keep them: Tuition reimbursement and voluntary turnover. Academy of Management journal, 47(3), 315-331.
- Bentler, P. M., and Chou, C.-P. (1987). Practical issues in structural modeling. Sociological Methods & Research, 16(1), 78-117.
- Bianchi, A., & Philips, J. G. (2005). The impact of the Internet, multimedia and virtual reality on behavior and society. *Cyber Psychology and Behavior*, 8(1), 39-51.
- Bingimlas, K. A. (2009). Barriers to the successful integration of ICT in teaching and learning environments: A review of the literature. *Eurasia journal of mathematics, science & technology education*, 5(3).
- Boell, S. K., Cecez-Kecmanovic, D., & Campbell, J. (2016). Telework paradoxes and practices: the importance of the nature of work. *New Technology, Work and Employment*, 31(2), 114–131. https://doi.org/10.1111/ntwe.12063
- Borges, A. P., & Joia, L. A. (2014). The role of gender in paradoxes associated with smartphone use by Brazilian executives. Paper presented at the 20th Americas Conference on Information Systems, AMCIS 2014, Retrieved from www.scopus.com
- Bradshaw, C., Atkinson, S., & Doody, O. (2017). Employing a qualitative description approach in health care research. *Global qualitative nursing research*, 4(1), 28-37. https://doi.org/10.1177/2333393617742282
- Braun, V., & Clarke, V. (2019). Reflecting on reflexive thematic analysis. *Qualitative Research in Sport, Exercise and Health*, 11(4), 589-597.
- Braun, V. and Clarke, V. (2006). Using thematic analysis in psychology, *Qualitative Research in Psychology*, 3(2), 77-101.

- Bruzzi, P., & Joia, L. A. (2015). Detecting and sorting the paradoxes associated with smartphone use by Brazilian professionals. Paper presented at the 2015 Americas Conference on Information Systems, AMCIS 2015, Retrieved from www.scopus.com
- Budhwar, P., Pereira, V., Mellahi, K., & Singh, S. K. (2019). The state of HRM in the Middle East: Challenges and future research agenda. *Asia Pacific Journal of Management*, 36(4), 905–933. https://doi.org/10.1007/s10490-018-9587-7
- Buettner, R. (2015). Towards a new personal information technology acceptance model: conceptualization and empirical evidence from a bring your own device dataset. 21st Americas Conference on Information Systems, AMCIS 2015.
- Byron, K., & Peterson, S. (2002). The impact of a large-scale traumatic event on individual and organizational outcomes: Exploring employee and company reactions to September 11, 2001. Journal of Organizational Behavior: The International Journal of Industrial, Occupational and Organizational Psychology and Behavior, 23(8), 895-910.
- Cabello, T. C., Revilla Camacho, M. Á., & Vega Vázquez, M. (2014). The relationship between organizational structure and market orientation: An empirical aproach. *The International Journal of Management Science and Information Technology (IJMSIT)*, (11-Jan-Mar), 1-48.
- Carbonell, X., Chamarro, A., Oberst, U., Rodrigo, B., & Prades, M. (2018).
 Problematic use of the internet and smartphones in university students: 2006–2017. *International journal of environmental research and public health*, 15(3), 475.
- Cascio, W. F., & Montealegre, R. (2016). How technology is changing work and organizations. Annual Review of Organizational Psychology and Organizational Behavior, 3, 349-375.
- Cavalieri, I. C., & Almeida, H. N. (2018). Power, empowerment and social participation-the building of a conceptual model. *European Journal of Social Science Education and Research*, 5(1), 174-185.
- Cavana, R., Delahaye, B., & Sekeran, U. (2001). *Applied business research: Qualitative and quantitative methods*. John Wiley & Sons.
- Cameron, K.S. & Quinn, R.E. (1988). *Organizational paradox and transformation*. Ballinger Publishing Co/Harper and Row Publishers.

- Cascio, W. F., & Montealegre, R. (2016). How technology is changing work and organizations. *Annual Review of Organizational Psychology and Organizational Behavior*, 3, 349-375.
- Cavana, R., Delahaye, B., & Sekeran, U. (2001). *Applied business research: Qualitative and quantitative methods*. John Wiley & Sons.
- Cavazotte, F., Heloisa Lemos, A., & Villadsen, K. (2014). Corporate smart-phones: Professionals' conscious engagement in escalating work connectivity. *New Technology, Work and Employment, 29*(1), 72-87.
- Central Agency for Information Technology, (2016). *Consolidated Kuwait national ICT indicators report 2016*. http://www.cait.gov.kw
- Central Statistical Bureau. (2020). Population estimates of Kuwait by age group, nationality & gender. https://www.csb.gov.kw/Pages/Statistics_en?ID=67&ParentCatID=1
- Chae, M.S. & Yeum, D.S. (2010). The impact of mobile technology paradox perception and personal risk-taking behaviours on mobile technology adoption. *Management Science and Financial Engineering*, *16*(2), 115-138.
- Chang, Y., Chien, C., & Shen, L. F. (2021). Telecommuting during the coronavirus pandemic: Future time orientation as a mediator between proactive coping and perceived work productivity in two cultural samples. *Personality and Individual Differences*, 171, 110508. https://doi.org/10.1016/j.paid.2020.110508
- Chin, W. W., & Newsted, P. R. (1999). Structural equation modeling analysis with small samples using partial least squares in *Statistical strategies for small sample research* (pp. 307-341). Thousand Oaks: Sage Publications.
- Chudoba, K. M., Wynn, E., Lu, M., & Watson-Manheim, M. B. (2005). How virtual are we? Measuring virtuality and understanding its impact in a global organization. *Information systems journal*, 15(4), 279-306.
- Chyung, S. Y., Roberts, K., Swanson, I., & Hankinson, A. (2017). Evidence-based survey design: The use of a midpoint on the Likert scale. *Performance Improvement*, 56(10), 15-23
- Clegg, S. R., da Cunha, J. V., & Cunha, M. P. (2002). Management paradoxes: A relational view, *Human Relations*, 55(5), 483-503. https://doi.org/10.1177/0018726702055005425

Cochran, W. G. (1977). Sampling Techniques: 3d Ed. Wiley.

- Cohen, J., Cohen, E., 1983. Applied Multiple Regression/Correlational Analysis for the Behavioral Sciences. Hillsdale, Erlbaum, NJ.
- Cohen, J. (1988). *Statistical power analysis for behavioural sciences*. Hillsdale: Lawrence Erlbaum Associates.
- Creswell, J. W. (2003). Research Design: Qualitative, quantitative, and mixed methods approahces. California: Sage Publications.
- Creswell, J. W. (2009). Research Design: Qualitative, Quantitative, and Mixed Methods Approaches, NY: SAGE publishers.
- Creswell, J. W. (2011). Controversies in mixed methods research in *Qualitative research* (pp. 269-285). London: Sage.
- Creswell, J. W. (2013). Steps in conducting a scholarly mixed methods study.
- Creswell, J.W. & Plano Clark, V. (2007) *Designing and Conducting Mixed Methods Research.* Thousand Oaks, CA: Sage. <u>https://doi.org/10.1177/1094428108318066</u>
- Creswell, J. W., & Plano Clark, V. L. (2007). Designing and Conducting Mixed Methods Research. In Australian and New Zealand Journal of Public Health (3rd ed., Vol. 31, Issue 4). Sage Publications. https://doi.org/10.1111/j.1753-6405.2007.00096.x.
- Creswell, J. W., & Plano Clark, V. (2017). *Designing and Conducting Mixed Methods Research*. New York: Sage publications.
- Crotty, M. (1998). The foundations of social research: Meaning and perspective in the research process. In *Sage*. St Leonards: Sage Publications.
- Civil Service Commission [CSC]. (2020). The handbook of policies, procedures and rules for gradual return of work in government agencies. https://portal.csc.gov.kw/CSCPortal/viewerservlet?fileName=safty_guide.pdf& amp;fileType=PDF&_afrLoop=398367945560519
- Communication and Information Technology Regulatory Authority [CIRTA]. (2020). *CITRA annual report*. https://citra.gov.kw/sites/En/Pages/Versions.aspx
- Cunha, M. P. e., & Putnam, L. L. (2019). Paradox theory and the paradox of success. *Strategic Organization*, 17(1), 95–106. https://doi.org/10.1177/1476127017739536

- Davcik, S. N. (2014). The use and misuse of structural equation modeling in management research: A review and critique. *Journal of Advances in Management Research*, 11(1), 47-81.
- De Haas, H. (2010). Migration and development: A theoretical perspective. International Migration Review, 44(1), 227–264.
- Demerouti, E., Nachreiner, F., Bakker, A. B., & Schaufeli, W. B. (2001). The job demands-resources model of burnout. *Journal of Applied Psychology*, 86(3), 499–512. https://doi.org/10.1037/0021-9010.86.3.499
- Denscombe, M. (2014). The good research guide: for small-scale social research projects. McGraw-Hill Education (UK).
- Devine, D. J. (2002). A review and integration of classification systems relevant to teams in organizations. *Group Dynamics: Theory, Research, and Practice*, 6(4), 291.
- De Vries, M. F. K. (2013). Organizational paradoxes: *Clinical approaches to management*. Routledge.
- De Vries Kedem, M., Haapio-Kirk, L., Hawkins, C., & Miller, D. (2020). Ageing with smartphones 'From Below': Insights from Japan, Uganda, Al-Quds and Ireland. *AoIR Selected Papers of Internet Research*, 2020.
- Derks, D., & Bakker, A. B. (2014). Smartphone use, work–home interference, and burnout: A diary study on the role of recovery. *Applied Psychology*, 63(3), 411-440.
- Derks, D., ten Brummelhuis, L. L., Zecic, D., & Bakker, A. B. (2014). Switching on and off: Does smart phone use obstruct the possibility to engage in recovery activities?. *European Journal of Work and Organizational Psychology*, 23(1), 80-90.
- Derks, D., Bakker, A. B., & Gorgievski, M. (2021). Private smartphone use during worktime: A diary study on the unexplored costs of integrating the work and family domains. *Computers in Human Behavior*, 114, 106530.
- Dillman, D. A., Phelps, G., Tortora, R., Swift, K., Kohrell, J., Berck, J., & Messer, B. L. (2008). Response rate and measurement differences in mixed-mode surveys using mail, telephone, interactive voice response (IVR) and the Internet. *Social science research*, 38(1), 1-18.

- Dwivedi, Y. K., Rana, N. P., Janssen, M., Lal, B., Williams, M. D., & Clement, M. (2017). An empirical validation of a unified model of electronic government adoption (UMEGA). *Government Information Quarterly*, 34(2), 211-230.
- Eckhardt, A., Laumer, S., & Weitzel, T. (2009). Who influences whom? Analyzing workplace referents' social influence on IT adoption and non-adoption. *Journal of information technology*, 24(1), 11-24.
- Eisenhardt, K. M. (1989). Making fast strategic decisions in high-velocity environments. *Academy of Management journal*, *32*(3), 543-576.
- Etikan, I., & Bala, K. (2017). Sampling and sampling methods. *Biometrics & Biostatistics International Journal*, 5(6), 00149.
- Etikan, I., Musa, S. A., & Alkassim, R. S. (2016). Comparison of convenience sampling and purposive sampling. *American journal of theoretical and applied statistics*, *5*(1), 1-4.
- Ezzamel, M., & Willmott, H. (1998). Accounting for teamwork: A critical study of group-based systems of organizational control. *Administrative Science Quarterly*, 358-396.
- Fairhurst, G. T., Smith, W. K., Banghart, S. G., Lewis, M. W., Putnam, L. L., Raisch, S., & Schad, J. (2017). Diverging and converging: Integrative insights on a paradox meta-perspective. *Academy of Management Annals*, 10(1), 173-182.
- Faul, F., Erdfelder, E., Lang, A. G., & Buchner, A. (2007). G* Power 3: A flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behavior research methods*, 39(2), 175-191.
- Feeney, B. C. (2007). The dependency paradox in close relationships: Accepting dependence promotes independence. *Journal of Personality and Social Psychology*, 92(2), 268.
- Field, A. (2013). *Discovering statistics using IBM SPSS statistics*. London: Sage Publications.
- Fishbein, M., & Ajzen, I. (1975). *Belief, attitude, intention, and behavior: An introduction to theory and research.* Reading, MA: Addison-Wesley.
- Flick, U. (2018). An introduction to qualitative research. Thousand Oaks, CA: Sage.
- Folami, L. B., & Jacobs, F. (2005). The Joint Effect of Task Characteristics And Organizational Context On Job Performance: A Test Using SEM. *Journal of Business & Economics Research (JBER)*, 3(7), 25-40.

- Fonner, K. L., & Roloff, M. E. (2010). Why teleworkers are more satisfied with their jobs than are office-based workers: When less contact is beneficial. *Journal of Applied Communication Research*, 38(4), 336-361. https://doi.org/10.1080/00909882.2010.513998
- Fonner, K. L., & Roloff, M. E. (2012). Testing the connectivity paradox: Linking teleworkers' communication media use to social presence, stress from interruptions, and organizational identification. *Communication Monographs*, 79(2), 205-231.
- Ford, J.K., Weissbein, D.A., & Plamondon, K.E. (2003). Distinguishing organizational from strategy commitment: Linking officers" commitment to job behaviors and satisfaction. *Justice Quarterly*, 20, 159-185.
- Fornell, C., & Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of marketing research*, *18*(1), 39-50.
- Gajendran, R. S., & Harrison, D. A. (2007). The good, the bad, and the unknown about telecommuting: Meta-analysis of psychological mediators and individual consequences. *Journal of Applied Psychology*, 92, 1524_1521.
- GanjiNia, H., Gilaninia, S., & Sharami, R. P. M. (2013). Overview of employees empowerment in organizations. Oman Chapter of Arabian Journal of Business and Management Review, 34(983), 1-6.
- Gebauer, J., Shaw, M. J., & Subramanyam, R. (2007). Once built well, they might come: An empirical study of mobile e-mail. College of Business, University of Illinois.
- Gefen, D., & Straub, D. (2005). A practical guide to factorial validity using PLS-Graph: Tutorial and annotated example. *Communications of the AIS*, *16*, 91-109.
- Gerbing, D. W., & Hamilton, J. G. (1996). Viability of exploratory factor analysis as a precursor to confirmatory factor analysis. Structural Equation Modeling: A Multidisciplinary Journal, 3(1), 62-72.
- Glaser, B. G., & Strauss, A. L. (2017). Discovery of grounded theory: Strategies for qualitative research. Routledge.
- Goodhue, D. L., & Thompson, R. L. (1995). Task-technology fit and individual performance. *MIS Quarterly: Management Information Systems*, 19(2), 213-233. Retrieved from www.scopus.com

- Gholami, R., Sulaiman, A., Ramayah, T. and Molla, A. (2013). Senior managers' perception on green information systems (IS) adoption and environmental performance: Results from a field study. *Information & Management*, 50, 431-438.
- Grewal, R., Cote, J. A., & Baumgartner, H. (2004). Multicollinearity and measurement error in structural equation models: Implications for theory testing. *Marketing science*, 23(4), 519-529.
- Griffin, M. (2013). *Introductory to survey design workshop*. Presented at the 25th Australasian Conference on Information Systems, Melbourne.
- Guimarães, N. M. (2021). Complexity, Design and Culture: convergence for digital experiences. *Interaction Design and Architecture(S)*, 47(21), 274–305.
- Handy, C. B. (1995). The age of paradox. MA: Harvard Business Press.
- Hair, J. F., Anderson, R. E., Tatham, R. L., & Black, W. C. (1992). *Multivariate data analysis with readings*. New York: Maxwell Macmillan.
- Hair, J. F., Black, W. C., Babin, B. J., Anderson, R. E., and Tatham, R. L. (2006).*Multivariate Data Analysis* (6th Ed ed.). Uppersaddle River:: Pearson Prentice Hall.
- Hair, J. F., Ringle, C. M., & Sarstedt, M. (2011). PLS-SEM: Indeed a silver bullet. *The Journal of Marketing Theory and Practice*, 19(2), 139–152.
- Hair, J. F., Ringle, M. C. and Sarstedt, M. (2012). Partial least squares: The better approach to structural equation modeling. *Long Range Planning*, 45, 312-319.
- Hair, J., Hult, G., Ringle, C., & Sarstedt, M. (2014). *A primer on partial least squares structural equation modeling (PLS-SEM)*. Thousand Oaks: Sage Publications.
- Hair, J.F., Hult, G.T.M., Ringle, C.M. and Sarstedt, M. (2017). A primer on partial least squares structural equation modeling (PLS-SEM). Thousand Oaks; Sage publications.
- Hargrave, T. J., & Van de Ven, A. H. (2017). Integrating dialectical and paradox perspectives on managing contradictions in organizations. *Organization Studies*, 38(3-4), 319-339.
- Hassan, S. (2012). Employee attachment to workplace: A review of organizational and occupational identification and commitment. *International journal of organization theory and behavior*, 15(3), 383.

- Hempel, P. S., Zhang, Z. X., & Han, Y. (2012). Team empowerment and the organizational context: Decentralization and the contrasting effects of formalization. *Journal of management*, 38(2), 475-501. https://doi.org/10.1177/0149206309342891
- Henseler, J., Ringle, C. M., and Sinkovics, R. R. (2009). The use of partial least squares path modeling in international marketing. In *New challenges to international marketing* (pp. 277-319): Emerald Group Publishing Limited.
- Hertzog, M. A. (2008). Considerations in determining sample size for pilot studies. *Research in nursing & health*, *31*(2), 180-191.
- Hinkin, T. R. (1998). A brief tutorial on the development of measures for use in survey questionnaires. *Organizational Research Methods*, 1(1), 104-121.
- Holloway, I., & Todres, L. (2003). The status of method: flexibility, consistency and coherence. *Qualitative Research*, *3*(3), 345-357.
- Hoaglin, D. C., Iglewicz, B., & Tukey, J. W. (1986). Performance of some resistant rules for outlier labeling. *Journal of the American Statistical Association*, 81(396), 991-999.
- Huang, L. V., & Zhang, K. (2019). Engagement, Formality, and Visibility: Managing Paradoxes of Using Mobile Instant Messaging for Work. International Journal of Communication, 13, 20.
- Hu, J., Zhang, L., Liang, W. and Wang, Z. (2009) 'Incipient mechanical fault detection based on multifractal and MTS methods', *Petroleum Science*, 6(2), 208–216.
- Hu, Q., & Plant, R. (2001). An empirical study of the casual relationship between IT investment and firm performance. *Information Resources Management Journal*, 14(3), 15-26.
- Huawei. (2020). Huawei Kuwait ICT talent development 2020. file:///C:/Users/DELL/Downloads/White paper+ Huawei_KFAS_ Kuwait ICT Talent Development 2020_vF (1).PDF
- Imran, M., Maqbool, N., & Shafique, H. (2014). Impact of technological advancement on employee performance in banking sector. *International Journal of Human Resource Studies*, 4(1), 57. https://doi.org/10.5296/ijhrs.v4i1.5229
- Ingram, A. E., Lewis, M. W., Barton, S., & Gartner, W. B. (2016). Paradoxes and Innovation in Family Firms: The Role of Paradoxical Thinking.

Entrepreneurship: Theory and Practice, 40(1), 161–176. https://doi.org/10.1111/etap.12113

- Islam, J., Talukder, M., & Hu, H. (2011). The impact of technology, job complexity and religious orientation on managerial performance. *Australasian Accounting*, *Business and Finance Journal*, 5(4), 19-42.
- Jacobs, R. L. (2011). Developing a research problem and purpose statement. In T. S. Rocco & T. G. Hatcher (Eds.), *The Handbook of Scholarly Writing and Publishing* (pp. 125–141). San Francisco, CA: Jossey-Bass.
- Jarvenpaa, S. L., & Lang, K. R. (2005). Managing the paradoxes of mobile technology. *Information Systems Management*, 22(4), 7-23. https://doi.org/10.1201/1078.10580530/45520.22.4.20050901/90026.2
- Jarvenpaa, S. L., Lang, K. R., & Tuunainen, V. K. (2005). Friend or foe? The ambivalent relationship between mobile technology and its users. In *Designing ubiquitous information environments: Socio-technical issues and challenges* (pp. 29-42). Springer, Boston, MA.
- Jarzabkowski, P., Lê, J. K., & Van de Ven, A. H. (2013). Responding to competing strategic demands: How organizing, belonging, and performing paradoxes coevolve. *Strategic Organization*, 11(3), 245-280. https://doi.org/10.1177/1476127013481016
- Johnson, D. S., Bardhi, F., & Dunn, D. T. (2008). Understanding how technology paradoxes affect customer satisfaction with self-service technology: The role of performance ambiguity and trust in technology. *Psychology and Marketing*, 25(5), 416-443. https://doi.org/10.1002/mar.20218
- Johnson, R. B., & Onwuegbuzie, A. J. (2004). Mixed methods research: A research paradigm whose time has come. *Educational Researcher*, *33*(7), 14-26.
- Johnson, R. B., Onwuegbuzie, A. J., & Turner, L. A. (2007). Toward a definition of mixed methods research. *Journal of Mixed Methods Research*, 1(2), 112-133.https://doi.org/10.1177/1558689806298224
- Johns, G., (2006). The essential impact of context on organizational behavior. *Academy of Management Review*, 31(1), 386-408.
- Jones, M., Orlikowski, W., & Munir, K. (2004). Structuration theory and information systems: A critical reappraisal. Social theory and philosophy for information systems, 297-328.

- Joseph, J., Klingebiel, R., & Wilson, A. J. (2016). Organizational structure and performance feedback: Centralization, aspirations, and termination decisions. *Organization Science*, 27(5), 1065-1083.
- Joshi, A., Kale, S., Chandel, S., & Pal, D. K. (2015). Likert scale: Explored and explained. *British Journal of Applied Science & Technology*, 7(4), 396.
- Judge, T. A., Thoresen, C. J., Bono, J. E., & Patton, G. K. (2001). The job satisfaction– job performance relationship: A qualitative and quantitative review. *Psychological bulletin*, 127(3), 376.
- Jules, C., Good, D. (2014). Introduction to special issue on paradox in context: Advances in theory and practice. The Journal of Applied Behavioral Science, 50, 123-126. https://doi.org/10.1177/0021886314524920
- Kabasakal, H., Dastmalchian, A., Karacay, G., & Bayraktar, S. (2012). Leadership and culture in the MENA region: An analysis of the GLOBE project. *Journal of World Business*, 47(4), 519-529.
- Kakodkar, A., Ronge, B., Patankar, A., Mule, S., & Pawar, P. (2017). A concept of knowledge and technology enabled empowerment of rural Indian villages. *Current Science*, 112(4), 750.
- Kemp, S. (2017). *Digital 2017: Kuwait*. https://datareportal.com/reports/digital-2017-kuwait?rq=KUWAIT
- Kemp, S. (2018). Digital 2018: Kuwait. https://datareportal.com/reports/digital-2018kuwait?rq=KUWAIT
- Kemp, S. (2019). *Digital 2019: Kuwait*. https://datareportal.com/reports/digital-2019-kuwait?rq=KUWAIT
- Kemp, S. (2020). *Digital 2020: Kuwait*. https://datareportal.com/reports/digital-2020kuwait?rq=KUWAIT
- Kim, S., Sturman, E., & Kim, E. S. (2015). Structural Equation Modeling: Principles, Processes, and Practices. In The Palgrave Handbook of Research Design in Business and Management (pp. 153-172). Palgrave Macmillan, New York.
- Kitzinger, J. (1995). Qualitative research: Introducing focus groups. BMJ, 311(7000), 299-302.
- Kline, R. B. (2005). *Principles and practice of structural equation modelling* (2nd ed.). New York: Guilford Press.

- Kline, R. B. (2015). Principles and practice of structural equation modeling. Guilford publications.
- Kock, N., & Hadaya, P. (2018). Minimum sample size estimation in PLS-SEM: The inverse square root and gamma-exponential methods. *Information Systems Journal*, 28(1), 227-261.
- Konradt, U., Hertel, G., & Schmook, R. (2003). Quality of management by objectives, task-related stressors, and non-task related stressors as predictors of stress and job satisfaction among teleworkers. *European Journal of Work and Organizational Psychology*, 12, 61-79.
- Krosnick, J. A., & Fabrigar, L. R. (1997). Designing rating scales for effective measurement in surveys. Survey measurement and process quality, 141-164.
- Krosnick, J.A. & Presser, S. (2010). Question and Questionnaire Design' in: Handbook of Survey Research. 2nd Edn. Emerald. pp.263-313.
- Kuss, D. J., Kanjo, E., Crook-Rumsey, M., Kibowski, F., Wang, G. Y., & Sumich, A. (2018). Problematic mobile phone use and addiction across generations: The roles of psychopathological symptoms and smartphone use. *Journal of Technology in Behavioral Science*, 3(3), 141-149.
- Kuwait, Civil Service Bureau. (2012). Announcement No (11) Guidelines for the behavior of government employees. Retrieved from https://www.csc.net.kw:8888/csc/nastashreaat.jsp?d2=2012&d6=11&d7=8&d8 =null&d9=2
- Koopmans, L., Bernaards, C., Hildebrandt, V., van Buuren, S., Van der Beek, A. J., & de Vet, H. C. (2013). Development of an individual work performance questionnaire. *International journal of productivity and performance management*.
- Koopmans, L., Bernaards, C. M., Hildebrandt, V. H., Schaufeli, W. B., De Vet Henrica, C. W., & Van Der Beek, A. J. (2011). Conceptual frameworks of individual work performance: A systematic review. *Journal of Occupational and Environmental Medicine*, 53(8), 856–866. https://doi.org/10.1097/JOM.0b013e318226a763
- Krosnick, J. A., & Fabrigar, L. R. (1997). Designing rating scales for effective measurement in surveys. *Survey measurement and process quality*, 141-164.

- Kwon, M., Lee, J. Y., Won, W. Y., Park, J. W., Min, J. A., Hahn, C., & Kim, D. J. (2013). Development and validation of a smartphone addiction scale (SAS). *PloS one*, 8(2), e56936.
- Labor Market Information System. (2020). *Numerical and percentage distribution for the employees by nationality*. https://lmis.csb.gov.kw/En/IntegratedData_GraphView.aspx?Graph_ID=26
- Lai, K. (2018). Estimating standardized SEM parameters given nonnormal data and incorrect model: Methods and comparison. *Structural Equation Modeling: A Multidisciplinary Journal*, 25(4), 600-620.
- Lambert, E. G., Paoline III, E. A., & Hogan, N. L. (2006). The impact of centralization and formalization on correctional staff job satisfaction and organizational commitment: An exploratory study. *Criminal Justice Studies*, 19(1), 23-44.
- Lanette, S., & Mazmanian, M. (2018, April). The Smartphone" Addiction" Narrative is Compelling, but Largely Unfounded. In *Extended Abstracts of the 2018 CHI Conference on Human Factors in Computing Systems* (pp. 1-6).
- Langfred, C. W. (2000). The paradox of self-management: Individual and group autonomy in work groups. *Journal of Organizational Behavior*, 21(5), 563-585.
- Langfred, C. W., & Moye, N. A. (2004). Effects of task autonomy on performance: an extended model considering motivational, informational, and structural mechanisms. *Journal of applied psychology*, 89(6), 934.
- Lee, M., Kim, K., Lee, K. Y., & Hong, J. H. (2018). Hotel Employees' Use of Smartphones and Performance: Reflective–Formative Estimation Approach. Applying Partial Least Squares in Tourism and Hospitality Research.
- Leedy, P. D., & Omrod, J. E. (2013). *Practical research: Planning and designing*. New Jersey: Pearson.
- Leonardi, P. M., Treem, J. W., & Jackson, M. H. (2010). The connectivity paradox: Using technology to both decrease and increase perceptions of distance in distributed work arrangements. *Journal of Applied Communication Research*, 38(1), 85-105. https://doi.org/10.1080/00909880903483599
- Leung, L. (2011). Effects of ICT connectedness, permeability, flexibility, and negative spillovers on burnout and job and family satisfaction. *Human Technology*, 7(3), 250-267.

- Leung, A. K. Y., Liou, S., Miron-Spektor, E., Koh, B., Chan, D., Eisenberg, R., & Schneider, I. (2018). Middle ground approach to paradox: Within-and betweenculture examination of the creative benefits of paradoxical frames. *Journal of personality and social psychology*, 114(3), 443.
- Leung, L. (2011). Effects of ICT connectedness, permeability, flexibility, and negative spillovers on burnout and job and family satisfaction. *Human Technology*, *7*(3), 250-267.
- Levy, E. C., Rafaeli, S., and Ariel, Y. (2016). The effect of online interruptions on the quality of cognitive performance. *Telemat. Inform.* 33, 1014–1021. doi: 10.1016/j.tele.2016.03.003
- Lewis, M. W. (2000). Exploring paradox: Toward a more comprehensive guide. Academy of Management Review, 25(4), 760-776. https://doi.org/10.5465/AMR.2000.3707712
- Lewis, M. W., & Smith, W. K. (2014). Paradox as a metatheoretical perspective: Sharpening the focus and widening the scope. *Journal of Applied Behavioral Science*, 50(2), 127-149. https://doi.org/10.1177/0021886314522322
- Liao, H., & Chuang, A. (2004). A multilevel investigation of factors influencing employee service performance and customer outcomes. Academy of Management journal, 47(1), 41-58.
- Li, C., Yuan, J. and Qi, Z. (2015) 'Risky group decision-making method for distribution grid planning', *International Journal of Emerging Electric Power Systems*, 16(6), pp. 591–602.
- Lieb, P. S. (2003). The Effect of September 11 th on Job Attribute Preferences and Recruitment. *Journal of Business and Psychology*, *18*(2), 175-190.
- Lin, Y. H., Lin, Y. C., Lee, Y. H., Lin, P. H., Lin, S. H., Chang, L. R., ... & Kuo, T. B. (2015). Time distortion associated with smartphone addiction: Identifying smartphone addiction via a mobile application (App). *Journal of psychiatric research*, 65, 139-145.
- Liao, H., & Chuang, A. (2004). A multilevel investigation of factors influencing employee service performance and customer outcomes. Academy of Management Journal, 47(1), 41-58.

- Lippert, S. K., & Govindarajulu, C. (2006). Technological, organizational, and environmental antecedents to web services adoption. *Communications of the IIMA*, *6*(1), 14.
- Li, P.P., (2016) Global implications of the indigenous epistemological system from the East: How to apply yinyang balancing to paradox management. *Cross Cultural & Strategic Management*, 23(1), 42–77.
- Liu, Y., Xu, S., & Zhang, B. (2020). Thriving at Work: How a Paradox Mindset Influences Innovative Work Behavior. *Journal of Applied Behavioral Science*, 56(3), 347–366. https://doi.org/10.1177/0021886319888267
- Lopez-Fernandez, O., Honrubia-Serrano, L., Freixa-Blanxart, M., & Gibson, W. (2014). Prevalence of problematic mobile phone use in British adolescents. *CyberPsychology, Behavior, and social networking*, 17(2), 91-98.
- Lopez-Fernandez, O., Kuss, D. J., Pontes, H. M., Griffiths, M. D., Dawes, C., Justice, L. V., ... & Billieux, J. (2018). Measurement invariance of the short version of the problematic mobile phone use questionnaire (PMPUQ-SV) across eight languages. *International Journal of Environmental Research and Public Health*, 15(6), 1213.
- Lomranz, J., & Benyamini, Y. (2016). The ability to live with incongruence: Aintegration the concept and its operationalization. *Journal of Adult Development*, 23(2): 79–92.
- Lüscher, L. S., & Lewis, M. W. (2008). Organizational change and managerial sensemaking: Working through paradox. Academy of management Journal, 51(2), 221-240.
- Mahdi, K. (2018, March 14). Kuwait's Plan for the Future. [Paper presentation]. International Conference on Innovational and Economic Diversification in GCC's National Development plans 2018, Kuwait. https://kif.kdipa.gov.kw/wpcontent/uploads/khalid-mahdi-english.pdf
- Marchiori, D. M., Mainardes, E. W., & Rodrigues, R. G. (2018). Do Individual Characteristics Influence the Types of Technostress Reported by Workers? *International Journal of Human–Computer Interaction*, 1, 1-13.
- Mardia, K. V. (1970). Measures of multivariate skewness and kurtosis with applications. *Biometrika*, 57(3), 519-530.

- Marsh, H. W., Balla, J. R., and Hau, K.-T. (1996). An evaluation of incremental fit indices: A clarification of mathematical and empirical properties. *Advanced structural equation modeling: Issues and techniques*, 315-353.
- Mazmanian, M. (2013). Avoiding the trap of constant connectivity: When congruent frames allow for heterogeneous practices. *Academy of Management Journal*, *56*(5), 1225-1250.
- Mazmanian, M. A., Orlikowski, W. J., & Yates, J. (2005). Crackberries: The social implications of ubiquitous wireless e-mail devices. In *Designing ubiquitous information environments: Socio-technical issues and challenges* (pp. 337-343). Springer, Boston, MA.
- Mazmanian, M., Yates, J., & Orlikowski, W. (2006). *Ubiquitous email: Individual experiences and organizational consequences of Blackberry use*. Paper presented at the Annual Meeting of the Academy of Management, Atlanta.
- Mazmanian, M., Orlikowski, W. J., & Yates, J. (2013). The autonomy paradox: The implications of mobile email devices for knowledge professionals. *Organization Science*, 24(5), 1337-1357. https://doi.org/10.1287/orsc.1120.0806
- McKelvie, S. J. (1978). Graphic rating scales—How many categories?. *British Journal* of Psychology, 69(2), 185-202.
- Middleton, C. A., & Cukier, W. (2006). Is mobile email functional or dysfunctional? two perspectives on mobile email usage. *European Journal of Information Systems*, 15(3), 252-260. https://doi.org/10.1057/palgrave.ejis.3000614
- Middleton, C. A. (2007). Illusions of balance and control in an always-on environment: A case study of BlackBerry users. *Continuum*, 21(2), 165-178.
- Middleton, C., Scheepers, R., & Tuunainen, V. K. (2014). When mobile is the norm: Researching mobile information systems and mobility as post-adoption phenomena. *European Journal of Information Systems*, 23(5), 503-512. https://doi.org/10.1057/ejis.2014.21
- Miles, M. B., & Huberman, A. M. (1994). Qualitative data analysis: An expanded sourcebook. California: Sage Publications.
- Miron-Spektor, E., Argote, L. (2008, August). The effect of paradoxical cognition on individual and team innovation. Paper presented at the Academy of Management Meeting, Anaheim, CA. https://doi.org/10.5465/ambpp.2008.33718126

- Miron-Spektor, E., Gino, F., Argote, L. (2011). Paradoxical frames and creative sparks: Enhancing individual creativity through conflict and integration. *Organizational Behavior and Human Decision Processes*, 116(2), 229-240. https://doi.org/10.1016/j.obhdp.2011.03.006
- Miron-Spektor, E., Ingram, A., Keller, J., Smith, W. K., & Lewis, M. W. (2018). Microfoundations of organizational paradox: The problem is how we think about the problem. *Academy of Management Journal*, 61(1), 26-45.
- Moore, G., & Benbasat, I. (1991). Development of an instrument to measure the perceptions of adopting an information technology innovation. *Information Systems Research*, 2(3), 192-222.
- Mohajan, H. K. (2017). Two criteria for good measurements in research: Validity and reliability. *Annals of Spiru Haret University. Economic Series*, *17*(4), 59-82.
- Mohajan, H. K. (2018). Qualitative research methodology in social sciences and related subjects. *Journal of Economic Development, Environment and People*, 7(1), 23-48.
- Molina-Azorin, J. F. (2011). The use and added value of mixed methods in management research. *Journal of Mixed Methods Research*, 5(1), 7-24.
- Moreira, M. G., & Corso, K. B. (2019). The use of smartphones by university students who work: identifying paradoxes of technology. *Revista de Administração da Universidade Federal de Santa Maria*, *12*(5), 1027-1043.
- Morse, J. M. (2016). Mixed Method Design: Principles and Procedures. Routledge.
- Mowday, R. T., & Sutton, R. I. (1993). Organizational behavior: Linking individuals and groups to organizational contexts. *Annual review of psychology*, 44(1), 195-229.
- Müller-Bloch, C., & Kranz, J. (2015). A framework for rigorously identifying research gaps in qualitative literature reviews. Paper presented at the 2015 International Conference on Information Systems: Exploring the Information Frontier, ICIS 2015, Retrieved from www.scopus.com
- Munir, F. F. A. (2018). Reliability and Validity Analysis on the Relationship between Learning Space, Students' Satisfaction and Perceived Performance Using SMART-PLS. International Journal of Academic Research in Business and Social Sciences, 8(1), 775-783.

- Muzzio, H., & Palva, J. F. G. (2018). Organizational Creativity Management: Discussion Elements. *Revista de Administração Contemporânea*, 22(6), 922-939.
- Myers, M. D. (1997). Qualitative research in information systems. *MIS Quarterly*, 21(2).
- Myers, M. D., & Newman, M. (2007). The qualitative interview in IS research: Examining the craft. *Information and Organization* 17(1), 2-26.
- Nadler, J. T., Weston, R., & Voyles, E. C. (2015). Stuck in the middle: the use and interpretation of mid-points in items on questionnaires. *The Journal of general psychology*, 142(2), 71-89.
- Naik, Y. N., & Bisht, M. (2014). Structural Equation Model for Effectiveness of Technical Managers in Information Technology Industry. *International Journal* of Supply Chain Management, 3(3), 172-181.
- Nevitt, J., & Hancock, G. R. (2000). Improving the root mean square error of approximation for nonnormal conditions in structural equation modeling. *The Journal of Experimental Education*, 68(3), 251-268.
- Nunnally, J. C. (1978). Psychometric theory (2nd ed.): New York: McGraw-Hill.
- Oguz, F. (2016). Organizational influences in technology adoption decisions: A case study of digital libraries. College & Research Libraries, 77(3), 314-334.
- Orlikowski, W. J. (1991). Integrated information environment or matrix of control? The contradictory implications of information technology. *Accounting, Management and Information Technologies*, 1(1), 9-42. Park, C., & Jun, J. (2003). A cross-cultural comparison of Internet buying behaviour. *International Marketing Review*, 20(5), 534-553.
- Orlikowski, W. J. (2008). Using technology and constituting structures: A practice lens for studying technology in organizations in *Resources, co-evolution and artifacts* (pp. 255-305). London: Springer-Verlag London Limited.
- Orpen, C. (1985). The Effects of Need for Achievement and Need for Independence on the Relationship Between Perceived Job Attributes and Managerial Satisfaction and Performance, *International Journal of Psychology*, 20(2), 207-219. https://doi.org/10.1080/00207598508247733
- Ou, C. X. J., & Davison, R. M. (2011). Interactive or interruptive? Instant messaging at work. *Decision Support Systems*, 52(1), 61-72.

- Palinkas, L. A., Horwitz, S. M., Green, C. A., Wisdom, J. P., Duan, N., & Hoagwood, K. (2015). Purposeful sampling for qualitative data collection and analysis in mixed method implementation research. *Administration and Policy in Mental Health and Mental Health Services Research*, 42(5), 533-544.
- Parjanen, S. (2012). Experiencing creativity in the organization: From individual creativity to collective creativity. *Interdisciplinary Journal of Information, Knowledge, and Management,* 7, 109-128. Retrieved from http://www.ijikm.org/Volume7/IJIKMv7p109-128Parjanen586.pdf
- Park, S. H., Lee, E. H., Blackman, J., Ertmer, P., Simons, K., & Belland, B. (2005). Examining the barriers encountered when planning and implementing technology-enhanced PBL in the middle school classroom. In *Society for Information Technology & Teacher Education International Conference* (pp. 2039-2043). Association for the Advancement of Computing in Education (AACE).
- Park, N., Kim, Y. C., Shon, H. Y., & Shim, H. (2013). Factors influencing smart phone use and dependency in South Korea. Computers in Human Behavior, 29(4), 1763-1770.
- Park, Y., Fritz, C., & Jex, S. M. (2011). Relationships between work-home segmentation and psychological detachment from work: the role of communication technology use at home. *Journal of Occupational Health Psychology*, 16(4), 457.
- Park, J. H., Kim, M. K., & Paik, J. H. (2015). The factors of technology, organization, and environment influencing the adoption and usage of big data in Korean Firms, *Proceedings of 26th European Regional ITS Conference, Madrid 2015 127173, International Telecommunications Society (ITS).*
- Park, H. J., & Zhang, Y. (2021). Technology readiness and technology paradox of unmanned convenience store users. *Journal of Retailing and Consumer Services*, 102523.
- Parra, C. M., Gupta, M., & Mikalef, P. (2021). Information and communication technologies (ICT)-enabled severe moral communities and how the (Covid19) pandemic might bring new ones. *International Journal of Information Management*, 57, 1–16. https://doi.org/10.1016/j.ijinfomgt.2020.102271

- Paschal, A. O., & Nizam, I. (2016). Effects of Organisational Culture on Employees Performance: Case of Singapore Telecommunication, *International Journal of Accounting & Business Management*, 4(1), 19-26.
- Perrone, V., Zaheer, A., & McEvily, B. (2003). Free to be trusted? Organizational constraints on trust in boundary spanners. *Organization Science*, 14(4), 422-439.
- Perlow, L., & Weeks, J. (2002). Who's helping whom? Layers of culture and workplace behavior. Journal of Organizational Behavior: The International Journal of Industrial, Occupational and Organizational Psychology and Behavior, 23(4), 345-361.
- Perlow, L. A. (2012). Sleeping with your smartphone: How to break the 24/7 habit and change the way you work. Harvard Business Press.
- Perrone, V., Zaheer, A., & McEvily, B. (2003). Free to be trusted? Organizational constraints on trust in boundary spanners. *Organization Science*, *14*(4), 422-439.
- Pitichat, T. (2013). Smartphones in the workplace: Changing organizational behavior, transforming the future. LUX: A Journal of Transdisciplinary Writing and Research from Claremont Graduate University, 3(1), 13. https://doi.org/10.5642/lux.201303.13
- Podsakoff, P. M., MacKenzie, S. B., Lee, J. Y., & Podsakoff, N. P. (2003). Common method biases in behavioral research: a critical review of the literature and recommended remedies. *Journal of applied psychology*, 88(5), 879.
- Porter, M. E. (2008). Competitive strategy: Techniques for analyzing industries and competitors. Simon and Schuster.
- Putnam, L. L., Myers, K. K., & Gailliard, B. M. (2014). Examining the tensions in workplace flexibility and exploring options for new directions. *Human Relations*, 67, 413–440. https://doi.org/10.1177/0018726713495704
- Putnam LL, Fairhurst GT, and Banghart S (2016) Contradictions, dialectics, and paradoxes in organizations: A constitutive approach. Academy of Management Annals 10(1): 65–171.
- Rahmani, A. M., Babaei, Z., & Souri, A. (2021). Event-driven IoT architecture for data analysis of reliable healthcare application using complex event processing. *Cluster Computing*, 24(2), 1347-1360.
- Ragsdale, J. M., & Hoover, C. S. (2016). Cell phones during nonwork time: A source of job demands and resources. *Computers in Human Behavior*, 57, 54-60.

- Ratna, R., & Kaur, T. (2016). The impact of information technology on job related factors like health and safety, job satisfaction, performance, productivity, and work-life balance. *Journal of Business & Financial Affairs*, 5(171), 2167-0234.
- Ravand, H., & Baghaei, P. (2016). Partial least squares structural equation modeling with R. Practical Assessment, Research & Evaluation, 21(11), 1-16.
- Rice, R. E., Hagen, I., & Zamanzadeh, N. (2018). Media Mastery: Paradoxes in College Students' Use of Computers and Mobile Phones. *American Behavioral Scientist*, 62(9), 1229-1250.
- Rice, R. E., & Leonardi, P. M. (2013). Information and communication technology use in organizations. In L. P. & D. K. Mumby (Ed.), *The SAGE handbook of organizational communication* (pp. 425–448).
- Richardson, H. A., Vandenberg, R. J., Blum, T. C., & Roman, P. M. (2002). Does decentralization make a difference for the organization? An examination of the boundary conditions circumbscribing decentralized decision-making and organizational financial performance. *Journal of Management*, 28(2), 217-244.
- Rigdon, E. E. (2016). Choosing PLS path modeling as analytical method in European management research: A realist perspective. *European Management Journal*, 34(6), 598-605.
- Ringel, M. (2021). Smart City Design Differences: Insights from Decision-Makers in Germany and the Middle East/North-Africa Region. Sustainability, 13(4), 2143.
- Ringle, C. M., Sarstedt, M., & Straub, D. W. (2012). Editor's comments: a critical look at the use of PLS-SEM in" MIS Quarterly". *MIS quarterly*, 1-18.
- Rizzo, J. R., House, R. J., & Lirtzman, S. I. (1970). Role conflict and ambiguity in complex organizations. *Administrative science quarterly*, 150-163.
- Robert, C., & Casella, G. (2013). *Monte Carlo statistical methods. Springer texts in statistics*. Springer Science & Business Media.
- Robinson, K. A., Saldanha, I. J., & McKoy, N. A. (2011). Development of a framework to identify research gaps from systematic reviews. *Journal of Clinical Epidemiology*, 64(12), 1325-1330. https://doi.org/10.1016/j.jclinepi.2011.06.009
- Rosenberger, R. (2008). Perceiving other planets: Bodily experience, interpretation, and the mars orbiter camera. *Human Studies*, *31*(1), 63–75. https://doi.org/10.1007/s10746-007-9078-1

- Rosenberger, R., & Verbeek, P. (2015). A field guide to postphenomenology. In Postphenomenological Investigations: Essays on Human-Technology Relations (pp. 9–41).
- Rouibah, K., & Hamdy, H. (2009). Factors affecting information communication technologies usage and satisfaction: Perspective from instant messaging in Kuwait. Journal of Global Information Management, 17(2), 1–29. https://doi.org/10.4018/jgim.2009040101
- Rosso, B. D. (2014). Creativity and constraints: Exploring the role of constraints in the creative processes of research and development teams. *Organization Studies*, 35(4), 551-585.
- Rousseau, D. M., & Fried, Y. (2001). Location, location, location: Contextualizing organizational research. Journal of Organizational Behavior: The International Journal of Industrial, Occupational and Organizational Psychology and Behavior, 22(1), 1-13.
- Rusu, G., & Avasilcai, S. (2015). Organizational Context–Relationships with Human Resources Motivation and Work Performance. *Journal of Inventics*, *18*(1), 88.
- Sekaran, U., & Bougie, R. (2016). Research methods for business: A skill building approach. (7th ed.). Haddington: John Wiley & Sons
- Saragih, S. (2015). The effects of job autonomy on work outcomes: Self efficacy as an intervening variable. *International Research Journal of Business Studies*, 4(3), 23-34.
- Saunders, M., Lewis, P., Thornhill, A. and Wilson, J. (2009). Business research methods. *Financial Times, London*.
- Saunders, M. N., Lewis, P., and Adrian, T. (2016). *Research Methods for Business Students* (7th ed.). Harlow: Pearson Education Limited.
- Sarstedt, M., & Mooi, E. (2014). A concise guide to market research: The process, data, and methods using IBM SPSS statistics.2nd ed. Springer.
- Saxena, S., & Al-Tamimi, T. A. S. M. (2018). Visioning "smart city" across the Gulf Cooperation Council (GCC) countries. foresight.
- Schad, J., Lewis, M. W., Raisch, S., & Smith, W. K. (2016). Paradox research in management science: Looking back to move forward. *Academy of Management Annals*, 10(1), 5-64. https://doi.org/10.1080/19416520.2016.1162422

- Scheele, C., Yu, M., & Huang, Q. (2021). Geographic context-aware text mining: enhance social media message classification for situational awareness by integrating spatial and temporal features. *International Journal of Digital Earth*, 1-23.
- Schermelleh-Engel, K., Klein, A., & Moosbrugger, H. (2017). Estimating nonlinear effects using a latent moderated structural equations approach. In *Interaction and nonlinear effects in structural equation modeling* (pp. 203-238). Routledge.
- Schaufeli, W. B., Leiter, M. P., & Maslach, C. (2009). Burnout: 35 years of research and practice. *Career development international*.
- Schippers, M. C., & Hogenes, R. (2011). Energy management of people in organizations: A review and research agenda. *Journal of Business and Psychology*, 26(2), 193-203.
- Schlachter, S., McDowall, A., Cropley, M., & Inceoglu, I. (2018). Voluntary workrelated technology use during non-work time: A narrative synthesis of empirical research and research agenda. *International Journal of Management Reviews*, 20(4), 825-846. https://doi.org/ doi:10.1111/ijmr.12165
- Schmitt, T. A. (2011). Current methodological considerations in exploratory and confirmatory factor analysis. Journal of Psychoeducational Assessment, 29(4), 304-321.
- Scornavacca, E. (2010). An investigation of the factors that influence user acceptance of mobile information systems in the workplace. PhD, Victoria University of Wellington, Wellington.
- Schroeder, R. (2010). Mobile phones and the inexorable advance of multimodal connectedness. *New Media and Society*, 12(1), 75–90. https://doi.org/10.1177/1461444809355114
- Seibert, S. E., Wang, G., & Courtright, S. H. (2011). Antecedents and consequences of psychological and team empowerment in organizations: A meta-analytic review. *Journal of applied psychology*, 96(5), 981.
- Shin, D. H. (2014). Measuring the quality of smartphones: Development of a customer satisfaction index for smart services. *International Journal of Mobile Communications*, 12(4), 311–327. https://doi.org/10.1504/IJMC.2014.063650
- Shus, Q., Tu, Q., & Wang, K. (2011). The impact of computer self-efficacy and technology dependence on computer-related technostress: A social cognitive

theory perspective. International Journal of Human-Computer Interaction, 27(10), 923-939.

- Sia, S. K., & Appu, A. V. (2015). Work autonomy and workplace creativity: Moderating role of task complexity. *Global Business Review*, *16*(5), 772-784.
- Siau, K., Sheng, H., & Nah, F. (2003). Development of a framework for trust in mobile commerce. Paper presented at the 2nd Annual Workshop on HCI Research in MIS, Seattle.
- Slife, B. D., Williams, R. N., & Williams, R. N. (1995). What's behind the research?: Discovering hidden assumptions in the behavioral sciences. Sage.
- Smith, W., Erez, M., Jarvenpaa, S., Lewis, M. W., & Tracey, P. (2017). Adding complexity to theories of paradox, tensions, and dualities of innovation and change: Introduction to organization studies special issue on paradox, tensions, and dualities of innovation and change. *Organization Studies*, 38(3-4), 303-317. https://doi.org/10.1177/0170840617693560
- Smith, W. K., & Tushman, M. L. (2005). Managing strategic contradictions: A top management model for managing innovation streams. *Organization Science*, 16(5), 522-536. https://doi.org/10.1287/orsc.1050.0134
- Smith, W., & Lewis, M. (2011). Toward a theory of paradox: A dynamic equilibrium model of organizing. *Academy of Management Review*, 36(2), 381-403.
 Retrieved from www.scopus.com
- Smith, J. A., & Osborn, M. (2003). Interpretative phenomenological analysis. In J. A. Smith (Ed.), *Qualitative Psychology: A Practical Guide to Methods*. London: Sage.
- Sonnentag, S., Volmer, J., & Spychala, A. (2008). *Job Performance*. The Sage handbook of organizational behavior, (1), 427-447.
- Sousa, V. D., & Rojjanasrirat, W. (2011). Translation, adaptation and validation of instruments or scales for use in cross-cultural health care research: a clear and user-friendly guideline. *Journal of evaluation in clinical practice*, 17(2), 268-274.
- Sowon, K., Sikhwari, O., Nyemba-Mudenda, M., & Chigona, W. (2019). Smartphone paradoxes in working mothers' pursuit of work-life balance. *Communications in Computer and Information Science*, 933, 94–106. https://doi.org/10.1007/978-3-030-11235-6_7

- Stemler, S. (2001). An overview of content analysis, Practical Assessment, Research & Evaluation, 7. 137-146.
- Stephens, K. K. (2012). Multiple Conversations During Organizational Meetings Development of the Multicommunicating Scale. *Management Communication Quarterly*, 26(2), 195-223.
- Stevens, J. (1996). *Applied multivariate statistics for the social sciences*. New Jersey: Lawrence Erlbaum Associates.
- Stewart, G. L., & Barrick, M. R. (2000). Team structure and performance: Assessing the mediating role of intrateam process and the moderating role of task type. Academy of management Journal, 43(2), 135-148.
- Straub, D. W. (1989). Validating instruments in MIS research. *MIS Quarterly*, 13(2), 147-169.
- Straub, D., Boudreau, M. C. and Gefen, D. (2004). Validation Guidelines for Information Science. *Positivist Research Communications of the AIS*, 13, 380-427.
- Sutton, R. I., & Hargadon, A. (1996). Brainstorming groups in context: Effectiveness in a product design firm. *Administrative science quarterly*, 685-718.
- Sykes, T. A., Venkatesh, V., & Gosain, S. (2009). Model of acceptance with peer support: A social network perspective to understand employees' system use. *MIS quarterly*, 371-393.
- Tabachnick, B. G., and Fidell, L. S. (2007). Using multivariate statistics (5th ed.). Massachusetts, USA: Allyn & Bacon, Inc.
- Taboli, H., & Rahmani, M. (2014). Identify the relationship between the dimensions of organizational structure and organizational commitment. *Indian Journal of Fundamental and Applied Life Sciences*. 4 (54) 565, 573.
- Tarafdar, M., Tu, Q., Ragu-Nathan, B. S., & Ragu-Nathan, T. (2007). The impact of technostress on role stress and productivity. *Journal of Management Information Systems*, 24(1), 301-328.
- Tashakkori, A., & Teddlie, C. (2003). Issues and dilemmas in teaching research methods courses in social and behavioural sciences: US perspective. *International journal of social research methodology*, 6(1), 61-77.
- Tashakkori, A., & Teddlie, C. (2010). Sage handbook of mixed methods in social & behavioral research. California: Sage Publications, Incorporated.

- Taylor, S., & Todd, P. A. (1995). Understanding information technology usage: A test of competing models. *Information systems research*, *6*(2), 144-176.
- Ter Hoeven, C. L., van Zoonen, W., & Fonner, K. L. (2016). The practical paradox of technology: The influence of communication technology use on employee burnout and engagement. *Communication Monographs*, 83(2), 239-263. https://doi.org/10.1080/03637751.2015.1133920
- Tetik, N. (2016). The effects of psychological empowerment on job satisfaction and job performance of tourist guides. *International Journal of Academic Research in Business and Social Sciences*, 6(2), 221-239.
- Thornhill, S. (2006). Knowledge, innovation and firm performance in high-and low-technology regimes. *Journal of business venturing*, *21*(5), 687-703.
- Thornton, B., Faires, A., Robbins, M., & Rollins, E. (2014). The mere presence of a cell phone may be distracting: Implications for attention and task performance. *Social Psychology*,45 (6), 479–88.
- Torkzadeh, G., & Doll, W. J. (1999). The development of a tool for measuring the perceived impact of information technology on work. *Omega*, 27(3), 327-339.
- Tornatzky, L. G., Fleischer, M., & Chakrabarti, A. K. (1990). *Processes of technological innovation*. Lexington books.
- Tracy, S. J. (2004). Dialectic, contradiction, or double bind? Analyzing and theorizing employee reactions to organizational tension. *Journal of Applied Communication Research*, 32(2), 119-146.
- Tsai, C. C., & Chai, C. S. (2012). The" third"-order barrier for technology-integration instruction: Implications for teacher education. *Australasian Journal of Educational Technology*, 28(6).
- Tsoukas, H. & Hatch, M. J. (2001). Complex thinking, complex practice: The case for a narrative approach to organizational complexity. *Human relations*, 54(8), 979-1013.
- Tukey, J. W. (1977). Exploratory data analysis, MA: Addison-Wesley.
- Van Ark, B. (2016). The productivity paradox of the new digital economy. *International Productivity Monitor*, (31), 3-18.
- Van Deursen, A. J., Bolle, C. L., Hegner, S. M., & Kommers, P. A. (2015). Modeling habitual and addictive smartphone behavior: The role of smartphone usage

types, emotional intelligence, social stress, self-regulation, age, and gender. *Computers In Human Behavior*, 45, 411-420.

- Van de Ven, A.H. Angle, H., & Poole, M.S. (1989). *Research on the management of innovation:* New York: HarperCollins, 637-662.
- Van Laethem, M., Van Vianen, A. E., & Derks, D. (2018). Daily fluctuations in smartphone use, psychological detachment, and work engagement: The role of workplace telepressure. Frontiers in psychology, 9, 1808. https://doi.org/10.3389/fpsyg.2018.01808
- Van Zoonen, W., & Rice, R. E. (2017). Paradoxical implications of personal social media use for work. *New Technology, Work and Employment, 32*(3), 228-246. https://doi.org/10.1111/ntwe.12098
- Vaismoradi, M., & Snelgrove, S. (2019). Theme in qualitative content analysis and thematic analysis. In Forum Qualitative Sozialforschung/Forum: Qualitative Social Research, 20(3), 55-64.
- Velicer, W., & Fava, J. (1998). Affects of variable and subject sampling on factor pattern recovery. Psychological Methods, 3(2), 231.
- Verbeek, P. P. (2008). Obstetric ultrasound and the technological mediation of morality: A postphenomenological analysis. *Human Studies*, 31(1), 11-26.
- Verbeek, P.-P. (2016). Toward a theory of technological mediation. A program for postphenomenological research. In J. K. Berg & C. Robert (Eds.), *Technoscience* and Postphenomenology. The Manhattan Papers (pp. 1–189). Lexington Books.
- Verkasalo, H., & Salmeron, B. J. (2009, August). Analysis of the contextual behaviour of mobile subscribers. In *International Conference on Communications Infrastructure. Systems and Applications in Europe* (pp. 252-266). Springer, Berlin, Heidelberg.
- Venkatesh, V., Bala, H., & Sykes, T. A. (2010). Impacts of information and communication technology implementations on employees' jobs in service organizations in India: A multi-method longitudinal field study. *Production and Operations Management*, 19(5), 591-613.
- Venkatesh, V., Brown, S., & Bala, H. (2013). Bridging the qualitative-quantitative divide: Guidelines for conducting mixed methods research in information systems. MIS Quarterly, 37(1), 21-54.

- Venkatesh, V., Morris, M. G., Davis, G. B., & Davis, F. D. (2003). User acceptance of information technology: Toward a unified view. *MIS quarterly*, 425-478.
- Vince, R., & Broussine, M. (1996). Paradox, defense and attachment: Accessing and working with emotions and relations underlying organizational change. *Organization Studies*, 17(1), 1-21.
- Vinzi, V. E., Chin, W. W., Henseler, J., & Wang, H. (2010). Handbook of partial least squares (Vol. 201, No. 0). Berlin: Springer.
- Wall, T. D., Wood, S. J., & Leach, D. J. (2004). Empowerment and performance. *International review of industrial and organizational* psychology, 19, 1-46.
- Watson-Manheim, M. B., Chudoba, K. M., & Crowston, K. (2012). Perceived discontinuities and constructed continuities in virtual work. *Information systems journal*, 22(1), 29-52. Waxin, M. F., & Bateman, R. (2016). Labour localisation and HRM practices in the Gulf countries. In P. Budhwar, & K. Mellahi (Eds.). Handbook of human resource management in the Middle East: 284–308. Cheltenham: Edward Elgar.
- W. E., & Jung, J. Y. (2001). Exploring the digital divide internet connectedness and age. *Communication Research*, 28(4), 536-562.
- Wessels, C. (2017). Flexible Working Practices: How Employees Can Reap the Benefits for Engagement and Performance (No. EPS-2017-418-LIS).
- Williams, D., Coles, L., Wilson, K., Richardson, A., & Tuson, J. (2000). Teachers and ICT: current use and future needs. *British Journal of Educational Technology*, 31(4), 307–320.
- Williams, K. D. & Karau, S. J. (1991) 'Social loafing and social compensation: The effects of expectations of co-worker performance', *Journal of Personality and Social Psychology*, 61, 570-581.
- Wilmer, H. H., Sherman, L. E., & Chein, J. M. (2017). Smart phones and cognition: A review of research exploring the links between mobile technology habits and cognitive functioning. *Frontiers in psychology*, *8*, 605.
- Wong, I. A., Ruan, W. J., Cai, X., & Huang, G. I. (2021). Green-Induced tourist equity: the cross-level effect of regional environmental performance. *Journal of Sustainable Tourism*, 29(7), 1043-1062.

- Xiao, M., Meredith, R., & Gao, S. (2016). An investigation of how and why managers use tablets to support decision making, Proceedings of Australasian Conference on Information Systems, Wollongong.
- Xie, Y. (2011). Values and limitations of statistical models. *Research in social stratification and mobility*, 29(3), 343-349.
- Yoo, W., Mayberry, R., Bae, S., Singh, K., He, Q. P., & Lillard Jr, J. W. (2014). A study of effects of multicollinearity in the multivariable analysis. International journal of applied science and technology, 4(5), 9.
- Yun, H., Kettinger, W. J., & Lee, C. C. (2012). A new open door: The smartphone's impact on work-to-life conflict, stress, and resistance. *International Journal of Electronic Commerce*, 16(4), 121–151.
- Yunus, S., & Mostafa, A. M. S. (2021). Flexible working practices and job-related anxiety: Examining the roles of trust in management and job autonomy. *Economic and Industrial Democracy*, 1–29. https://doi.org/10.1177/0143831X21995259
- Yuvaraj, S. & Nadheya, R. (2018). A study on the role of technology on employee behaviour and their performance, *International Journal of Mechanical Engineering and Technology*, 9(7), 244-251.
- Yvonne Feilzer, M. (2010). Doing mixed methods research pragmatically: Implications for the rediscovery of pragmatism as a research paradigm. *Journal* of mixed methods research, 4(1), 6-16.
- Zhu, K., Kraemer, K. L., & Dedrick, J. (2004). Information technology payoff in ebusiness environments: An international perspective on value creation of ebusiness in the financial services industry. *Journal of management information systems*, 21(1), 17-54.
- Zhu, K., & Kraemer, K. L. (2005). Post-adoption variations in usage and value of ebusiness by organizations: Cross-country evidence from the retail industry. *Information Systems Research*, 16(1), 61-84. https://doi.org/10.1287/isre.1050.0045
- Zhuang, W., Hsu, M. K., Brewer, K. L., & Xiao, Q. (2012). Paradoxes of social networking sites: an empirical analysis. *Management Research Review*, 36(1), 33-49.
Appendix A Back-To-Back Translation





الجدول 3-2 البنود المدرجة في الاستبيان ومصادرها



حولي - شارع المثنى - مقابل المعهد السالي للخدمات الإدارية (المركز التجلري تريتا) Hawalli - Al Muthana St. - Opposite to High Institute for Administrative Services (Formerly Faculty of Business Studies)

www.ibnsina-translation.com



تعتمدون لدى جميع الوزارات والسفارات بدولة الكويت معتمدون لدى جميع الوزارات والسفارات بدولة الكويت IBN SINA FOR CERTIFIED TRANSLATION CERTIFIED TO ALL MINISTRIES AND EMBASSIES IN THE STATE OF KUWAIT

أحمل دائمًا هاتفي الذكي الخاص معي في كل مكان أذهب	EEP 2	إشكالية التمكين –
إليه للبقاء على اتصال بعملي.		الاستعباد
غالبًا ما أتحقق من هاتفي الذكي الخاص قبل البدء في مهمة	EEP 3	(EEP)
هامة.		
أشعر بالقدرة علمي أداء مسؤوليات وظيفتي أثناء استخدام	EEP 4	
هاتفي الذكي الخاص.		
يزبد هاتفي الذكي الخاص من استقلاليتي ومرونتي في	EEP 5	
العمل.		
أشعر بالالتزام بالرد على الفور على المكالمات الهاتفية أو	EEP 6	
الرسائل من العمل.		
لـن أتخلـى أبـدًا عـن اسـتخدام هـاتفي الـذكي الخـاص بالعمـل،	IDP 1	
حتى لو تأثرت حياتي اليومية به كثيرًا.		
أنا غير قادر على القيام بالمهام اليومية للعمل بدون هاتفي	IDP 2	إشكالية مستقل-
الـذكي الخـاص، حيـث يـتم حفـظ جميـع الجـداول والعناصـر		غير مستقل
الشخصية عليه.		(IDP)
يساعدني هاتفي الذكي الخاص على أداء المهام اليومية	IDP 3	
ىشكل مستقل.		
أنا قادر على القيام بمهامي اليومية للعمل بدون هاتفي	IDP 4	
الذكي الخاص.		
أشعر بالاستقلالية عند استخدام هاتفي الذكي الخاص.	IDP 5	





حولي - شارع المثنى - مقابل المعهد الصالي للخدمات الإدارية (المعهد التجاري سابقاً) Hawalli - Al Muthana St. - Opposite to High Institute for Administrative Servies (Formerly Faculty of Business Studies)

www.ibnsina-translation.com





قعتمدون لدى جميع الوزارات والسفارات بدولة الكويت IBN SINA FOR CERTIFIED TRANSLATION CERTIFIED TO ALL MINISTRIES AND EMDASSIES IN THE STATE OF KUWAIT

الجدول 3-2 البنود المدرجة في الاستبيان ومصادرها (يُتبع)

المراجع	العناصر المقتبسة من المؤلفات	الكود	التعريف الإجرائي	البعد
كويمانز وآخرون. (2013)، يـون وآخرون. (2012) تارف دار وآخرون.	يساعدني هاتفي الذكي الخاص في أداء مهامي اليومية في العمل بأقل وقت وجهد. يعما هاتف الخاص	TSP1	قد يـؤدي اسـتخدام الهواتـف الذكيـة الخاصـة لأداء المهـام المسـندة إلـى تحسـين جـودة المهـام المنجـزة فـي العمـل مـن خـلال تعظـيم المرونـة فـي التعامـل مـع المهـام المختلفة وتعلـم	أداء المهام (TSP)
(2007)	يعت مصمي المصامع على تحسين جودة المهام التي أكملها في العمل.	1012	الأساليب الحديثة المفيدة للعمل، إلا أنه، في نفس الوقت، يمكن أن يعيق إكمــال المهمــة بســبب الانقطاعــات المحمة .	
ئورنتــون وآخــرون. (2014)	يـــؤدي مجــرد وجــود هــاتفي الــذكي الخــاص إلـــى انخفـاض قــدرتي علـى التركيـز علـى أداء المهام.	TSP3	المتحررة.	
بيانٽسي وفيليسبس (2005)	يزيد هاتفي الخاص من مرونتي في التعامل مع المهام المختلفة.	TSP4		
كوپمــانز وأخــرون. (2013)	يتطلب إكمال مهام عملي من خلال هاتفي الذكي الخاص وقتًا أكثر من المعتاد.	TSP5		



ibn _{OUN}





معتمدون لدى كافة الجهات الحكومية والسفارات في دولة الكويت

for Translation & Typing

Certified from all Ministries & Embassies in Kuwait

The dimension	Code	Items excerted from the literature
	TCX 1	My job always requires the support of my smartphone.
	TCX 2	I often need to use my own smartphone to meet work commitments.
Technological	TCX 3	I spend my entire working day connected "through my own smartphone".
Context (TCX)	TCX 4	During a normal work day, all my work related communications are through my own smartphone.
	TCX 5	I often need to use my own smartphone to keep in touch with my work.
	TCX 6	I find myself busy on my smartphone "online" when I should be doing my work.
	ocx 1	Top management encourages the use of private smartphones at work.
	ocx 2	The organization's strategy encourages the use of private smartphones at work.
Organizational Context (OCX)	ocx 3	The organization's policies encourage the use of private smartphones at work.
x/	ocx 4	Tactical plans (implementation of work plans) encourage the use of private smartphones at work.
	OCX 5	It is common in our organization to use private smartphones for the benefit of work.
	EEP 1	It's hard for me to turn off my smartphone during working hours.
The problem of empowerment — enslavement	EEP 2	I always carry my own smartphone with me everywhere I go to keep in touch with my work.
(EEP)	EEP 3	I often check my own smartphone before starting an important task.
📞 22629975 - 60020764 🕓 51256426		

ibnkhaldoun.trans@gmail.com

9 Hawatil, Ibn Khaldoun Street, Sinyar Complex,Opposite Cairo Complex

حولي - شـارع ابن خـلدون - مجمع ستيارالتجاري - مقابل مجمع القاهرة Whotholdown20 🛐 ibntholdownfrom ۞@bnthaldown_print_translation



	1	bn oun تابن الترجة والطباعة
Kh for Transla	aldo	د معتمدون لدى كافة الجهات المكومية والسفارات في دولة الكويت ping
Certified from all Ministries &	Embassies	s in Kuwait
	EEP 4	I feel able to perform my job responsibilities while using my own smartphone.
	EEP 5	My own smartphone increases my independence and flexibility at work.
	EEP 6	I feel obligated to respond promptly to phone calls or messages from work.
	IDP 1	I would never give up using my smartphone for work, even if my daily life affected it a lot.
The problem of independent –	IDP 2	I am unable to do daily work tasks without my own smartphone, as all schedules and personal items are saved on it.
noni- independent (IDP)	IDP 3	My own smartphone helps me perform daily tasks independently.
	IDP 4	l am able to do my daily work tasks without my own smartphone.
	IDP 5	I feel independent when using my own smartphone.
	TSPI	My own smartphone helps me perform my daily work tasks with minimal time and effort.
	TSP2	My own smartphone improves the quality of the tasks I complete at work.
Task Performance (TSP)	TSP3	Just having my own smartphone reduces my ability to focus on performing tasks.
(1317	TSP4	My own smartphone increases my flexibility in dealing with various tasks.
	TSP5	It takes more time than usual to complete my work tasks through my own smartphone.

22629975 - 60020764 🕥 51256426

ibnkhaldoun.trans@gmail.com

9 Hawal II, Ibn Khaldoun Street, Sinyar Complex, Opposite Cairo Complex

حولي - شــارع ابن خـلدون - مجمع سنيارا لتجاري - مقابل مجمع القاهرة المنابع المالي (@ibnthaldown_print_translation) مجمع القاهرة (Dinthaldown_print_translation)



Survey Questionnaire

THE ROLE OF TECHNOLOGY PARADOX ON THE RELATIONSHIP BETWEEN CONTEXTUAL FACTORS AND TASK PERFORMANCE

Dear Participant,

Thank you for taking part in this research.

A- Research Background

A Smartphone has become an indispensable part of contemporary workers' lives. It grants them the freedom to work whenever and wherever they prefer, at the same time, takes away such freedom by increasing their connectivity to work. Thus, Technology is a double-edged sword, and Smartphones, in particular, incorporate various technologies (e.g., *applications – email, social media, and other online tools*) in a single device, are designed to be carried 24/7, and support their users in various ways. Therefore, users are more susceptible to encountering the *positive* and *negative* influences of this device "*simultaneously*" than with any other form of technology. In this research, we refer to this phenomenon as a *Technology Paradox*.

This survey questionnaire seeks to find the influence of *Technology paradox* on *Task performance* of employees in various industries in Kuwait.

B- What is required

The first part – *demographic details*, provides us your background, please place a *tick mark* ($\sqrt{}$) (manual form) or *activate the radio button* (electronic form). In the second part, circle ONE response for each item about your experience in using your smartphone for work. Some questions might sound repetitive- they are supposed to. Please answer all questions.

C- The benefits

Your answers are valuable to us, as it may lead to the refinements in Smartphone usage and introduction of more training programs by various organizations in Kuwait in connection to the areas where improvements are required to promote Smartphone usage, if necessary. Thank you in advance

Maria Alhadad Ph.D. Candidate Azman Hashim International Business School (AZIBS) Universiti Teknologi Malaysia (UTM) Kuala Lumpur, Malaysia

	2200			-	Part one:	Personal I	Petails			
2				80.	Gender					
Dea	r Part	icipa	nt, Th	nank			Male		Female	
you	for ta	king	part i	in	Answer		0		0	
this	resear	ch. T	This							
ques	stionn	aire	conta	ins						
ques	stions	abou	it you	r	Age					
expe	erienco	e wit	h usir	ng		20-29	30-39	40-49	50 and above	d e
you	r priva	ite sr	nartp	hone	Answer	0	0	0	0	
for v	work.	Som	e ques	stions		0	0	0	0	
migl	ht sou	nd re	epetit	ive,						
they	are su	лрро	sed to).	Educatio	nal Qualifi	cation			
Plea	ise ans	wer	all			Diploma	Bachelor	's Master	's Ph.D	
ques	stions.									
Please I	Ignore this o	uestionn	aire if you	do not (🧷	Answer	0	0	0	0	
Sector Typ	x	Public	Pr	rivate	Part two:U	Ising Priv	ate Sma	rtphone	for work	×
Sector Typ	x	Public	Pr	rivate	Part two:U My job alv private sm	lsing Priv vays requ	ate Sma iires the e.	rtphone support	for work	×
Sector Typ	×	Public	Pr	rivate	Part two:U My job alv private sm	Ising Priv vays requ aartphon Strongly lisagree	ate Sma iires the e. Visagree	rtphone support Neither agree nor disagree	for work of my Agree	*
Answer Which Ind	xe lustry do ye	Public O	Pr n?	rivate	Part two:U My job alv private sm d Answer	Ising Priv vays requ hartphon Strongly Lisagree	ate Sma nires the e. Disagree	rtphone support agree nor disagree	for work of my Agree	×
Answer Which Ind	xe lustry do ye Education &Training	Public O ou work i	Pr n? Banking, Finance & Insurance	Health Care	Part two:U My job alv private sm	Ising Priv vays requ hartphon Strongly G	ate Sma iires the e. Disagree	rtphone support agree nor disagree	for work of my Agree	×
Answer Which Ind	tustry do yo Education &Training	Public O uu work i IT	Pr Banking, Finance & Insurance	Health Care	Part two:U My job alv private sm 3 4 Answer I frequent smartpho	Ising Priv vays requ hartphon Strongly Lisagree O	ate Sma iires the e. Disagree O use my et work	rtphone support agree nor disagree O	for work of my Agree	*
Sector Typ Answer Which Ind Answer	tustry do ye Education &Training	Public O nu work i IT O	Pr Banking, Finance & Insurance	Health Care	Part two:U My job alv private sm 3 4 4 4 4 4 4 5 4 5 6 6 1 1 frequent smartpho	Ising Priv vays requ hartphon Strongly c O ly need to ne to me Strongly c lisagree c	ate Sma nires the e. Disagree O use my et work	rtphone support nor disagree O y private obligatic Neither agree nor disagree	for work of my Agree ³ O ms.	× 0
Sector Typ Answer Which Ind Answer	erience 2 to 5 Years	Public O u work i IT O S to 10 Years	Pr Banking, Finance & Insurance O	rivate O Health Care O	Part two:U My job alv private sm d Answer	Ising Priv vays requ hartphon Strongly c O ly need to ne to me Strongly c O	ate Sma iires the e. Disagree O Duse my et work	rtphone support agree nor disagree O y private obligatic Neither agree nor disagree	for work of my Agree ⁸ Ons. Agree ⁹	ik a
Sector Typ Answer Which Ind Answer Work Expe	the hustry do yet Education &Training O erience 2 to 5 Years O	Public O nu work i IT O S to 10 Years O	Pr Banking, Finance & Insurance O 15 to 20 years	Health Care O more than 20 years O	Part two:U My job alw private sm d Answer I frequent smartpho 3 4 4 5 6 6 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	Ising Priv vays requ hartphon Strongly c O ly need to ne to me Strongly c O uy entire v hy private	ate Sma iires the e. Disagree O use my et work Disagree O vorkday e smartp	rtphone support Neither agree nor disagree O y private obligatio Neither agree nor disagree O connect thone.	for work of my Agree ⁵ O ms. Agree ⁵ O	k s

Appendix B Pilot Test Survey Questionnaire

smartr	unications	through	my priva	te		the use	e of privat	e smartp	hones fo	r work
office c	ohone are Strongly	work-rela	nted. Neither agree	Agree	ş		Strongl disagre	/ Disagree	Neither agree nor disagree	Agre
	disagree	0	disagree	0		Answe	r O	0	0	С
Answe	ar O	0	0							
l frequ	ently need	i to use m	ny private	e h my		The po the use	licies of t e of privat	he organ te smartp	ization er hones fo	r wor
work.	Strongly	ay conne	Neither		0		Strong) disagre	/ Disagree	Neither agree nor disagree	Agn
	disagree	Disagree	nor disagree	Agree		Answe	r O	0	0	С
Answe	er O	0	0	0						
The se use of	nior mana private sn	gement e nartphone	encourag es for wo	es the irk.		The tar plans) smartp	ctical plar encourag hones fo	is (implen e the use r work.	nentatior of privat Neither	n of w te
	Strongly	Disagree	Neither agree	Agree	s		Strongl disagre	^y Disagree	e agree nor disagree	Agr
lt is con private	nmon in oi smartpho	ur organiz	zation to	use		l often cl starting i	neck my p an import	orivate sr ant task	nartphor	ne be
It is con private	nmon in oi smartpho Strongly disagree	ur organiz nes for w Disagree	zation to ork. Neither agree nor disagree	use Agree	S	l often cl starting a	neck my p an import Strongly disagree	orivate sr ant task Disagree	Neither agree nor disagree	ne be Agre
It is con private Answer	nmon in oi smartpho Strongly disagree	ur organiz nes for w Disagree	zation to ork. Neither agree nor disagree	use Agree	s	l often cl starting a Answer	Strongly disagree	orivate sr ant task Disagree	Neither agree nor disagree	Agree
It is con private Answer It is hare	nmon in or smartphor disagree O d for me to hone off d Strongly disagree	ur organiz nes for w Disagree O o turn my luring wo	zation to ork. Neither agree nor disagree C Private rk hours. Neither agree nor disagree	Agree O Agree	8	I often cl starting a Answer I feel em responsi smartph	beck my p an import Strongly disagree O powered bilities who one. Strongly disagree	Disagree	Neither agree nor disagree O rm my jol my priva Neither agree nor	Agre
It is con private : Answer It is hard smartpl	Strongly disagree	ur organiz nes for w Disagree O turn my luring wo Disagree	zation to ork. Neither agree nor disagree C Private rk hours. Neither agree nor disagree	use Agree Agree	\$	I often cl starting a Answer I feel em responsi smartph	powered bilities wi disagree	Disagree	Neither agree nor disagree O rm my jol my prive Neither agree nor disagree	Agre O Agre O Agre
It is con private : Answer It is hard smartpl Answer	nmon in or smartphoi Strongly disagree O d for me to hone off d Strongly disagree O shave my p rywhere I k.	ur organiz nes for w Disagree O Disagree O Disagree O private sr go to stag	zation to ork. Neither agree nor disagree Private rk hours. Neither agree nor disagree O martphor y connect	Agree	s	Answer I feel em responsi smartph Answer My priva autonom	beck my p an import Strongly disagree O powered bilities wh one. Strongly disagree O te smartp y and fle:	brivate sr cant task Disagree O to perfor nile using Disagree O shone inc kibility at	Neither agree nor disagree O rm my jol my prive Neither agree nor disagree O	Agree

to phone calls or messages from work.	perform everyday tasks independently.
Neither Strongly Disagree Agree S disagree disagree	Neither Strongly Disagree agree Agre disagree disagree disagree
Answer O O O	Answer O O O
I will never give up using my private smartphone for work, even if my daily life is greatly affected by it. Strongly Disagree Agree S nor disagree	I am capable of doing my everyday tasks work without my private smartphone. Strongly Disagree Agree nor disagree Answer
Answer () () () ()	
I am incapable of doing everyday tasks for work without my private smartphone, as all schedules and personal items are saved on it. Neither	l feel independent when using my privat smartphone. Strongly Disagree agree Agree disagree disagree
disagree Disagree Agree Agree disagree	Answer O O O
My private smartphone helps me to me perform my everyday tasks at work with minimal time and effort.	My private smartphone increases my flexibility in handling different tasks.
Neither Strongly Disagree agree Agree S disagree nor Agree disagree	Strongly Disagree agree Agree disagree disagree
Answer O O O	Answer O O O
My private smartphone improves the quality of my the work tasks I complete at work.	My private smartphone requires me to spend more time completeing my work tasks than I intended
Strongly Disagree Agree S disagree Disagree disagree	Neither Strongly Disagree agree disagree nor Agrei disagree
Answer O O O	Answer O O O O
The "mere presence" of my private smartphone leads to a decrease in my ability to focus on performing tasks.	Back Submit Clear This content is neither created nor endorsed by Google Abuse Terms of Service - Privacy Policy
Neither	

	القسم الأول: البيانات الشخصية
	النوع
شكراً لك على المشاركة	انثی ذکر
يتضمن هذا الاستبيان أسئلة	الاجابة
تتعلق بتجربتك الخاصة	
باستخدام هانفك الدكي الشخص العمل م أثناء فترة	
الدوام. قد تبدو بعض الأسئلة	وما 50 00 00 00 00 00 00
مكررة لكن من المفترض أن	ورق فوق
تكون كذلك. لذا يرجي	الإجابة
الإجابة على جميع الأسئلة. لا	
توجد إجابات صحيحه او خاطئة؛ ذحب مهتممن فقط	المؤهل العلمي
برايك برأيك	اخری دکتوراه ماجستیر بکالوریوس دبا
تجاهل هذا الاستبيان اذا اكنت لا تستخدم الهاتف الذكي الشخص. للعما	🔾 🔿 🤇 (الإجابة
المعاوية الما	0
عام خاص	القسم الثاني: استخدام الهاتف الذكي الشخصي أثناء الدوام
الإجابة	يرجى الإشارة إلى مدى موافقتك على العبارات التالية. لا توجد اجابات صحيحة أو خاطئة؛ نحن مهتمون فقط برأيك
قطاع العمل	
البنوكـو التعليم تون الرعاية التمويل IT أو تون الصحية و	تتطلب وظيفتي دائمًا دعم هاتفي الذكي الخاص بي بيذيبية
التمويل التدريب	لا اوافق وافق أوافق ولا أعارض شدة أوافق أعارض بشدة
	0 0 0 0
سنوات الخبرة	
و ما 20 من 15 من 5 - 10 من 2 - 5 فوق -20 سنة سنة سنوات الإجابة	كثيرًا ما أحتاج إلى استخدام هاتفي الخاص للوفاء بالتزامات العمل.
Clear selection	لا اوافق وافق أوافق و لا آعارض بشدة شدة أعراض.
	5
Page 2 of 3	0 0 0 0

في الذكي الخاص.	ن خلال هات	متصلاً مر	للي بأكمله	ي يوم عم	اقض	الإنترنت" وم بعملي	ذكي "عبر بغي أن أة	ل هاتفي ال عندما يا.	غولاً على	ىي مىتىر	جد نفس	1
	أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	افق دة		أعارض بشدة	أعارض	وافق و لا ا ارض	الا ا ، و أعا	أوافق	افق حدة
الاجابة	0	0	0	0	С	الاجابة	0	0	С)	0	C
, المرتبطة ي الخاص	ع اتصالاتي هاتفي الذكر	تکون جمب من خلال	ل العادي، بالعمل	يوم العما	خلال	خاصة في العمل.	الذكية ال	ام الهواتف	استخدا	ة العليا	ع الإدارة	تشجي
	أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	افق دة		أعارض بشدة	أعارض	وافق و لا ارض	الا ا , و أعا	أوافق	افق حة
الاجابة	0	0	0	0	С	الاجابة	0	0	С)	0	С
اص للبقاء بال بعملي	الذكي الخا .على اتص	دام هاتفي	لی استخا	ج کثیرًا إ	أحتا	نف الذكية في العمل	ندام الهوا: الخاصة.	ظمة استخ	جية المند	ستراتي	شجع ام	y
	أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	افق دة		أعارض بشدة	أعارض	وافق و لا أ ارض	الا ا ، د أعا	أوافق	افق دة
الاجابة	0	0	\bigcirc	0	C	الاجابة	0	0	С)	0	C
					0	 كية الخاصة	لهواتف الذ	استخدام ا	ىۇسستنا	ع في ہ	من الشاه	
اتف الذكي ة في العم الاجابة	ستخدام الهو الخاص. أعارض شيقة	ة المنظمة ا ق أعارض	ستراتيجية لا اواف أعارض	تشجع ا ة أوافز ()	افز د	كية الخاصة صالح العمل الاجابة	لهواتف الذ ر اعارض بشدة (استخدام ا أعارض	ىۇسستنا لا اوافق و لا أعارض	ع في ه اوافق (من الشاة افق ـدة (-
اتف الذكية ة في العمل الاجابة يية الخاصة في العمل.	ستخدام الهو الخاص. أعارض بشدة الهواتف الذك	ة المنظمة ا ق اعارض ا	ستراتيجية لا اواف أعارض اعارض	تشجع ا ة أوافز ص (افز بد تش	كية الخاصة صالح العمل الاجابة لخاص أثناء اعات العمل	لهواتف الذ ا. أعارض يشدة في الذكي ا س.	استخدام ا أعارض تشغيل هاتن	مؤسستنا لا اوافق و لا أعارض ي إيقاف ن	نع في ه أوافق صب علي	من الشائ فق يص	
اتف الذكية ة في العمل الاجابة ثية الخاصة في العمل الاجابة	ستخدام الهو الخاص. بشدة () () () الهواتف الذك المرة () () ()	ة المنظمة ا ق أعارض ف أعارض أعارض أعارض	ستراتيجية لا اواف أعارض ات المنظمة و لا أعارض أعارض	تشجع ا ة أوافز جع سياسا آوافز	افز ــــــــــــــــــــــــــــــــــــ	كية الخاصة صالح العمل الاجابة لخاص أثناء اعات العمل	لهواتف الذ اعارض بشدة أعارض بشدة أعارض بشدة	استخدام ا أعارض تشغيل هاتن أعارض	مؤسستنا و لا أعارض ي إيقاف ن و لا أعارض	ع في م أوافق عب علر أوافق	من الشائ دة صدة يصع دة ر	
اتف الذكية ة في العمل الاجابة يية الخاصة في العمل الاجابة تة في العمل	ستخدام الهو الخاص بشدة م الهواتف الذك أعارض أعارض م ينفيذ خطط تنفيذ خطط	ة المنظمة ا ق أعارض أعارض أعارض أعارض أعارض أعارض أعارض أ	ستراتيجية ق و لا أعارض أعارض ن و لا لا اواف أعارض أعارض إ الخطط ا استخدام	تشجع ا 5 أوافز جع سياسا 5 أوافز 7 أوافز	افز تد تش افز	كية الخاصة صالح العمل الاجابة باعات العمل الاجابة مكان أذهب	لهواتف الذ اعارض بشدة أعارض بشدة اعارض بشدة بقاء على ات بشرة أعارض	استخدام ا أعارض تشغيل هاتة أعارض إليه لل أعارض	سؤسستنا و لا أعارض أعارض و لا الا اوافق أعارض ني الذكي و لا لا اوافق	نع في م اوافق عب عليٰ أوافق نمّا هاتة	من الشائ افق) یصا افق أحمل دانا ی	
اتف الذكية ة في العمل الاجابة في العمل الاجابة ة في العمل	ستخدام الهو الخاص بشدة م أعارض أعارض أعارض ننفيذ خطط أعارض أعارض	ة المنظمة ا ق أعارض أعارض أعارض ليكتيكية (أ إلهواتف ال أعارض ر	ستراتيجية لا اواف أعارض أعارض ن و لا اعارض أعارض إلخطط ا استخدام و لا اواف أعارض أعارض أعارض	تشجع ال 3 أوافز جع سياسا 3 أوافز 3 أوافز 5 أوافز	اقر لند القر الفر	كية الخاصة صالح العمل الاجابة اعات العمل الاجابة صال بعملي الاجابة	لهواتف الذ أعارض بشدة أعارض بشدة أعارض بشدة أعارض بشدة أعارض	استخدام ا أعارض تشغيل هاتن أعارض أعارض أعارض	سؤسستنا بولا اعارض ي إيقاف : ولا اعارض اعارض ولا اعارض ولا	نع في م أوافق عب علي أوافق أوافق	من الشاة افق يحم افق احمل داه افق 	

	، استخدام کی الخاص	طيفتي أثناء هاتفي الذك	ؤوليات وخ	ی أداء مس	القدرة عل	أشعر با	ص بالعمل، ية به كثيرًا	الذكي الخا ياتي اليوم	ام هاتفي ا تأثرت ح	عن استخدا .حتی لو	خلی أبدًا ع	لن أت
	0	ب أعارض بشدة	أعارض	لا اوافق و لا	أوافق	ِ افق شدة		أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	ِ افق شدة
	الاجابة	0	0	اعارض	0	С	الاجابة	0	0	0	0	0
	رونتي في العمل	نلاليتي و م	ں من استة	ذكي الخام	هاتفي ال	يزيد	دون هاتفي ل والعناصر خصية عليه	ية للعمل ب يع الجداو الش-	بهام اليوم حفظ جم	, القيام بالم حيث يتم	قادر على ي الخاص،	أنا غير الذكر
		أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	ِ افق شدة		أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	ِ افق شدة
	الاجابة	0	0	0	0	C	الاجابة	0	0	0	0	С
	الهاتفية أو من العمل	المكالمات الرسائل. أعادة	الفور على	بـالرد على لا اوافق	بالالتزام	أشعر	هام اليومية كل مستقل	ی أداء المر بش.	لخاص عل	في الذكي ا لا اوافق	عدني هات	يساء
	20. 04.	بشدة	أعارض	و لا أعارض	أوافق	شدة		اعارض ہشدۃ	أعارض	و لا أعارض	أوافق	يافق شدة
	الاجابة ا	0	0	0	0	_	الاجابة	0	0	0	0	1
ي ر	بدون هاتف _ر ذكي الخاص	ية للعمل ال	مي اليوم	قيام بمها	ِ على ال	أنا قادر	ء الوظيفي	الأدا				
Ļ	بدون هاتفږ ذکي الخاص	ية للعمل الا. أعارض بشدة	مي اليوم أعارض	قیام بمها لا اوافق و لا أعارض	ِ على ال افق	أنا قادر ىق دة أو	اء الوظيفي مي اليومية وقت وجهد	الأدا ي أداء مهاه عمل بأقل ر	الخاص فر في ال	في الذكي	عدني ھات	يسا
ų U	بدون هاتفږ ذکي الخاص الاجابة	ية للعمل . اللا أعارض بشدة	مي اليوم أعارض	قیام بمها لا اوافق و لا أعارض	_ على ال افق	أنا قادر بق أو دة (اء الوظيفي مي اليومية وقت وجهد	الأدا ي أداء مهاه عمل بأقل أعارض بشدة	الخاص فو .في ال أعارض	في الذكي في الوافق و لا أعارض	عدني هات أوافق	یسا. افق شدة
	بدون هاتفږ ذکي الخاص الاجابة	ية للعمل . الا أعارض بشدة	مي اليوم أعارض	قیام بمها لا اوافق و لا أعارض	_ على ال افق (أنا قادر يق أو ^ا	اء الوظيفي مي اليومية وقت وجهد الاجابة	الأداء ي أداء مهار عمل بأقل م أعارض بشدة	الخاص فو .في ال أعارض (في الذكي في الوافق و لا أعارض	عدني هات أوافق (یسا. افق شدة ()
	بدون هاتف <u>ي</u> ذكي الخاص الاجابة	ية للعمل الا أعارض بشدة	مي اليوم أعارض	قیام بمها لا اوافق و لا أعارض	_ على ال افق (أنا قادر ق أو 5 (اء الوظيفي مي اليومية وقت وجهد الاجابة	الأداء مها، عمل بأقل ، اعارض بشدة	الخاص فر .في ال أعارض (في الذكي لا اوافق و لا أعارض	عدني هات اوافق (يسا. افق شدة
	بدون هاتفږ ذکي الخاص الاجابة ذکي الخاص	ية للعمل . الا بشدة ص هاتفي الد	مي اليوم أعارض استخدام	قيام بمها لا اوافق و لا أعارض لالية عند لا اوافق	_ على ال افق الاستقلا	أنا قادر ق أو) (اشعر ب	اء الوظيفي مي اليومية وقت وجهد الاجابة الاجابة التي أكملها .في العمل	الأداء ي أداء مهام عمل بأقل م بشدة بشدة ص	الخاص فر .في ال أعارض صين جر	في الذكي ا و لا أعارض ص	عدني هات اوافق ماتفي الخ	يسا. افق شدة يعمل ه
	بدون هاتفږ ذکي الخاص الاجابة	ية للعمل . الا أعارض بشدة ماتفي الا أعارض بشدة	مي اليوم أعارض استخدام أعارض	قيام بمها لا اوافق و لا أعارض لالية عند و لا أعارض	_ على ال افق الاستقلا افق	أنا قادر يق أو ^ا) (.أشعر ب	اء الوظيفي مي اليومية وقت وجهد الاجابة الاي أكملها .في العمل	الأداء ي أداء مها، عمل بأقل ، بشدة بشدة ودة المهام بشدة بشدة	الخاص فو .في ال أعارض حسين جو أعارض	في الذكي ا و لا اعارض ص على ت اص على ت و لا إعارض	عدني هات أوافق ماتفي الخ أوافق	يسا. شدة رافق شدة
	بدون هاتفو ذكي الخاص الاجابة ذكي الخاص	ية للعمل . الم بشدة بشدة ماتفي اللا أعارض بشدة	مي اليوم أعارض استخدام أعارض	قيام بمها لا اوافق و لا أعارض (لا اوافق و لا أعارض	_ على ال افق الاستقار افق	أنا قادر يق أو ^ا .ة .أشعر ب .ق أوا	اء الوظيفي مي اليومية وقت وجهد الاجابة الاجابة .في العمل الاجابة	الأداء عمل بأقل م بشدة أعارض ودة المهام أعارض بشدة	الخاص فو .في ال أعارض مسين جو أعارض	في الذكي ا و لا أعارض ص على ت و لا لا اوافق و لا أعارض	عدني هات أوافق ماتفي الخ أوافق	يسا. افق سعمل ه شدة افق س

ى انخفاض أداء المهام	, الخاص إل تركيز على	نفي الذكي تي على ال	. وجود ها: قدر.	دي مجرد	يۇ
	أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	<u>ا</u> فق شدة
الاجابة	0	0	0	0	0
ل مع المهام المختلفة.	في التعامإ	ن مرونتي	الخاص مر	يد ھاتفي	ين
	أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	ِ افق شدة
الاجابة	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
	11 121-	New	lue alte	11-51	Ilen
حي الحاص من المعتاد	هائفي الد وقتًا أكثر.	, من حلال	مهام عملي	ب إكمال ا	يتطل
	أعارض بشدة	أعارض	لا اوافق و لا أعارض	أوافق	<u>ِ</u> افق شدة
الاجابة	\circ	\bigcirc	\bigcirc	\bigcirc	С
_			P	age 3 of	3

Appendix C Pilot Test Results Of The Survey Questionnaire

The following pilot study results were for the two dimensions of contextual factors (CNF); technological context (TCX) and organizational context (OCX), two dimensions of technology paradox (TCP), Empowerment-enslavement paradox (EEP) and independence-dependence paradox (IDP) and task performance as a higher-order construct.

			Std.				
	N	Mean	Deviation	Skew	ness	Kurt	osis
					Std.		Std.
	Statistic	Statistic	Statistic	Statistic	Error	Statistic	Error
TCX1	30.0	4.0	0.8	0.0	0.4	-1.4	0.8
TCX2	30.0	3.7	0.9	-0.4	0.4	-0.5	0.8
TCX3	30.0	3.7	1.1	-0.2	0.4	-1.1	0.8
TCX4	30.0	4.1	0.6	-0.1	0.4	-0.4	0.8
TCX5	30.0	4.3	0.7	-0.5	0.4	-0.6	0.8
TCX6	30.0	3.6	0.9	0.3	0.4	-0.6	0.8
OCX7	30.0	4.1	0.9	-0.6	0.4	-0.4	0.8
OCX8	30.0	4.0	0.9	-0.1	0.4	-1.6	0.8
OCX9	30.0	3.9	0.6	0.1	0.4	-0.3	0.8
OCX10	30.0	3.8	0.8	0.0	0.4	-0.9	0.8
OCX11	30.0	3.6	1.0	-0.1	0.4	-1.0	0.8
EEP12	30.0	4.0	1.1	-1.3	0.4	1.7	0.8
EEP13	30.0	4.2	0.7	-0.9	0.4	1.8	0.8
EEP14	30.0	3.9	0.8	-0.3	0.4	-0.2	0.8
EEP15	30.0	3.7	0.8	-0.9	0.4	2.5	0.8
EEP16	30.0	3.4	0.9	-0.4	0.4	-0.8	0.8
EEP17	30.0	3.7	1.1	-0.9	0.4	-0.1	0.8
IDP18	30.0	3.9	0.7	0.1	0.4	-1.1	0.8
IDP19	30.0	3.9	0.6	0.0	0.4	0.2	0.8
IDP20	30.0	4.0	1.0	-0.8	0.4	-0.1	0.8
IDP21	30.0	3.7	0.8	0.3	0.4	-0.7	0.8
IDP22	30.0	4.1	0.9	-0.6	0.4	-0.4	0.8
TSP23	30.0	4.1	0.8	-0.6	0.4	0.1	0.8
TSP24	30.0	4.0	0.9	-0.6	0.4	-0.4	0.8
TSP25	30.0	4.2	0.7	-0.2	0.4	-0.8	0.8
TSP26	30.0	3.8	0.6	0.1	0.4	-0.3	0.8
TSP27	30.0	3.9	1.1	-0.6	0.4	-0.9	0.8

Table 1: Skewness and Kurtosis



Figure 1: Reflective Measurement Model Before reduction

Dimensions	Cronbach's	rho_A	CR	AVE
	Alpha			
EEP	0.724	0.724	0.816	0.432
IDP	0.411	0.411	0.684	0.313
OCX	0.781	0.781	0.853	0.541
TCX	0.617	0.617	0.776	0.431
TSP	0.838	0.838	0.886	0.609

Table 2: Reliability and Validity Before Reduction

Dimensions	EEP	IDP	OCX	TCX	TSP
EEP1	0.475				
EEP2	0.709				
EEP3	0.567				
EEP4	0.741				
EEP5	0.792				
EEP6	0.606				
IDP1		0.484			
IDP2		0.766			
IDP3		0.571			
IDP4		0.401			
IDP5		0.507			
OCX1			0.621		
OCX2			0.633		
OCX3			0.839		
OCX4			0.800		
OCX5			0.759		
TCX1				0.787	
TCX2				0.762	
TCX3				0.660	
TCX4				0.670	
TCX5				0.262	
TSP1					0.790
TSP2					0.714
TSP3					0.738
TSP4					0.846
TSP5					0.805

Table 3: Factor Loading Before Reduction

Table 4: Discriminant Validity

Dimensions	Dimensions						
	EEP	IDP	OCX	TCX	TSP		
EEP	0.657						
IDP	0.300	0.560					
OCX	0.706	0.140	0.736				
ТСХ	0.737	0.420	0.594	0.656			
TSP	0.244	0.041	0.246	0.050	0.780		



Figure 2: Reflective measurement model after reduction

Cronbach's Alpha	rho_A	CR	AVE
0.735	0.735	0.852	0.659
0.439	0.439	0.732	0.482
0.830	0.830	0.899	0.749
0.813	0.813	0.890	0.730
0.811	0.811	0.889	0.727
	Cronbach's Alpha 0.735 0.439 0.830 0.813 0.811	Cronbach's Alpha rho_A 0.735 0.735 0.439 0.439 0.830 0.830 0.813 0.813 0.811 0.811	Cronbach's Alpharho_A CR0.7350.7350.7350.7350.4390.4390.8300.8300.8130.8130.8110.8110.812

Table 5: Reliability and Validity After Reduction

Dimensions	EEP	IDP	OCX	TCX	TSP
EEP2	0.705				
EEP4	0.876				
EEP5	0.844				
IDP2		0.830			
IDP3		0.640			
IDP5		0.590			
OCX3			0.800		
OCX4			0.900		
OCX5			0.892		
TCX1				0.910	
TCX2				0.843	
TCX3				0.807	
TSP1					0.851
TSP4					0.908
TSP5					0.796

Table 6: Factor Loading After Reduction

Table 7: Discriminant Validity After Reduction

	Dimensions					
Dimensions	EEP	IDP	OCX	TCX	TSP	
EEP	0.812					
IDP	0.173	0.694				
OCX	0.594	0.069	0.865			
ТСХ	0.546	0.157	0.678	0.854		
TSP	0.223	-0.059	0.224	0.079	0.853	

Thus, eliminating the factor loadings below 0.7, the reduced questionnaire is given in Appendix D.

	Part one: Personal Details
	Gender
Dear Participant, Thank	Male Female
you for taking part in	Answer O O
this research. This questionnaire contains questions about your experience with using	Age 20-29 30-39 40-49 50 and above
your private smartphone	Answer O O O
for work. Some questions might sound repetitive, they are supposed to.	Educational Qualification
Please answer all questions. Please Ignore this questionnaire if you do not your private smartphone for work	Diploma Bachelor's Master's Ph.D
Public Private Answer O O	My job always requires the support of my private smartphone. Strongly Disagree Agree S disagree Disagree disagree
Which Industry do you work in? Banking, Education IT Finance Health &Training & Care	Answer O O O
Answer O O O	During a typical work day, all my communications through my private smartphone are work-related.
Work Experience	Strongly Disagree Agree S disagree Disagree Nor Agree
2 to 5 5 to 10 15 to 20 more than 20 Years Years years years	Answer O O O
Answer O O O O	

Appendix D The Final Survey Questionnaire

			Neither				geonon			
	Strongly disagree	Disagree	agree nor disagree	Agree	S		Strongly disagree	Disagree	Neither agree nor disagree	Agre
Answer	0	0	0	0		Answer	0	0	0	С
The stra he use Answer	tegy of th of private Strongly disagree	ne organi smartph Disagree	zation pr iones for Neither agree nor disagree	Agree	s	l feel em responsi smartph	powered bilities w one. Strongly disagree	l to perfo hile using Disagree	rm my jo g my priva Neither agree nor disagree	ib ate Agr
						Answer	0	0	0	C
l will ne smartp greatly	ever give (bhone for affected	up using r work, eve by it.	ny privati n if my d	e aily life	is -	My priva quality o work.	te smart f my the	phone im work tas	nproves t ks I comj	he plete
l will ne smartp greatly Answe	ever give of ohone for affected Strongly disagree	up using r work, eve by it. Disagree	Neither agree nor disagree	e aily life Agree	s	My priva quality o work.	te smart f my the Strongly disagree	phone im work tas Disagree	Neither agree nor disagree	he plet
I will ne smartp greatly Answe	ever give up obone for r affected Strongly disagree	up using r work, eve by it. / Disagree	ny private n if my d Neither agree nor disagree	e aily life Agree	is S	My priva quality o work. Answer	te smart f my the Strongly disagree	phone im work tas Disagree	Neither agree nor disagree	he plete Ag
I will ne smartp greatly Answe I am in work w schedu it.	ever give i ohone for affected Strongly disagree ir O capable o vithout my ules and p Strongly disagree	f doing ev private s ersonal it	ny private n if my d Neither agree nor disagree O veryday t martpho ems are s Neither agree nor disagree	e aily life Agree O asks for ne, as a saved o Agree	s	My priva quality o work. Answer The "me smartph ability to	te smart f my the Strongly disagree O re presei one lead focus of	phone im work tas Disagree O nce" of m is to a dee n perform	Neither agree nor disagree O	Ag (
I will ne smartp greatly Answe I am in work w schedu it.	ever give uphone for r affected Strongly disagree r O capable o vithout my ules and p Strongly disagree	p using r work, eve by it. Disagree of doing ev private s ersonal it	Neither agree nor disagree O veryday t martpho ems are s Neither agree nor disagree	e aily life Agree O asks for ne, as a saved o Agree	is S II n	My priva quality o work. Answer The "me smartph ability to	te smart f my the Strongly disagree O re presei one lead focus of Strongly disagree	phone im work tas Disagree O nce* of m is to a deo n perform Disagree	Neither agree nor disagree O hy private crease in ning task Neither agree nor disagree	Ag (my s.

	Strongly disagree	Disagree	Neither agree nor disagree	Agree	S
Answer	0	0	0	0	
Back	Subr	nit		Clear	forn

Appendix E Qualitative Interview Protocol

1.0			Introduction
		1.1	Thank interviewee for participation, introduce study focus
		1.2	Explain interview structure and inform the interviewee that it is recommended to share anecdotes of experiences
			of smartphone usage for work-related purposes
		1.3	Demographic details (gender, age, time in current role, previous education and work experience, experience with smartphone usage).
2.0	Research Questions		Contextual Factors (CNF)
		А.	Technological Context (TCX)
	RQ ₁ Which	2.1	Does your job always require the support of your private smartphone? Explain
	factors drive the	2.2	During a typical workday, are all types of communications through your private smartphone work-related?
	emergence of		Explain.
	the technology	2.3	Can you explain how often you are using your smartphone to meet your work obligations? Are there particular
	paradox in		times when this will fluctuate?
	mobile		How do you decide when and where to be connected through smartphone – for instance, some employees
	technology		decide to take their cell phones to meetings, lunch, everywhere – do you have similar experiences (probe
	usage based on	_	interviewee for more stories/examples)? Why do you do this?
	the experiences	В.	Organizational Context (OCX)
	of employees in	2.4	How do the policies of the organization encourage the use of private smartphones for work? Are there any
	Kuwait?		workplace policies that might restrict or guide the use of smartphones? Explain
		2.5	How does the strategy of the organization promote the use of private smartphones for work?
		2.5	Can you explain how the tactical plans (implementation of work plans) encourage the use of private smartphones for work?

3.0		Technology Paradox (TCP) and Task Performance (TSP)
	3.1	How responsive are you to communication through ICTs? For example, are you responding to emails instantly,
RQ ₂ Does the		or is there a particular way that you work?
TCP in mobile		Why do you take this approach?
technology	3.2	Do you feel an obligation to immediately respond to phone calls or messages from work?
usage have	3.3	How does the use of your private smartphone make you feel empowered or enslaved? Explain.
relationships	3.4	How does the use of your private smartphone make you feel independent or dependent? Explain.
with CNF and		How does your private smartphone help you to perform everyday tasks independently?
employee TSP?	3.5	Does your organization make you feel empowered or enslaved (in terms of using your smartphone for work)? Explain.
RQ ₃ Which specific	3.6	Does your organization make you feel independent or dependent (in terms of using your smartphone for work)? Explain
dimensions of		will you give up using your private smartphone for work, if your daily life is greatly affected by it?
CNF and the	3.7	Does the feeling of being empowered or enslaved influence your task performance? Explain.
TCP in mobile		Does your private smartphone increase your flexibility in handling different tasks?
usage are related	3.8	Does the feeling of being independent or dependent influence your task performance? Explain.
to employee		Are you capable of doing your everyday tasks for work without your private smartphone?
TSP?	3.9	What is your view about the association between the context, paradox in mind, and your task performance with specific reference to smartphone usage?
RQ ₄ Does the		If we conceptualize that there is an environment supportive to smartphone usage (<u>contextual factors</u>), and the
TCP in mobile		feeling of 'positive and negative' about the impact of the smartphone on your work (technology paradox), and
technology	4.1	then, there is an outcome to be produced by you in the form of <u>task performance</u> how do you relate these three?
usage mediate		Does your private smartphone improve the quality of the tasks you complete at work?
the influence of		
CNF on		
employee TSP?		
		CLOSURE
	5.1	Are there any other comments you would like to add to end this interview or any sections that you would like us to go back to for further discussion?
	5.2	Thank participant, and state the date of when the transcript will be sent for their review. Remind them that
		contact details are available on the information form

Appendix F Empowerment Theme

Respondents	Excerpts of the in-depth interviews Empowerment Theme
IR ₁	"Yes, we are empowered by the organization to use the information that we find most suitable in a given situation."
IR ₂	"We find empowerment to use the phones in our decision-making activities, and sometimes we find answers to the questions
	asked by our students very easily in the Apps, which not only clarifies their doubts but also, it uses creative ways to make
	the student comprehend it through audio-visual aids and registers it in their minds."
IR ₃	"The organization in which I serve is technology governed and naturally has a strong technological context. This strong
	context or background of technology empowers us to make the best use of technology to produce quality work. I know my
	powers and line of authority. I have a clear picture of the limits of my usage of resources, and I do not have to consult my
	superiors on either the resource utilization or use of manpower, and it is left to my discretion."
IR4	"There is no end for HR issues in any organization as somebody would fall sick, somebody needs leave, somebody has to
	report to the Vice-president, and there is a major crisis at times where HR intervention is a must. So, in one way, I am
	enslaved by technology, as I can be accessed from anywhere by my boss. But yes, it also empowers me to make decisions
	as I have all the data and information at my fingertips. I can call it 'a double-edged sword'."
IR5	"As I report to the top level of management at my capacity of Senior Legal Consultant, I do have my privileges, and I am
	definitely empowered in my job to make decisions at my own discretion on issues such as legal requirements in a case,
	fixation of fees for consultancy, using the company resources for executing the tasks, assuring the type of deliverables to
	the customer, deciding on the timeline and so on. My senior management has never interfered or questioned my decisions
	on these matters, and this empowerment has enabled me to perform much better."

Respondents	Organizational context (OCX) and $\rightarrow Empowerment-Enslavement Paradox (EEP)$
IR ₁	"Our organizational policies strategies and their execution are supportive to the smartphones as the enabler
нц	technology. This automatically empowers us to make use of the facilities available through smartphones to ease our
	work and the Apps mentioned before are very handy in carrying out our day to day activities. That does not mean
	work, and the Apps mentioned before the very handy in currying out our day-to-day activities. That does not mean
	indi I diways jeel that smartphones are an enabler to the execution of strategic plans of the organization, but
	sometimes, it generates a feeling as if it has made us slaves to technology. Our critical thinking abilities are being
	reduced due to the smartphone Apps, and to some extent, we are enslaved to it. It has now become a habit for me to
	look into my smartphone for a plausible solution to a problem than thinking about it myself. I feel that if smartphones
	are used only as information providers to our decision-making, then fine, it empowers us, but if it is used as a tool to
	make decisions itself, then we are enslaved to it as our discretion takes the back seat."
IR ₂	"Our organization is supportive of the use of smartphones in teaching-learning activities. For higher classes, we
	have provided mobile Apps to the students to learn lessons through smartphones during their free time. We find
	empowerment to use the phones in our decision-making activities, and sometimes we find answers to the questions
	asked by our students very easily in the Apps, which not only clarifies their doubts but also it uses creative ways to
	make the student comprehend them through audio-visual aids and registers it in their minds. To be frank, these Apps
	have reduced the burden of teaching, and we need not 'know it all' as smartphones assist us even in teaching. But
	on the flip side of it, we have observed that our knowledge base is diminishing, and we do not try to remember or
	recall even the simplest of the formulae, as they are stored in the App, and yes, in one way, we are also enslaved to
	the technology."
IR ₃	"Our organizational policies are congenial to the usage of smartphones during the working hours, and in fact, we
	cannot operate without some of the mobile Apps which are very handy. xxxx are the Apps we use almost on a daily
	basis and are quintessential for our job performance. I personally feel that the smartphone empowers me in my work
	more than enslavement, as it has become an indispensable tool in my kit. Of course, to some extent, I am enslaved to
	<i>it; however, as the positive impact is more than the negative, I feel it will remain a part of our day-to-day operations.</i>

The Relationship Between Organizational Context and Empowerment-Enslavement Paradox

Respondents	Organizational context (OCX) and \rightarrow Empowerment-Enslavement Paradox (EEP)
IR4	"Our organization is tuned to the use of smartphones in its functionality. It is more of an empowerment to us as it
	facilitates the accomplishments of our tasks much easier. For instance, I use xxxx App for simplifying complex
	workflows, xxx recruits to facilitate my hiring requirements, SmartH2R for my day-to-day activities as an HR manager,
	and there are many such Apps, which are indispensable to me as an HR Manager. It is difficult for me to answer if it
	empowers or enslaves me, as my answer to both is yes, but the positives outweigh the negatives, so I am happy the
	organization is supportive of its usage."
IR ₅	"As mentioned before, I frequently use smartphones to accomplish my day-to-day tasks as an accountant. Our
	organizational policies support the usage of mobiles during work Even my individual work is facilitated through
	mobile Apps, and I cannot be comfortable without it. My smartphone has empowered me to a great deal, but, at the
	same time, made me a slave to it also to some extent, as it rules me in my activities, and without it, I become a sort of
	handicapped. But I must mention that the positives are more than the negatives both in quantity as well as its
	magnitude."

Appendix G Enslavement Theme

Respondents	Excerpts of the in-depth interviews Enslavement Theme
IR ₁	"This adoption of technology has led to a situation of being at an advantage as well as at a disadvantage. Smartphone
	has given me the ability to contact any of our resources and retrieve the information I want for use at my work, but at
	the same time, my dependency on it has enslaved me to it."
IR ₂	" But on the flip side of it, we have observed that our knowledge base is diminishing, and we do not try to remember
	or recall even the simplest of the formulae, as they are stored in the App, and yes, in one way, we are also enslaved to
	the technology."
IR ₃	"Our organizational policies are congenial to the usage of smartphones during the working hours, and in fact, we
	cannot operate without some of the mobile I feel that the smartphone empowers me in my work more than
	enslavement, as it has become an indispensable tool in my kit. Of course, to some extent, I am enslaved to it; however,
	as the positive impact is more than the negative, I feel it will remain a part of our day-to-day operations."
IR ₄	"There is no end for HR issues in any organization as somebody would fall sick, somebody needs leave, somebody has
	to report to the Vice-president, and there is a major crisis at times where HR intervention is a must. So, in one way, I
	am enslaved to the technology, as I can be accessed from anywhere by my boss."
IR5	"The best part is, I can retrieve information from anywhere or any person from my organization. But at the same time,
	yes, it has enslaved me to itself as, if I don't have my smartphone with me, I get lost."

Appendix H Independence Theme

Respondents	Excerpts of the in-depth interviews Independence Theme
IR ₁	" He was not only impressed but inquired about the App I had used. This made me feel independent of my work, as I could produce the result instantly from wherever I was."
IR ₂	"it made me feel independent as I don't have to take the help of others or my physical presence with the student is not required always."
IR3	"Smartphone is my source for immediate access to any design data, maintenance data or order specifications in the oil rigs where I work most of the time, and it gives me the power to make independent decisions, but at the same time I am easily traceable by my boss, and it has snatched away my freedom completely, and it appears as if I am on the job 24/7."
IR4	It enables me to make independent decisions, and after I am convinced about the action plan based on the results I get from my App, I discuss it with my superior, and he gets easily convinced as my arguments are all based on facts and figures as supported through my App."
IR ₅	"I feel independent in decision making in connection to my job through the use of the right technology support."

Appendix I Dependence Theme

Respondent	Excerpts of the in-depth interviews Dependence Theme
IR ₁ Accountant	"But on the flip side of it, the smartphone has at times made me feel completely dependent on it as I feel
	handicapped without it"
IR ₂ Teacher	"sometimes, yes, it does make me feel that I am over-dependent on it, and it even snatched my independence
	as anyone can contact me anytime, even for trivial issues that waste my productive time.
IR ₃ Engineer	"While I feel I am independent in my operations, I also feel that I am over-dependent on technology as my
	operation will be standstill without it.
IR4 HR Manager	the situation is dichotomous as it also makes me feel dependent on it all the time, but the advantages are higher,
	so I ignore the negative effect of it."
IR 5 Legal Consultant	"Yes, the technology also makes me feel that I am over-dependent on it."

Appendix J Technological Context Theme

Respondents	Excerpts of the in-depth interviews Technological Context theme
IR ₁	"in our organizationI had to submit the balance sheet to the Vice-president for the quarter year with the available
	data in a few hours. I had xxxx mobile App installed on my smartphone, and within half an hour, I could project the assets
	and liabilities on the Balance Sheet as required by the Vice-president. He was not only impressed but inquired about the
	App I had used."
IR ₂	"
	aspect which may affect the students' career.
IR ₃	"The organization in which I serve is technology governed and naturally has a strong technological context. This strong
	context or background of technology empowers us to make the best use of technology to produce quality work. I have a
	clear picture of the limits of my usage of resources, and I do not have to consult my superiors on either the resource
	utilization or use of manpower, and it is left to my discretion."
IR4	"In our organization smartphone, usage is official. As an HR manager, it is a must for my day-to-day operations. Most
	of my communications with the employees during duty hours will be through smartphones. Sometimes the employees have
	doubts about the usage of some of the newly introduced HR schemes, and they call me, and I clarify their doubts over the
	phone. Most of the photos of the department activities are communicated to me through WhatsApp or uploaded to the
	site, and the link on the internet is communicated to me. In one way, all the interactions which last for a few minutes will
	be settled through the phone, and usage of smartphone is inevitable during the duty hours."
IR5	"As a legal consultant, I need to stay in touch with my clients throughout my service period as they keep contacting me
	through WhatsApp, especially when a legal issue pops up. So, my smartphone is an indispensable part of my professional
	life. Our organization has responded very well to this change in information handling and adopted very fast and
	introduced expertise both at the level of human and physical resources. This background motivates us to exploit the
	smartphone to the best of its use as we need to beat the competition."

Technological Context	(further ex	planation of	f the survey	questionnaire)
U	`	1		1 /

Respondents	Technological context (TCX)
IR ₁	1. My job always requires the support of my private smartphone
	"Smartphone is a part of my support device to establish communication with the client. Being an accountant, my
	smartphone is on during the entire working hours. Any queries on, e.g., the pay-slips of the employees, claim settlements, balance sheets, cash flow statements, profit loss accounts, atc. will be through smartphone, and I need to keep my device.
	close to me as I may receive calls from my superiors or colleagues at any time. Most of the queries will be answered
	through my smartphone. Only when more clarification of detailed discussions is required, my boss calls for personal
	interaction."
	2. During a typical workday, all my communications through my private smartphone are work-related.
	"I could say most of the communication is work-related. I do receive personal calls, but they are few, as my family and
	friends know that I have a busy schedule in my office. The rest of the communication is all work-related. When I report to
	the Senior Vice-president of the company, typically, they are issues connected to bookkeeping, financial statements, employee records connected to salary, pay rolls bills and payments, client invoicing, and it card receipts, company assets
	financial analysis reports government client reporting advise on decision making public accounting insurance claims
	etc. Sometimes just a few minutes' discussions on these aspects may save a lot of time for the higher management. But
	there are occasions when I have to speak longer to provide further details, so the duration of the call may even extend to
	an hour or so, especially when my boss is offsite and needs some clarifications on the financial aspects of the company."
	<u>3. I frequently need to use my private smartphone to meet work obligations.</u>
	"Yes, the connectivity to all the stakeholders of business during the entire work-day is through my smartphone. If my
	superior has something to communicate with, invariably, she contacts me inrough the smartphone. If the issue is small and can be sorted out over the phone, it will be settled. However, if documents have to be produced personal meetings
	may be necessary. If I can quickly recall, there was a call from my superior on 'bills and payments.' We have an application
	in our office which is tailor-made to undertake this activity. I need to capture the situation first into my application over
	the smartphone, in which I drop the invoice and auto-enter all the data into the database. I forward it to my superior for
	approval from my device to my superior's device. My superior will then sanction the payment on a priority basis and
	payment will be made. It will be immediately reconciled into the system by synchronizing the transaction to the system
	database. On some days, this operation will take place for hours until all the pending bills are cleared. So, on some days
	my entire work-day is connected through a smartphone.

Respondents	Technological context (TCX)
IR ₂	1. My job always requires the support of my private smartphone "Even though my smartphone will not be used during my teaching period, it will be active most of the time during the rest of the working hours. For instance, if I need to know the details of additional topics I need to cover in a chapter or change of schedule in timetable or an emergency meeting, or if I need an immediate response from my colleagues, I need to use my smartphone. So, my job demands the use of my smartphone all the time during duty hours, except during meetings and classes."
	2. During a typical workday, all my communications through my private smartphone are work-related. "Yes, as we are under the instruction that we are not supposed to take personal calls during the duty hours, I generally do not respond to calls from relatives or friends. Most of the time, the discussions over the phone are with my superiors and colleagues about the latest updates and rules issued, implementation of a decision of school councils, local community associations, strategic issues of school, resources utilization, operational issues, parent issues, department policies, legislative issues, environmental issues, etc. Some parents have a troubled child whom we will have to speak to even though there are trained counselors who respond to parent queries. But parents feel comfortable when they speak to the class teacher and know about the performance of their children. So, mostly yes, the mobile phones are used during the duty hours to discuss the issues related to school, students, and other matters in the school where a physical meeting may not be necessary."
	3. I frequently need to use my private smartphone to meet work obligations. "Not always, I can claim that my work-day is connected through the smartphone. But occasionally, yes, there have been days when I have been connected to the Vice-principal of the school for almost the entire day. For instance, when we had our school 'open day' dwing which the entire day parents used to visit us and discuss the issues related to their children.
	I had to get connected to the Vice-principal throughout the day periodically. This was because certain issues were the prerogatives of the Vice-principal, but the parents discussed it with me as I was more accessible to them. To seek the consent of my superior, I had to be in touch with the Vice-principal before I made any commitment to the parents. This

process was on for the entire day. So, there are days when I don't use my smartphones continuously for the entire day, and there are days when I do it depending upon the nature of the task to be accomplished on that day."

Respondents	Technological context (TCX)
IR3	1. <u>My job always requires the support of my private smartphone</u> "My job will be difficult to execute if not for my smartphone. During the field visits in the oil rigs, for technical assistance from my colleagues, the first contact point would be through my smartphone. If the technical issue is resolved, there will be no need for the technician to visit the maintenance site, or else, I may have to summon the concerned technician to be at the maintenance site during the scheduled maintenance program. During my work last week, there was a leakage in the oil seal from a diesel engine, and I had to quickly send the photo of the cylinder head through WhatsApp to the technician so that he has a clear idea of which seal he needs to procure despite me sending the specifications of the same. There are similar other issues to be resolved which demand the usage of smartphones during my work so taking the smartphone to my work site will be unavoidable and it becomes a part of my work."
	2.During a typical workday, all my communications through my private smartphone are work-related. "My duty is mainly in oil rigs, and the calls I receive are official. I mainly deal with rigging activities, so calls will be from the head office and are related to the inventory control, drilling operational issues and deviations from the plan if any, safety issues, compliance check, the effectiveness of activities, materials, and supplies, condition monitoring, procedure evaluation, drilling budget, daily operation reviews, work service order, dispatch of materials required at the rig, follow- ups on maintenance and repair, the scope of a project plan, review of project schedule, resources optimization, etc. Detailed meetings and documentation will be made during the personal meetings; however, some clarifications and inquires do happen as and when the problem arises, and smartphones are handy during that time." <u>3. I frequently need to use my private smartphone to meet work obligations.</u>

"Yes, I will be using my smartphone throughout the work-day mostly. As told before, I am a rig engineer, and I cannot be
there on-site all the time because I have to look after some overall maintenance issues at times. During those days, when
I am out of the worksite, the technicians will contact me on the smartphone and take my approval on repair or replace of
the part of the machine. They communicate to me the type of mechanical breakdown and even send to my smartphone the
picture of the component which has failed. I will have to study the situation and procure the part from the inventory or
approve replacement to the technician so that he can use the components available on the site. During some major
breakdown, the entire day may be used in the smartphone for bringing back the rig into its original working condition."

Respondents	Technological context (TCX)
IR4	1. <u>My job always requires the support of my private smartphone.</u> "Most of my communication with the employees during the duty hours will be through the smartphones. Sometimes the employees have doubts about the usage of some of the newly introduced HR schemes, and they call me, and I clarify their doubts over the phone. Most of the photos of the department activities are communicated to me through WhatsApp or uploaded to the site, and the link on the internet is communicated to me. In one way, all the interactions which last for a few minutes will be settled through the phone, and usage of smartphone is inevitable during the duty hours."
	2.During a typical workday, all my communications through my private smartphone are work-related. "Rarely do I use the mobile phone for personal calls during duty hours unless there is an emergency at home. It will be used mainly to handle issues such as HR strategy implementation, bridging the management and employee relations, recruitment issues, scheduling the meetings, training and development, employee welfare, HR support to business needs, developing a positive working environment, handling conflicts of interest, pays and benefits, performance appraisal, legal compliance, etc."
	3. <u>I frequently need to use my private smartphone to meet work obligations.</u> "I use the smartphone very often during my working hours. As an HR manager bridging the management and employee relations is an important part of my work, and it demands lots of negotiations with either of the sides. Even during the

training and development period, continuous usage of the phone will be required as there is a lot of coordination work to
be undertaken. In short, I can say that I would be dysfunctional without the smartphone available during the duty hours."

Respondents	Technological context (TCX)
IR5	1. <u>My job always requires the support of my private smartphone.</u> "Using my smartphone during working hours is a necessity for my profession. My clients keep calling over the phone. The nature of the claim, the policies, and procedures to be followed, the documents to be submitted, scheduling the meeting, etc., are all facilitated through phone calls. In fact, when I start my office from home, the first thing I check is if I have carried my smartphone with me or not. It is quintessential for my day-to-day operations, and it is a part of my work requirement. All the firsthand information regarding the legal issue will be collected through the phone call, and the client will be asked to submit the documents in person during the scheduled meeting. So, ye,s smartphone is part of my work requirement."
	 2.During a typical workday, all my communications through my private smartphone are work-related. "My nature of duty demands the usage of my smartphone to a great deal. The calls will be on discussing legal issues, drafting statutory and legal submissions, legal opinions, reviewing legal materials, supply of relevant documents, legal clauses, progress monitor of cases, service delivery, dispute handling, taxation issues, real estate deals, employment agreement, etc." 3. I frequently need to use my private smartphone to meet work obligations.
	"Yes, I will have to use my smartphone throughout the day sometimes. Say the drafting statutory and legal submissions need lots of information from various sources, and I need to make calls often. Similarly, while giving legal opinions, again and again, the clients call me while cross verifying if all the documents are ready for submission, I will have to check the concerned whenever there is a missing document, and I do it over the smartphone."
Respondents	Technological context (TCX) \rightarrow Empowerment-Enslavement Paradox (EEP)
-----------------	-------------------------------------------------------------------------------------------------------------------------
IR_1	My organization has accepted smartphones as one of the technologies which have to be used during the job execution.
	This adoption of technology has led to a situation of being at an advantage as well as at a disadvantage. Smartphone
	has given me the ability to contact any of our resources and retrieve the information I want for use at my work, but at
	the same time, my dependency on it has enslaved me to it. Without my smartphone at work, I feel I am handicapped,
	and every now, and then I look into my smartphone for account statements, bills, payment receipts, pending bills, etc.,
	from the past records."
IR_2	"Smartphone is one of the technologies which is being used in our school to make us perform better Smartphone
	provides me access to communicate with any teacher and take inputs while making decisions on important aspects
	which may affect the students. But sometimes, I also feel I am enslaved to it as I feel I am accessible to my higher
	authorities no matter whether I am at work or home. I am planning for a picnic with my family, and my superior calls
	me to work on the same day because my presence there is a must. If the smartphone was not there probably, I could
	have had many opportunities to enjoy my leave."
IR ₃	"As an engineering firm, we are attached to the latest technologies, and the context in which the technology is being
	used is very well supportive of our functionality. While the technology supports our thinking and decision-making
	abilities in connection to the job, at times, it also gives us a feeling that on many issues, we had much more freedom
	before this technology was introduced. A smartphone is my source for immediate access to any design data,
	maintenance data, or order specifications in the oil rigs where I work most of the time, and it gives me the power to
	make independent decisions, but at the same time, I am easily traceable by my boss, and it has snatched away my
	freedom completely, and it appears as if I am on the job 24/7. No matter what situation I have at home, being a
	maintenance engineer, I have to report it to the site as soon as I receive the call on a maintenance issue."

The Relationship Between Technological Context and Empowerment-Enslavement Paradox

Respondents	Technological context (TCX) → Empowerment-Enslavement Paradox (EEP)
IR4:	"The technology has been supportive in executing the HR aspects in my organization. The technology usage in the
	organization and the impact it has created on our functionality are considerable. There is no end to HR issues in any
	organization as somebody would fall sick, somebody needs leave, somebody has to report to the Vice-president, and
	there is a major crisis at times where HR intervention is a must. So, in one way, I am enslaved by technology, as I can
	be accessed from anywhere by my boss. But yes, it also empowers me to make decisions as I have all the data and
	information at my fingertips. I can call it 'a double-edged sword'."
IR5	"Smartphone as a technology is very well accepted in our organization. Our organization has provided a smartphone
	to all the employees who report to Vice-president in the organization, and they are all installed with Mobile Apps
	relevant to our work. Smartphone is a quick reference manual for me for any legal issue, and I have almost anything
	and everything in it to accomplish my job. The best part is, I can retrieve information from anywhere or any person
	from my organization. But at the same time, yes, it has enslaved me, especially when the feeling of being lost without
	my smartphone."

Respondents	Technological context (TCX) and \rightarrow Independence/Dependence Paradox (IDP)
IR ₁	"The technological context in our organization as stated before is strong on the usage of the smartphone during the
	working hours and to what extent it makes me feel independent in my work or dependent can be explained through
	the following situation very clearly. I was to submit the balance sheet to the Vice-president for the quarter year with
	the available data in a few hours. I had 'xxxxx' mobile App installed on my smartphone, and within half an hour, I
	could project the assets and liabilities on the Balance Sheet as required by the Vice-president. He was not only
	impressed but inquired about the App I had used. This made me feel independent of my work, as I could produce the
	result in no time from wherever I was. But on the flip side of it, the smartphone has at times made me feel completely
	dependent on it as I feel handicapped without it. But the positive feeling is much greater than the negative, and hence
	I continue to exploit it for my advantage."
R ₂	"As stated before, technology context is strongly developed in my school, and we are free to use the smartphone
	except in the restricted areas and times. This context does have a bearing on my feeling of being both independent
	as well as dependent on my smartphone. Independent in the sense that it helps me to facilitate teaching-learning.
	One of my students had difficulty understanding the). Although I explained to him the concept of finding the LCM,
	he wanted to test whether what he had understood was right. I gave him the APK App, which automatically gives the
	LCM, and asked him to practice manually and test it on the smartphone. It worked so well with him the outcome was
	achieved with ease. So, the technological context does make me feel independent as I don't have to take the help of
	others, or my physical presence with the student is always required, and it makes me feel independent. But sometimes,
	yes, it does make me feel that I am over-dependent on it, and it even snatched my independence as anyone can contact
	me anytime, even for trivial issues which waste my productive time. But, by and large, it provides me the confidence
	to act independently in my work."

The Relationship Between Technological context and Independence-Dependence Paradox

Respondents	Technological context (TCX) and \rightarrow Independence/Dependence Paradox (IDP)
R ₃	"The organization in which I serve is technology governed and naturally has a strong technological context. So, yes,
	I feel quite independent with my smartphone usage to make appropriate decisions, as I have supporting tools
	available at my fingertips. I am free to browse the details of the latest technology available in my field I could
	easily find the specifications of the latest oil pump specially designed for such conditions and had to just recommend
	it to my Chief Engineer and take his consent and place the order. When I replaced the unit, I got the problem of
	maintenance reduced completely, and even my boss was very happy as it was a long-time maintenance issue in our
	oil rig. So, yes, smartphone technology has made me feel independent in my operational domain. There are always
	ill effects associated with anything that is tried new, and so is the case of technology adoption. While I feel I am
	independent in my operations, I also feel that I am over-dependent on technology, and I feel handicapped if I don't
	have my smartphone with me. But, as long as it serves the purpose, I feel that the positive effect will overcome the
	negative influence."
IR ₄	"Our organization considers technology as the driving force to the business and hence the technology context is very
	strong particularly towards mobile technology. In my opinion, smartphone usage has made me independent in my
	decision-making as I have references to base my decisions on HR issues. Our company always aims towards the
	'right-sizing' of the organization. But, as an HR manager, I need to be accountable for the number I arrived at after
	the analysis of the workforce. It is in such situations I find smartphone technology comes to my rescue, and there are
	some very efficient Apps available for the rightsizing of the organization based on the job analysis. I have a mobile
	App 'xxxxx', which I find very handy in such situations. It enables me to make independent decisions, and after I am
	convinced about the action plan based on the results, I get from my App, I discuss it with my superior, and he gets
	easily convinced as my arguments are all based on facts and figures as supported through my App. So, It makes me
	feel independent in my task performance. But the situation is dichotomous as it also makes me feel dependent on it
	all the time, but the advantages are higher, so I ignore the negative effect of it."

Respondents	Technological context (TCX) and \rightarrow Independence/Dependence Paradox (IDP)
IR5	"Our organization has a strong technology orientation. This has given us access to all the supporting technologies,
	particularly smartphone-based technologies. I feel independent in decision-making in connection to my job through
	the use of the right technology support. My smartphone gives me all possibilities in a legal situation or all the support
	documents in a legal issue, and with minimum interference of my higher authorities, I can offer consultancy to my
	clients, and they are satisfied with the results I produce. There are checklists that I often make use of before arriving
	at a legal decision. None of my decisions have been disregarded by my superiors in the consultancy I have offered
	in insurance claims, tort claims, breach of contract, equitable claims, landlord/tenant issues, and so on. On the other
	side of it, yes, the technology also makes me feel that I am over-dependent on it. If connectivity is poor on a day, I
	am lost, and I cannot make a decision as I need to browse for the alternatives. But the negative effect is minimal
	compared to the use of it, and hence I feel quite positive about being more independent with my technology."

Respondents	Technological context (TCX) and $\rightarrow \frac{Task Performance (TSP)}{Task Performance (TSP)}$
IR ₁	"The technology in my organization is supportive of the smartphone usage, to a great extent, and most of my tasks are
	technology-enabled as mentioned before, and I can say with confidence that it has enhanced my task performance."
IR ₂	"Our school is technology-enabled in administration, teaching, and services. With particular emphasis on mobile
	technologies, I have found that it has contributed to the enhancement in my task performance. For instance, for
	teaching mathematics after my presentation in the class, I make my students make use of the mobile Apps to learn the
	concepts faster. So the learning outcomes are achieved with greater efficiency."
IR ₃	"Our firm being an engineering organization it cannot survive the competition without the best use of the latest
	technology, and hence it has a strong technology context. Today as most of the former technologies are available at
	our fingertips through the smartphone, I find it very handy to use in my design calculations and maintenance
	troubleshooting, and, yes, the technology context has contributed to my engineering task performance."
IR ₄	"We consider technology as a driver to our employee productivity. My observation is that the technology context
	provided in our organization has improved the task performance of the employees."
IR ₅	"Technology context is supportive of the use of smartphones in our organizations. As a consultant, I had to remember
	many legal codes in the former years, and working used to be very taxing. But the smartphones have made everything
	available for instant use through the Apps, and it has immensely contributed to my task performance."

The Relationship Between Technological Context and Task Performance

Appendix K Organizational Theme

Respondents	Excerpts of the in-depth interviews Organizational Context Theme
IR ₁	"the policies of the organization do support smartphone usage. While there is an instruction to avoid using smartphones for personal use during office hours by very nature, we do not use them for any unofficial purpose during the working hours unless it is an emergency."
IR2	" there are not many strategies developed in our school to promote smartphone adoption; however, there are teaching apps introduced into our smartphones that can be transferred to the computers/laptops and displayed to the students through xx software. There is a team that works on strategic planning to introduce all the useful apps to teach math and science in particular. So, yes, building a strong support strategy to facilitate smartphone usage is on the agenda of the strategic planning in our school."
IR ₃	"A smartphone was provided to me by my company on the day of induction, and it was a part of our company policy to use smartphones for official communications as and when it is required. My job will be difficult to execute if not for my smartphone. During the field visits in the oil rigs, for technical assistance from my colleagues, the first contact point would be through my smartphone. If the technical issue is resolved, there will be no need for the technician to visit the maintenance site, or else, I may have to summon the concerned technician to be at the maintenance site during the scheduled maintenance program."
IR4	"In fact, the use of smartphones has transformed the HR practices in the sense that tracking and monitoring HR activities have become much easier and faster with the use of smartphones. All the organizational policies, procedures, practices, and systems can be viewed easily on the smartphone. our organizational policies are very supportive to its use."
IR5	"Our organization has a clear smartphone strategy mainly aimed towards the maximization of its usability. There are several mobile apps that we regularly use during our work, and we should make sure they are the most recent apps and have all the necessary features. Documentation is the most important part of legal consultancy, and digitization of the legal documents is a critical factor to be considered. We spend much time strategizing on how to ensure security to these legal documents in the digital form, store them in a secure database and retrieve them with ease at the shortest possible time."

Respondents	Organizational context (OCX) and \rightarrow Independence-Dependence Paradox (IDP)
IR ₁	"We have a good organizational context supportive of the adoption of technology in the form of smartphones. However,
	I do not see the impact of these on my feeling of independence or dependence on technology. In fact, the technology
	context of the organization can have an impact on such a feeling. The organization does have an influence on my
	behavioral aspects in the performance of the job, such as my desire to stay competitive in my field, my desire to acquire
	higher qualifications, or my desire to contribute to the organization. But I do not find how the policies and procedures
	which have strategic orientation can make me feel independent or dependent on technology."
IR ₂	"Our organizational context is aligned with smartphone usage, and it helps us in our task performance. I have given
	examples earlier that smartphone applications have been exploited by us to make learning easier and enjoyable for our
	students. So, I am confident about the positive association between organizational context and task performance."
IR ₃	"In my opinion, the organizational context is more about the organizational structure which is supportive of the strategies
	of the organization. It can help a great deal in utilizing the resources in the organization in accomplishing the
	organizational objectives. But my feeling of being dependent or independent of technology is more dependent on the
	technology content of the organization rather than the organizational context."
IR4	"The organizational context is more connected to my department, as it involves the building of the structure for
	accomplishing the organizational, mainly the human aspects. The policies, procedure, and tactics of the company may
	be complementary to the accomplishment of the objectives, but I don't think it can influence the state of being independent
	or dependent on technology."
IR5	"The organizational context can support the human aspect in the performance of the job, but it cannot have an impact
	on my being independent or dependent on technology as these are not very much related. For instance, the company
	policy may empower me to gain access to all the resources which are necessary for me to prepare a legal document. But
	the policies, procedures, and systems do not have much influence on my feeling of being independent or dependent on
	technology for executing a task."

The Relationship Between Organizational Context and Independence-Dependence Paradox

The Relationship Between Organizational Context and Task Performance (questionnaire)

Respondents	Organizational context (OCX) and \rightarrow Task Performance (TSP)
IR ₁	The policies of the organization encourage the use of private smartphones for work.
	"Smartphone usage during working hours has become a part of the culture of our organization. The policies of the
	organization are in favor of smartphone usage as it contributes to knowledge-sharing and facilitates the day-to-day
	activities, which include scheduling of the meetings, clarification on the bills and reports submitted to higher authorities,
	during the duty hours, and on the contrary, the company has introduced the policy of permitting the employees to bring
	their own mobile devices to the workplace "
	The strategy of the organization promotes the use of private smartphones for work.
	"Our organization has strategies to continuously enhance mobile apps for accountancy as it has an impact on our
	business performance. The main focus is on the functionality of the applications and how they can save time and increase
	focussing on the simplification of the processes involved in accounting, and it has a hearing on our performance too. So
	yes, the strategies of the organization are supportive of smartphone adoption."
	<u>The tactical plans (implementation of work plans) encourage the use of private smartphones for work.</u>
	Tactical issues are mainly dealing with how we operationalize our business strategy. So, while cell phone adoption is
	helping us in gathering the required information in the right place at the right time, tactical issues are left to our discretion
	on how we are going to use the information we receive to the best advantage of the organizations. So, yes, we are
	empowered by the organization to use the information that we find most suitable in a given situation."

Respondents	Organizational context (OCX) and \rightarrow Task Performance (TSP)
IR ₂	The policies of the organization encourage the use of private smartphones for work.
	"Our school has an atmosphere congenial to the use of smartphones except in some places, e.g., meeting rooms, classrooms, library, common rooms for teachers, and places where it could disturb the academic environment. Announcements will be made well in advance during the meetings or programs to keep the mobile phones in silent mode. The rest of the time, we are free to use our smartphones, as they are tools for communication between all the stakeholders of the school. The school principal and vice-principal have made a rule that teaching and non-teaching staff should be reachable during working hours. So, the organizational settings are congenial to the usage of smartphones."
	The strategy of the organization promotes the use of private smartphones for work. "As such, there are not many strategies developed in our school to promote smartphone adoption; however, there are teaching apps that are introduced into our smartphones which can be cast to the computers and displayed to the students through PowerPoint in Microsoft. There is a team that works on strategic planning to introduce all the useful apps to teach maths and science in particular, which is of interest as teaching is relatively difficult in those two courses. The advantage of these apps is that we teachers can browse them during travel or at home so that we are familiar with its functionality. So, yes, building a strong support strategy to facilitate smartphone usage is in the agenda of the strategic planning in our school."
	The tactical plans (implementation of work plans) encourage the use of private smartphones for work.
	"Yes, we are given the complete freedom to use the smartphones the way we feel most useful in accomplishing the institutional objectives. For instance, there are several apps available for teaching mathematical operators. But, depending upon the individual abilities of the students, we have the flexibility on which one would be a more appropriate mobile app for a student to learn faster and retain the knowledge gained for a longer duration. So, in that sense, the tactical issues are supportive of the smartphone usage."

Respondents	Organizational context (OCX) and \rightarrow Task Performance (TSP)
IR ₃	The policies of the organization encourage the use of private smartphones for work. "The very nature of my duty demands the use of smartphones, and the organization is well-set for the usage of it during the working hours. Most of the technical issues get resolved over the phone, and it saves our time and resources, so the organization encourages the use of mobile phones during working hours. Today's smartphones are so sophisticated that we can transfer the design data over the phone and use it on the site as and when required. There are occasions when the 'standard operating procedure' of the equipment has been observed in the smartphone before operating on the machine directly. Safety issues are saved on smartphones so that they can be handy on the sites. The first thing we look into during an emergency is the mobile phone. So, I can say that a positive atmosphere has been built towards the carrying of a smartphone to the workplace.
	The strategy of the organization promotes the use of private smartphones for work "Yes, we have a well-developed strategy to strengthen the apps which can be used on our smartphones on maintenance aspects. It is realized that a clear mobile strategy can make business goals more achievable. The ability to access distant servers, instant transfer of information, and videoconferencing are powerful tools that can provide the necessary boost to the maintenance operations in the rigs in both periodic and breakdown maintenance. Each year separate meetings are held to strengthen the smartphone support strategies in our organizations."
	The tactical plans (implementation of work plans) encourage the use of private smartphones for work. "I feel tactical plans are related to the way we execute the operations based on the information we have received over the smartphone. Even though the right kind of information with graphics or media is received, the operation of the particular machine component, say oil pump, may vary slightly depending upon the model. So, we will have to use our own judgment on how to use the information to the best of our abilities to operationalize the oil pump. So, yes, our policies are supportive of the tactical aspects of executing a technical task."

Respondents	Organizational context (OCX) → Task Performance (TSP)
IR4	The policies of the organization encourage the use of private smartphones for work. "Smartphones have provided flexibility and mobility to the workplace, and their use has been encouraged in the workplace. My job as an HR manager is to place the right person in the right place and supply to him/her the right information regarding recruitment, placement, training, development, performance, and promotion and smartphone is the tool that makes information handling much easier, and as a policy, its use is supported in the organization. Diversity of the workforce is the most distinguishing feature of today's HR environment, and handling a diverse workforce becomes much easier over the smartphone, as it provides flexibility and eases communication both in oral and written form supported with pictures and animation. In fact, the use of smartphones has transformed HR practices in the sense that tracking and monitoring HR activities have become much easier and faster with the use of smartphones. I had received a cell phone from my company the day I was put into work, and it has been upgraded as newer features were introduced, and it is a part of my paraphernalia. As all the organizational policies, procedures, practices, and systems can be viewed easily on the smartphone round the clock, our organizational policies are very supportive of its use.
	The strategy of the organization promotes the use of private smartphones for work "Strategic planning in terms of the mobile strategy is mainly on three aspects: customer relationship management (CRM), supply chain management (SCM), and enterprise resource planning (ERP) in our organization. Today the ERP aspects are brought to the smartphones, and employees are given access to the company database. We in the HR department are mainly involved in 'change management,' which is quintessential for the success of any new strategy introduced in the organization. As the usage of smartphones has both positive and negative impacts on employee performance, proper training is essential for the right usage of these devices. The key issue is about findings means to provide access to the company's resources to the employees so that they make the use of the same to enhance their productivity."
	The tactical plans (implementation of work plans) encourage the use of private smartphones for work.
	"As an HR manager, I need to use my abilities and experience on tactical issues. While strategies remain the same, tactical execution may change based on the situation. When I communicate with our employees, there are several issues that need to be considered, like their communication styles, ethnicity, cultural background, and a lot of

demographic aspects, which have a bearing on the way they act and behave in a given situation. So, I will have to change my tactical aspects of executing the HR strategies depending upon the person whom I am dealing with. The organization has provided to me the complete freedom to act the way I wish provided that the task is executed in a congenial manner."

Respondents	Organizational context (OCX) → Task Performance (TSP)
IR5	The policies of the organization encourage the use of private smartphones for work.
	"Computer accessed communication is a part of work culture in legal firms, so the organizational policies are
	conducive to its usage during consultation services. There are several legal activities apps on smartphones which
	include- online notary services, legal legislation, case browse, legal document generation, etc. So, our organization
	has policies which are very supportive of the usage of smartphones."
	2. The strategy of the organization promotes the use of private smartphone for work.
	"Our organization has a clear smartphone strategy mainly aimed towards the maximization of its usability. As mentioned before, there are several mobile apps which we use on a regular basis during our work, and we should make sure they are the most recent apps and have all the necessary features. Documentation is the most important part of legal consultancy, and digitization of the legal documents is a critical factor to be considered. We spend a lot of time in strategizing on how to ensure security to these legal documents in the digital form, store them in a safe database and retrieve them with ease at the shortest possible time."
	3. The tactical plans (implementation of work plans) encourage the use of private smartphones for work.
	"This point is not very much applicable to my profession, as we do not have many tactical issues to be handled in our profession. Conformance to the law has a single way of executing it irrespective of the background of the person, and not knowing the law is no excuse from the law. So, there are no situations where I need to be more tactical in my approach."

Respondent	Excerpts of the in-depth interviews Task Performance (TSP) Theme
IR ₁	"I frequently use smartphones to accomplish my day-to-day tasks as an accountant. Our organizational policies support the usage of mobiles during the work, and I can say that at least 60% of our communication with the peer group is through mobile."
IR ₂	" If I can recall it correctly, there will be parents having a troubled child with whom we have to speak to despite the fact that there are trained counselors who respond to parent queries. But parents feel comfortable when they speak to the class teacher and know about the performance of their children. So, mostly yes, the mobile phones are used during the duty hours to discuss the issues related to school, students, and the resources in the school where a physical meeting may not be necessary."
IR ₃	"I will be using my smartphone throughout the workday mostly. I am a rig engineer and cannot be there on-site all the time because I have to look after some overall maintenance at times. During those days when I am out of worksite, the technicians will contact me and take my approval on repair or replace the part of the machine. They communicate to me the type of mechanical breakdown and even send to my smartphone the picture of the component which has failed. I will have to study the situation and procure the part from the inventory or approve replacement to the technician so that he can use the components available on-site. During some major breakdown, the entire day may be used in the smartphone for bringing back the rig into its original working condition."
IR4	"I use my smartphone very often during my working hours. As an HR manager bridging the management and employee relations is an important part of my work, and it demands lots of negotiations with either of the sides. Even during the training and development period, continuous usage of the smartphone will be required as there is a lot of coordination work to be undertaken. In short, I can say that I would be dysfunctional without the smartphone available during duty hours."
IR5	" The nature of the claim, the policies, and procedures to be followed, the documents to be submitted, scheduling the meeting, etc., are all facilitated through phone calls. In fact, the first thing I check is if I have carried my smartphone with me or not. It is quintessential for my day-to-day operations, and it is a part of my work requirement. All the first-hand information regarding the legal issue will be collected through phone calls, and the client will be asked to submit the documents in person during the scheduled meeting. So, yes smartphone is part of my work requirement.

Appendix L Task Performance Theme

Empowerment-Enslavement Paradox (EEP) and \rightarrow Task Performance (TSP)
"In my opinion, any employee who feels empowered would have a better task performance. I had had several
situations in which when my Vice-president had given me complete autonomy, I had accomplished the task very well.
There was a case of our annual audit, and the VP had given me the complete authority to deal with the external
auditor. I had the liberty to fix the time frame as found fair to the staff members, decide the audit procedure with the
external auditor, maintain the fixed assets additions and disposal, debt agreements, lawsuits, etc., reconcile the bank
account, etc. There were many key decisions that I had to take in consultation with the auditor. The entire process
was so smoothly conducted that in the end, our VP congratulated me in our annual meeting for shouldering his
responsibilities, and it would not have been possible if he had not empowered me the entire task. It will be difficult
if the superiors interfere in every other activity in accomplishing a major task. So, yes, while empowerment may
enhance task performance, enslavement may bring down the task performance."
"In our school, the teachers are empowered to make decisions and take actions accordingly within the framework
of school rules and regulations. We have a clear understanding of our responsibilities, so we feel empowered in our
duties. There are many cases in which I had the freedom to take the initiative and empower myself in order to help
the student perform better. For instance, student counselling or sometimes parent counselling is an important duty
of a teacher I strongly believe that yes empowerment of the teachers will make them do better in their task,
whereas the feeling of enslavement could bring down the task performance."
-

Th Relationship between Empowerment-Enslavement Paradox and Task Performance

Respondents	Empowerment-Enslavement Paradox (EEP) and \rightarrow Task Performance (TSP)
R ₃ Engineer:	"During the orientation program, the employees in our company were not only briefed about our roles and
	responsibilities but also informed about the empowerment, job enrichment, job enlargement, and so on. Within the
	limits and capacity of our work, we are empowered to make decisions on certain issues which may need immediate
	attention. The higher authorities will never interfere in our decision-making on such occasions and support us as
	they are convinced about our corporate citizenship behavior. Not only me but all my colleagues, to a very great
	extent, are convinced that empowerment is positively associated with task performance, but it 'comes with a rider'
	in the sense that there must be a clear understanding about to what extent an employee is empowered to make a
	decision on an issue. Sometimes the feeling of enslavement may also increase the task performance because the
	enslaved nature of performance will be like sticking on to the rules and regulations and executing all the tasks as
	demanded by the higher authorities. So, I don't agree with the concept that empowerment can enhance task
	performance and enslavement can bring down the task performance; instead, there is a possibility that both can be
	contributing towards enhanced task performance."
IR5 Legal	"Empowerment goes by the designation and the importance of the tasks being handled by the employees. At lower
consultant:	levels of operation, complete empowerment, particularly when the employee is new to the organization, may not be
	advisable on the basis of company reputation and client satisfaction. The employee needs time to understand the
	organizational culture and be a part of the company, and know exactly how the senior officers may respond to a
	situation and emulate the same behavior. When I was newly recruited in the company about 15 years ago, I was not
	empowered to execute my tasks in the company to the extent I am today. As I report to the top level of management
	at my capacity of Senior Legal Consultant, I do have my privileges, and I am definitely empowered in my job to make
	decisions at my own discretion on issues such as legal requirements in a case, fixation of consultancy fees, using the
	company resources for executing the tasks, assuring the type of deliverables to the customer, deciding on the timeline
	and so on. My senior management has never interfered or questioned my decisions on these matters, and this
	empowerment has enabled me to perform much better. Yes, at senior levels, the empowerment may lead to better task
	performance, but at junior level, enslavement may produce better results, in my opinion."

Respondents	Contextual Factors (CNF) \rightarrow Technology Paradox (TCP) \rightarrow Task performance (TSP)
IR ₁	"At the time of joining the organization, we are provided adequate orientation towards the organizational culture.
	This provides us adequate exposure to the context in which we operate in the organization. It was understood by me
	that the organization had a clear vision, and the objectives were in alignment with the vision of the organization,
	and the latest technologies were in place to accomplish those objectives. Thus, the context was very supportive of the
	technology usage, say smartphones.
	During my experience in the organization, I could perceive both the positive and negative influences of smartphones.
	Sometimes I felt that it made me feel that as I have access to all the knowledge, I require I am well-equipped to
	perform any challenging task. But at the same time, I felt that my individual capacities were not being used, and I
	had become a slave to the technology. I observed that even for simple calculations, I used to take the help of my
	smartphone. However, over a period of time, I found that despite the negative influences it has, smartphones are
	contributing positively to my competencies, and it has improved my task performance.
	I had observed during my work experience that whenever I was empowered on my job, I would easily accomplish the
	task, and when I generated the feeling of being enslaved, either to technology or the instructions of my higher
	authorities, the task performance fell down. It is mainly due to the motivational aspect of performing the task. While

Appendix M The Mediation Effect of The Technology Paradox

	I am empowered, I am well-motivated, and it leads to better task performance, and when enslaved, my performance
	would fall down as there is no sense of ownership for the task accomplishment."
IR ₂	In my opinion, the school has supportive contextual factors that have an influence on the behavior of the teachers,
	and accordingly, they perform their duties and tasks. In our school, the context is set right for the usage of
	smartphones for productive use. I have personally experienced both the positive and negative influences of
	smartphones. Sometimes I feel empowered and sometimes enslaved to it. For instance, I am contacted by my peer
	group no matter where I am, and when the phone rings, it needs to be attended to, particularly if it is from the Vice-
	principal and the flow of my teaching is lost for the moment, which constitutes a sort of dependency on it. This has
	its influence on my task performance, as I would lose the flow of my teaching. Thus, the context, the state of my
	behavior towards smartphone usage, and my task performance are linked to each other in this order.
	It is important to note that the nature of my job is highly 'knowledge-intensive,' and this kind of job requires to be
	set-free in its execution. For creative ways of explaining a course topic, I need to be highly motivated and energized
	to deliver a quality lesson to my students. When things are more formalized, and I am asked to repeat the same things,
	the creativity takes the backseat, and task performance falls abruptly. So, the dual nature of technology making me
	empowered or enslaved has an effect, and it is controlled through the context provided by the administration and has
	its bearing on the outcome in the form of my class delivery."

LIST OF PUBLICATIONS

- International Journal of Technology and Human Interaction (IJTHI) Volume 18, Issue 4, Article 2.
- 2. Contextual Factors, Technology Paradox, and Job Performance in Smartphone Usage: A Systematic Review